UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT
Pursuant to Section 13 OR 15(d) of
The Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): October 31, 2018

Altair Engineering Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation)

001-38263 (Commission File Number) 38-2591828 (IRS Employer Identification No.)

1820 E. Big Beaver Road Troy, Michigan (Address of principal executive offices)

48083 (Zip Code)

Registrant's telephone number, including area code: (248) 614-2400

 $\begin{tabular}{ll} Not \ Applicable \\ (Former name or former address, if changed since last report) \end{tabular}$

	Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligations of the registrant under any of the following provisions:				
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)				
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)				
\boxtimes	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))				
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))				
Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§ 230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§ 240.12b-2 of this chapter).					
Emerging growth company $oxtimes$					
	f an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.				

Item 1.01 Entry into a Material Definitive Agreement.

A. Agreement and Plan of Merger

On November 5, 2018, Altair Engineering Inc., a Delaware corporation (the "Company"), entered into an Agreement and Plan of Merger (the "Merger Agreement") with Datawatch Corporation, a Delaware corporation ("Datawatch"), and Dallas Merger Sub, Inc., a Delaware corporation ("Purchaser") and a wholly owned subsidiary of the Company. The boards of directors of the Company and Datawatch have unanimously approved the Merger Agreement.

Pursuant to the Merger Agreement, upon the terms and subject to the conditions thereof, Purchaser will commence a tender offer (the "Offer") to purchase all of the issued and outstanding shares of common stock, par value \$0.01 per share (the "Shares"), of Datawatch at a purchase price of \$13.10 per Share, net to the seller in cash (the "Offer Price"), without interest and less any applicable withholding taxes. Purchaser will commence the Offer as promptly as practicable after the date of the Merger Agreement, but in no event more than 10 business days after the date of the Merger Agreement (or such later date as the parties may agree). The Offer will expire at one minute after 11:59 p.m. (Boston time) on the date that is 20 business days (calculated in accordance with Rule 14d-1(g)(3) under the Securities Exchange Act of 1934, as amended) following the commencement of the Offer, unless extended in accordance with the terms of the Merger Agreement, including as required by the applicable rules and regulations of the United States Securities and Exchange Commission (the "SEC").

The obligation of the Company to purchase Shares tendered in the Offer is subject to the satisfaction or waiver of a number of conditions set forth in the Merger Agreement, including (i) that there will have been validly tendered and not validly withdrawn Shares that represent not less than one share more than 50% of the total number of Shares outstanding at the time of the expiration of the Offer and (ii) the expiration or termination of any applicable waiting period (or extension thereof) under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended. The consummation of the Offer is not subject to any financing condition.

Following the completion of the Offer and subject to the satisfaction or waiver of the remaining conditions set forth in the Merger Agreement, (i) the Company will promptly accept for payment (such time of acceptance for payment, the "Offer Acceptance Time") and pay for all Shares validly tendered pursuant to the Offer and not validly withdrawn prior to the Offer Acceptance Time, and (ii) Purchaser will merge with and into Datawatch as soon as practicable without a meeting of Datawatch's stockholders in accordance with Section 251(h) of the General Corporation Law of the State of Delaware, with Datawatch surviving as a wholly-owned subsidiary of the Company (the "Merger").

At the effective time of the Merger (the "Effective Time"), any Shares not purchased pursuant to the Offer (other than Shares (i) owned by Datawatch's stockholders who have perfected their statutory rights of appraisal under Delaware law (ii) then owned by Datawatch or owned both at the commencement of the Offer and at the Effective Time by any wholly owned subsidiary of Datawatch and (iii) irrevocably accepted for purchase in the Offer or owned both at the commencement of the Offer and at the Effective Time by Altair, Purchaser or any other wholly owned subsidiary of Datawatch) will be automatically converted into the right to receive the Offer Price, net to the seller in cash, without interest and less any applicable withholding taxes.

Pursuant to the Merger Agreement, each stock option to purchase Shares ("Datawatch Options") that was granted pursuant to the Datawatch Third Amended and Restated 2011 Equity Compensation and Incentive Plan, as amended (the "Datawatch Stock Plan"), and is unexpired, unexercised and outstanding as of the Effective Time, whether vested or unvested, will be automatically cancelled and converted into the right to receive an amount in cash, without interest and less applicable tax withholding, equal to the product of (i) the excess, if any, of (A) the Offer Price over (B) the per share exercise price for such Datawatch Option, multiplied by (ii) the total number of Shares underlying such Datawatch Option as of immediately prior to the Offer Acceptance Time. No holder of a Datawatch Option that has an exercise price per Share that is equal to or greater than the Offer Price will be entitled to any payment with respect to such cancelled Datawatch Option before or after the Effective Time.

Pursuant to the Merger Agreement, each restricted stock unit ("Datawatch RSUs") that was granted under the Datawatch Stock Plan and is outstanding and vested immediately prior to the Effective Time (after giving effect to any accelerated vesting thereof upon completion of the Merger) will be automatically cancelled and converted into the right to receive an amount in cash, without interest, equal to the product of (i) the Offer Price multiplied by (ii) the total number of Shares that would otherwise have been issuable upon settlement of the vested portion of such Datawatch RSU, less any withholding taxes.

Each Datawatch RSU that was granted under the Datawatch Stock Plan and is outstanding and will not by its terms vest as of the Effective Time (after giving effect to any accelerated vesting thereof upon completion of the Merger) will be automatically converted into and become a substitute right to receive an amount in cash, without interest, equal to the product of (i) the Offer Price multiplied by (ii) the total number of Shares underlying the unvested portion of such Datawatch RSU, less any withholding taxes (such payment, the "Unvested RSU Consideration"). The payment of the Unvested RSU Consideration, if any, with respect to such substitute right will be payable on or after the date that such Datawatch RSU would have otherwise vested in accordance with its terms (and in any event not later than the earlier of (1) December 31st of the year during which such vesting date occurs or (2) the first regular payroll date following the date of termination for any such employee) but, subject to certain accelerated vesting terms, only if such conditions to vesting are satisfied prior to such vesting date.

Each holder of warrants to purchase Shares ("Datawatch Warrants") that are unexpired, unexercised and outstanding as of the Effective Time, will have the right, upon valid exercise of the Datawatch Warrants, to receive an amount in cash, without interest and less applicable tax withholding, if any, equal to the product of (i) the excess, if any, of (A) the Offer Price over (B) the per share exercise price for such Datawatch Warrants, multiplied by (ii) the total number of Shares underlying such Datawatch Warrants as of immediately prior to the Effective Time.

The Merger Agreement includes representations and warranties and covenants of the parties customary for a transaction of this nature. During the period starting on the date of the Merger Agreement and ending at the Effective Time or such earlier date on which the Merger Agreement is terminated, Datawatch has agreed to operate its business and the business of its subsidiaries in the ordinary course and has agreed to certain other operating covenants, as set forth more fully in the Merger Agreement. Datawatch has also agreed, subject to certain exceptions set forth in the Merger Agreement intended to permit its board of directors to comply with its fiduciary duties, (i) to recommend that Datawatch's stockholders accept the Offer and tender all of their Shares pursuant to the Offer; (ii) not to solicit the making of any proposal that could reasonably be expected to lead to an alternative acquisition proposal or participate in any discussions with any third party regarding an alternative acquisition proposal; and (iii) to certain restrictions on its ability to respond to any unsolicited acquisition proposals.

The Merger Agreement includes a remedy of specific performance for the Company, Purchaser and Datawatch. The Merger Agreement also includes termination provisions for both the Company and Datawatch. In addition, in connection with the termination of the Merger Agreement under specified circumstances, including termination by Datawatch to accept and enter into a definitive agreement with respect to an unsolicited superior proposal, Datawatch will be required to pay a termination fee of approximately \$6.1 million (the "Termination Fee"). A superior proposal is any bona fide and written proposal pursuant to which a third party would acquire, among other acquisition structures set forth in the Merger Agreement, more than 80% of the voting power of Datawatch, which proposal is binding on the third party making such proposal and is made pursuant to terms that the board of directors of Datawatch determines in good faith (after consultation with Datawatch's financial advisors and outside legal counsel) render the proposal reasonably likely to be consummated if accepted on its terms and the proposal to be more favorable from a financial perspective to Datawatch's stockholders than the Offer and the Merger. Any such termination of the Merger Agreement by Datawatch is subject to certain conditions, including Datawatch's compliance with certain procedures set forth in the Merger Agreement and a good faith determination by Datawatch's board of directors that the failure to take such action would be inconsistent with its fiduciary duties under applicable law (after providing the Company with prior written notice that Datawatch is taking such action and with an opportunity to negotiate), payment of the Termination Fee by Datawatch and the execution of a definitive agreement by Datawatch with such third party.

The foregoing description of the Merger Agreement is not complete and is qualified in its entirety by reference to the Merger Agreement, which is attached as Exhibit 2.1 to this report and incorporated herein by reference.

B. Tender and Support Agreement

Concurrently with the execution and delivery of the Merger Agreement, on November 5, 2018, Michael Morrison (Datawatch's President, Chief Executive Officer and board member), James Eliason (Datawatch's Chief Financial Officer and Treasurer), Ken Tacelli (Datawatch's Chief Operating Officer), each of the other members of Datawach's board and certain other stockholders, each entered into a tender and support agreement (the "Support Agreement") with the Company and Purchaser, which provides, among other things, that these stockholders will tender their Shares in the Offer and, if necessary, vote their Shares against any competing proposal. These stockholders may only withdraw their Shares from the Offer if the Support Agreement is terminated in accordance with its terms, including if the Merger Agreement is terminated. The Shares subject to the Support Agreement represent approximately 14.3% of the outstanding Shares as of October 31, 2018.

The foregoing description of the Support Agreement is not complete and is qualified in its entirety by reference to the Support Agreement, the form of which is attached as Exhibit 2.2 to this report and incorporated herein by reference.

C. Amendment to Credit Agreement

Effective October 31, 2018, the Company entered into the First Amendment to its Third Amended and Restated Credit Agreement, by and among the Company, as borrower, the lenders party thereto, and JPMorgan Chase Bank, N.A., as Administrative Agent (the "Amendment"), which amends the Third Amended and Restated Credit Agreement, dated as of October 18, 2017 by and among the Company, as borrower, the lenders party thereto and JPMorgan Chase Bank, N.A., as Administrative Agent, Sole Bookrunner and Sole Lead Arranger (the "Amended and Restated Credit Agreement").

Pursuant to the terms of the Amendment, the aggregate commitment amount of the lenders under the Amended and Restated Credit Agreement has been increased from \$100 million to \$150 million. Except as expressly modified by the Amendment, the Amended and Restated Credit Agreement remains in full force and effect.

The foregoing description of the Amendment is not complete and is qualified in its entirety by reference to the Amendment, the form of which is attached as Exhibit 10.1 to this report and incorporated herein by reference.

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant

The information set forth in Part C of Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference.

Item 7.01 Regulation FD Disclosure.

On November 5, 2018, the Company issued a joint press release with Datawatch entitled "Altair Announces Agreement to Acquire Datawatch."

The foregoing press release is furnished as Exhibit 99.1 to this report.

Additional Information

The Merger Agreement, the Support Agreement and the Amendment, and the foregoing descriptions of each agreement, have been included to provide investors and stockholders with information regarding the terms of each agreement and are not intended to provide any financial or other factual information about the Company, Purchaser or Datawatch. In particular, the representations, warranties and covenants contained in the Merger Agreement were made only for purposes of that agreement. The assertions embodied in the representations and warranties contained in the Merger Agreement were made solely for the benefit of the parties to the Merger Agreement and are qualified by information in confidential disclosure schedules delivered by Datawatch to the Company in connection with the signing of the Merger Agreement. Moreover, certain representations and warranties in the Merger Agreement were

made as of a specified date, may be subject to a contractual standard of materiality different from what might be viewed as material to stockholders, or may have been used for the purpose of allocating risk between the parties to the Merger Agreement rather than establishing those matters as facts. Accordingly, the representations and warranties in the Merger Agreement should not be relied on by any persons as characterizations of the actual state of facts and circumstances of Datawatch or the Company at the time they were made and the information in the Merger Agreement should be considered in conjunction with the entirety of the factual disclosure about Datawatch and the Company in Datawatch's and the Company's public disclosures filed with the SEC. Information concerning the subject matter of the representations and warranties may change after the date of the Merger Agreement, which subsequent information may or may not be fully reflected in public disclosures filed by the Company or Datawatch with the SEC.

Notice to Investors/Important Additional Information will be Filed with the SEC

The Offer has not yet commenced. This report and the attached exhibits are for informational purposes only and are neither an offer to purchase nor a solicitation of an offer to sell any securities of Datawatch nor are they a substitute for the tender offer materials that the Company will file with the SEC. The solicitation and the offer to purchase the Shares will only be made pursuant to a tender offer statement on Schedule TO, including an offer to purchase, a letter of transmittal and other related materials that the Company intends to file with the SEC. Thereafter, Datawatch will file with the SEC a Solicitation/Recommendation Statement on Schedule 14D-9 with respect to the Offer. THE TENDER OFFER MATERIALS (INCLUDING THE OFFER TO PURCHASE, A RELATED LETTER OF TRANSMITTAL AND CERTAIN OTHER TENDER OFFER DOCUMENTS) AND THE SOLICITATION/RECOMMENDATION STATEMENT WILL CONTAIN IMPORTANT INFORMATION. DATAWATCH'S STOCKHOLDERS ARE URGED TO READ THESE DOCUMENTS (AS EACH MAY BE AMENDED OR SUPPLEMENTED FROM TIME TO TIME) CAREFULLY AND IN THEIR ENTIRETY PRIOR TO MAKING ANY DECISIONS WITH RESPECT TO WHETHER TO TENDER THEIR SHARES INTO THE OFFER, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION, INCLUDING THE TERMS AND CONDITIONS OF THE OFFER. Once filed, investors will be able to obtain the tender offer statement on Schedule TO, the offer to purchase, the Solicitation/Recommendation Statement of Datawatch on Schedule 14D-9 and related materials with respect to the tender offer and the merger, free of charge at the website of the SEC at www.sec.gov or from the information agent named in the tender offer materials. Investors may also obtain, at no charge, the documents filed with or furnished to the SEC by the Company under the "Investor Relations" section of the Company's website at www.altair.com. Copies of the documents filed with or furnished to the SEC by Datawatch will be available at no charge under the "Investor Relations" section of Datawatch's website at www.datawatch.com.

In addition to the Offer to Purchase, the related Letter of Transmittal and certain other tender offer documents, as well as the Solicitation/Recommendation Statement, the Company and Datawatch file annual, quarterly and current reports and other information with the SEC. You may read and copy any reports or other information filed by the Company or Datawatch at the SEC public reference room at 100 F Street, N.E., Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the public reference room. The Company's and Datawatch's filings with the SEC are also available to the public from commercial document-retrieval services and at the SEC's website at www.sec.gov.

Forward Looking Statements

Certain statements either contained in or incorporated by reference into this document are forward-looking statements within the meaning of the safe harbor provisions of the Private Securities Litigation Reform Act of 1995. Such statements involve risks, uncertainties and assumptions. If such risks or uncertainties materialize or such assumptions prove incorrect, the results of the Company and its consolidated subsidiaries could differ materially from those expressed or implied by such forward-looking statements and assumptions. All statements other than statements of historical fact are statements that could be deemed forward-looking statements, including any statements regarding the expected benefits and costs of the Offer, the Merger and the other transactions contemplated by the Merger Agreement; the expected timing of the completion of the Offer and the Merger; the ability of the Company, Purchaser and Datawatch to complete the Offer and the Merger considering the various conditions to the Offer and the Merger, some of which are outside the parties' control, including those conditions related to regulatory approvals; any statements of expectation or belief; the possibility of any termination of the

Merger Agreement; and any statements of assumptions underlying any of the foregoing. The forward-looking statements contained in this document are based on current expectations and assumptions that are subject to risks and uncertainties which may cause actual results to differ materially from the forward-looking statements. Actual results may differ materially from current expectations because of risks associated with uncertainties as to:

- · expected benefits that may not materialize as expected;
- that the Offer and the Merger may not be timely completed, if at all;
- uncertainties as to how many of Datawatch's stockholders will tender their Shares in the Offer;
- the risk that competing offers or acquisition proposals will be made;
- the possibility that various conditions to the consummation of the Offer or the Merger may not be satisfied or waived, including that a
 governmental entity may prohibit, delay or refuse to grant approval for the consummation of the Offer or the Merger;
- the effects of disruption on Datawatch's business and the fact that the announcement and pendency of the transactions described in this release may make it more difficult to establish or maintain relationships with employees, suppliers and other business partners;
- that the parties are unable to successfully implement integration strategies; and
- other risks that are described in the Company's SEC reports, including, but not limited to, the risks described in the Company's Annual Report on Form 10-K for its fiscal year ended December 31, 2017 and Quarterly Reports on Form 10-Q and Datawatch's most recent Annual Report on Form 10-K for its fiscal year ended September 30, 2017 and Quarterly Reports on Form 10-Q.

The reader is cautioned not to unduly rely on these forward-looking statements. The Company expressly disclaims any intent or obligation to update or revise publicly these forward-looking statements except as required by law.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

Exhibit No.	Description		
2.1*	Agreement and Plan of Merger, dated as of November 5, 2018, by and among Altair Engineering Inc., Datawatch Corporation and Dallas Merger Sub, Inc.		
2.2	Form of Tender and Support Agreement, dated as of November 5, 2018, by and among Altair Engineering Inc., Dallas Merger Sub and each of the stockholders named therein.		
10.1	Amendment (to Third Amended and Restated Credit Agreement), dated as of October 31, 2018, by and among the Company, the lenders party thereto and JPMorgan Chase Bank, N.A., as Administrative Agent.		
99.1	Joint Press Release of Altair Engineering Inc. and Datawatch Corporation, dated November 5, 2018.		

^{*} Schedules omitted pursuant to item 601(b)(2) of Regulation S-K. The Company agrees to furnish supplementally a copy of any omitted schedule to the SEC upon request.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: November 5, 2018

Altair Engineering Inc.

By: /s/ Howard Morof

Name: Howard Morof

Title: Chief Financial Officer

AGREEMENT AND PLAN OF MERGER

by and among:

DATAWATCH CORPORATION,

a Delaware corporation;

ALTAIR ENGINEERING INC.,

a Delaware corporation; and

DALLAS MERGER SUB, INC.,

a Delaware corporation

Dated as of November 5, 2018

TABLE OF CONTENTS

Article I T	THE OFFER	Page 2
1.1	The Offer	2
1.2	Company Actions	5
Article II	MERGER TRANSACTION	7
2.1	Merger of Purchaser into the Company	7
2.2	Effect of the Merger	7
2.3	Closing; Effective Time	7
2.4	Merger Without Meeting of Stockholders	7
2.5	Certificate of Incorporation and Bylaws; Directors and Officers	8
2.6	Conversion of Shares	8
2.7	Surrender of Certificates; Stock Transfer Books	9
2.8	Dissenters' Rights	12
2.9	Company Equity Awards	12
2.10	Further Action	13
Article III	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	14
3.1	Due Organization; Subsidiaries, Etc.	14
3.2	Certificate of Incorporation and Bylaws	15
3.3	Capitalization, Etc.	15
3.4	SEC Filings; Financial Statements	17
3.5	Absence of Changes	19
3.6	Title to Assets	20
3.7	Real Property; Equipment	20
3.8	Intellectual Property	20
3.9	Contracts	23
3.10	Liabilities	25
3.11	Compliance with Law	25
3.12	Regulatory Matters	26
3.13	Privacy	26
3.14	Tax Matters	27
3.15	Employee Benefit Plans	30
3.16	Labor Matters	32
3.17	Environmental Matters	33
3.18	Transactions with Affiliates	33
3.19	Legal Proceedings; Orders	33
3.20		34
3.21		34
3.22		35
3 23		35

3.24	Fairness Opinion	36
	Financial Advisor	36
	Disclosure	36
	Rule 14d-10 Matters	36
5.27	Nuic 14d 10 Municis	50
Article IV	REPRESENTATIONS AND WARRANTIES OF PARENT AND PURCHASER	37
4.1	Due Organization	37
	Purchaser	37
	Authority; Binding Nature of Agreement	38
	Non-Contravention; Consents	38
	Disclosure	38
	Funds	39
4.7	Ownership of Company Common Stock	39
	Litigation	39
	Management Agreements	39
	Financial Advisor	40
	Independent Investigation	4(
Article V	CERTAIN COVENANTS OF THE COMPANY	40
5.1	Covenants of the Company	40
5.2	Confidentiality	44
5.3	Access and Investigation	44
5.4	No Solicitation	45
	I ADDITIONAL COVENANTS OF THE PARTIES	50
	Filings and Approvals	50
	Employee Benefits	53
	Indemnification of Officers and Directors	54
	Notification of Certain Matters	56
	Stockholder Litigation	56
	Disclosure	56
	Section 16 Matters	56
	Stock Exchange Delisting; Deregistration	57
	Transfer Taxes Rule 14d-10 Matters	57
		57
	Takeover Statutes	57
	Control of the Company's Business	57
0.13	Non-USRPHC Certificate and Notice	58
Article V	II CONDITIONS PRECEDENT TO THE MERGER	58
	No Restraints	58
	Consummation of Offer	58
·		50
Article V	III TERMINATION	58
8.1	Termination	58
8.2	Effect of Termination	60

8.3	Expenses; Termination Fees	60			
Article IX MISCELLANEOUS PROVISIONS					
9.1	Notices	62			
9.2	Amendment	63			
9.3	Waiver	63			
9.4	No Survival of Representations and Warranties	63			
9.5	Entire Agreement; No Reliance; Counterparts	63			
9.6	Applicable Law; Jurisdiction; Specific Performance; Remedies; Waiver of Jury Trial	64			
9.7	Third Party Beneficiaries	65			
9.8	Assignability	65			
9.9	Cooperation	65			
9.10	Severability	66			
9.11	Obligation of Parent	66			
9.12	Construction	66			
9.13	Non-Parties	67			
Exhibit A	Certain Definitions				

Ex

Exhibit B Certificate of Incorporation of the Surviving Corporation

Conditions of the Offer Annex I

AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger ("**Agreement**") is made and entered into as of November 5, 2018, by and among: ALTAIR ENGINEERING INC., a Delaware corporation ("**Parent**"); DALLAS MERGER SUB, INC., a Delaware corporation and a Subsidiary of Parent ("**Purchaser**"); and DATAWATCH CORPORATION, a Delaware corporation (the "**Company**"). Capitalized terms used in this Agreement are defined in <u>Exhibit A</u>.

Recitals

WHEREAS, upon the terms and subject to the conditions of this Agreement, Purchaser has agreed to commence a cash tender offer (as it may be extended and amended from time to time as permitted under this Agreement, the "**Offer**") to acquire all of the outstanding shares of Company Common Stock ("**Shares**") for thirteen dollars and ten cents (\$13.10) per share of Company Common Stock (such amount, or any different amount per share paid pursuant to the Offer to the extent permitted under this Agreement, being the "**Offer Price**"), net to the seller in cash, without interest and less any applicable Tax withholdings;

WHEREAS, following the consummation of the Offer, upon the terms and conditions set forth herein and in accordance with Section 251(h) of the DGCL, Purchaser will be merged with and into the Company (the "Merger") with the Company as the surviving corporation in the Merger (the "Surviving Corporation"), whereby (except as otherwise provided herein), upon the terms and subject to the conditions of this Agreement, (i) each issued and outstanding Share (other than Shares irrevocably accepted for purchase in the Offer, Shares owned by Parent or Purchaser as of both the commencement of the Offer and the Effective Time, Shares owned by the Company as of the Effective Time, and Dissenting Shares), will be converted into the right to receive the Offer Price, net to the seller in cash, without interest and less any applicable Tax withholdings and (ii) the Company shall become a wholly-owned Subsidiary of Parent as a result of the Merger;

WHEREAS, the Company Board has unanimously (i) determined that this Agreement and the other transactions contemplated herein, including the Offer and the Merger, are advisable and in the best interests of the Company and its stockholders, (ii) agreed that the Merger shall be effected under Section 251(h) and other relevant provisions of the DGCL, (iii) approved the execution, delivery and performance by the Company of this Agreement and the consummation of the transactions contemplated herein, including the Offer and the Merger, in accordance with the DGCL, and (iv) resolved to recommend that the stockholders of the Company accept the Offer and tender their Shares to Purchaser pursuant to the Offer;

WHEREAS, the Board of Directors of Parent has, on the terms and subject to the conditions set forth herein, approved this Agreement, the Offer, the Merger and the other transactions contemplated herein;

WHEREAS, the Board of Directors of Purchaser has determined that, on the terms and subject to the conditions set forth herein, this Agreement, the Offer, the Merger and the other transactions contemplated herein are advisable and in the best interests of Purchaser and its stockholders, and has approved this Agreement, the Offer, the Merger and the other transactions contemplated herein;

WHEREAS, concurrently with the execution and delivery of this Agreement, and as a condition and inducement to Parent's and Purchaser's willingness to enter into this Agreement, Parent and Purchaser have entered into agreements with the directors, executive officers and certain stockholders of the Company pursuant to which, among other things, such persons have, subject to the terms thereof, irrevocably agreed to tender in the Offer the Shares beneficially owned by each of them (each, a "Support Agreement" and collectively, the "Support Agreements"); and

WHEREAS, Parent, Purchaser and the Company acknowledge and agree that the Merger shall be governed by and effected pursuant to Section 251(h) of the DGCL and shall be effected as soon as practicable following the consummation (within the meaning of Section 251(h) of the DGCL) of the Offer upon the terms and subject to the conditions set forth in this Agreement.

NOW, **THEREFORE**, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth below, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Parent, Purchaser and the Company agree as follows:

ARTICLE I

THE OFFER

1.1 The Offer.

(a) Provided that this Agreement shall not have been terminated in accordance with <u>Article VIII</u> and that the Company has complied with its obligations under <u>Section 1.2</u>, as promptly as practicable after the date of this Agreement but in no event more than ten (10) business days after the date of this Agreement (or such later date as the parties may agree), Purchaser shall (and Parent shall cause Purchaser to) commence (within the meaning of Rule 14d-2 under the Exchange Act) the Offer.

(b) Subject to the terms and conditions of this Agreement and to the satisfaction, or waiver (to the extent permitted under this Agreement) by Purchaser, of the conditions set forth in Annex I (collectively, the "Offer Conditions") as of the Expiration Date in accordance with Section 1.1(d), Purchaser shall promptly on or after the Expiration Date accept for payment (such time of acceptance for payment, the "Offer Acceptance Time") and pay (by delivery of funds to the Depository Agent) for all Shares validly tendered and not properly withdrawn pursuant to the Offer that Purchaser becomes obligated to purchase pursuant to the Offer. Parent shall on or prior to the Expiration Date provide or cause to be provided to Purchaser, on a timely basis, the funds necessary to purchase any Shares that Purchaser becomes obligated to purchase pursuant to the Offer. The Company shall register (and shall instruct its transfer agent to register) the transfer of the Shares accepted for payment by Purchaser effective as promptly as practicable after the Offer Acceptance Time.

(c) The Offer shall be made by means of an offer to purchase (the "Offer to Purchase") in accordance with the terms set forth in this Agreement, and with consummation thereof subject to the Minimum Condition and the other Offer Conditions. Purchaser expressly reserves the right to (i) increase the Offer Price, (ii) waive any Offer Condition other than the Minimum Condition and (iii) make any other changes in the terms and conditions of the Offer not inconsistent with the terms of this Agreement; provided, however, that unless otherwise provided by this Agreement, without the prior written consent of the Company (which consent may be withheld in the sole discretion of the Company), Purchaser shall not (A) decrease the Offer Price other than in accordance with Section 1.1(f), (B) change the form of consideration payable in the Offer, (C) decrease the maximum number of Shares sought to be purchased in the Offer, (D) impose conditions to the Offer in addition to the Offer Conditions, (E) amend, modify or supplement any of the terms of the Offer or the Offer Conditions in a manner that adversely affects holders of Shares, (F) amend, modify or waive the Minimum Condition, or (G) extend or otherwise change the Expiration Date in a manner other than as required or permitted by this Agreement. The Offer may not be terminated prior to the Expiration Date, unless this Agreement is terminated in accordance with Article VIII.

(d) Unless extended pursuant to and in accordance with the terms of this Agreement, the Offer shall expire at one minute after 11:59 p.m. (Boston time) on the date that is twenty (20) business days (calculated in accordance with Rule 14d-1(g)(3) under the Exchange Act) following the commencement (within the meaning of Rule 14d-2 under the Exchange Act) of the Offer (the "Initial Expiration Date") or, in the event the Initial Expiration Date has been extended pursuant to and in accordance with this Agreement, the date and time to which the Offer has been so extended (the Initial Expiration Date, or such later date and time to which the Initial Expiration Date has been extended pursuant to and in accordance with this Agreement, is referred to as the "Expiration Date").

(e) Notwithstanding any provision to the contrary contained in this Agreement but subject to the parties' respective rights to terminate the Agreement pursuant to Article VIII, (i) if, as of the scheduled Expiration Date, any Offer Condition is not satisfied and has not been waived, Purchaser may, in its discretion (and without the consent of the Company or any other Person), extend the Offer on one or more occasions, for an additional period of up to 10 business days per extension, to permit such Offer Condition to be satisfied, (ii) Purchaser shall extend the Offer from time to time for: (A) any period required by any Law, any interpretation or position of the SEC or the staff thereof or any rules and regulations of the NASDAQ Stock Market ("NASDAQ") applicable to the Offer, and (B) periods of up to 10 business days per extension, until any waiting period (and any extension thereof) applicable to the consummation of the Offer under the HSR Act or any applicable foreign Antitrust Law shall have expired or been terminated, (iii) if, as of the scheduled Expiration Date, any Offer Condition is not satisfied and has not been waived, at the request of the Company, Purchaser shall extend the Offer on one or more occasions for an additional period of up to 10 business days per extension, to permit such Offer Condition to be satisfied, and (iv) if the Company has delivered a Change of Status Notice to Parent, and the Company so requests, Purchaser shall extend the Offer on one or more occasions so that the Expiration Date does not occur earlier than two (2) business days immediately after the expiration of the Notice Period or Intervening Event Notice Period, as applicable; provided, however, that in no event shall Purchaser: (1) be required to extend the Offer beyond the earlier to occur of (x) the valid termination of this Agreement in

compliance with <u>Section 8.1</u> and (y) the End Date (such earlier date, the "Extension Deadline"), (2) be permitted to extend the Offer beyond the Extension Deadline without the prior written consent of the Company, or (3) be permitted, without the prior written consent of the Company, which shall not be unreasonably withheld, or required by the Company, to extend the Offer by periods exceeding an aggregate of 20 business days in the event that (x) the Company has not delivered to Parent a Change of Status Notice and (y) each Offer Condition (other than the Minimum Condition) shall have been satisfied or waived as of any then scheduled Expiration Date and the Minimum Condition shall not have been satisfied as of such scheduled Expiration Date. Purchaser shall not terminate the Offer prior to any scheduled Expiration Date without the prior written consent of the Company except in the event that this Agreement is terminated pursuant to <u>Section 8.1</u>.

- (f) The Offer Price shall be adjusted appropriately and proportionately to reflect the effect of any stock split, reverse stock split, stock dividend (including any dividend or distribution of securities convertible into Company Common Stock), reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to Company Common Stock occurring on or after the date hereof and at or prior to the Offer Acceptance Time, and such adjustment to the Offer Price shall provide to the holders of Company Common Stock the same economic effect as contemplated by this Agreement prior to such action; provided that nothing in this Section 1.1(f) shall be construed to permit the Company to take any action with respect to its securities that is prohibited by the terms of this Agreement.
- (g) Nothing in this <u>Section 1.1</u> shall be deemed to impair, limit or otherwise restrict the right of the Company, Purchaser or Parent to terminate this Agreement pursuant to <u>Section 8.1</u>. In the event that this Agreement is terminated pursuant to the terms hereof, Purchaser shall (and Parent shall cause Purchaser to) as promptly as practicable irrevocably and unconditionally terminate the Offer, shall not acquire any Shares pursuant to the Offer and shall cause any depository acting on behalf of Purchaser to return, in accordance with applicable Law, all tendered Shares to the registered holders thereof.
- (h) As promptly as practicable on the date of commencement of the Offer (within the meaning of Rule 14d-2 under the Exchange Act), Parent and Purchaser shall (i) file with the SEC a tender offer statement on Schedule TO with respect to the Offer (together with all amendments and supplements thereto and including exhibits thereto, the "Schedule TO") that will contain or incorporate by reference the Offer to Purchase and form of the related letter of transmittal and (ii) cause the Offer to Purchase and related documents to be disseminated to holders of Shares, in each case as and to the extent required by applicable Law. Parent and Purchaser agree that they shall cause the Schedule TO and all exhibits, amendments or supplements thereto (which together constitute the "Offer Documents") filed by either Parent or Purchaser with the SEC to comply in all material respects with the Exchange Act and the rules and regulations thereunder and other applicable Laws. Each of Parent, Purchaser and the Company agrees to promptly correct any information provided by it for use in the Offer Documents if and to the extent that such information shall have become false or misleading in any material respect, and Parent further agrees to use all reasonable efforts to promptly cause the Offer Documents as so corrected to be filed with the SEC and to promptly be disseminated to holders of Shares, in each case as and to the extent required by applicable Law. The Company

shall promptly furnish in writing or otherwise make available to Parent, Purchaser or Parent's legal counsel in writing any information concerning the Acquired Entities and, to the extent reasonably available to the Company, their stockholders or other equity owners that is required in connection with any action contemplated by this Section 1.1(h). The Company and its counsel shall be given reasonable opportunity to review and comment on the Offer Documents prior to the filing thereof with the SEC, and Parent and Purchaser shall give due consideration to any such comments. Parent and Purchaser agree to provide the Company and its counsel with any comments (including a summary of any oral comments) that Parent, Purchaser or their counsel may receive from the SEC or its staff with respect to the Offer Documents promptly after receipt of such comments (and shall give the Company prompt telephonic notice of any material discussions with the SEC or its staff), and the Company and its counsel shall be given reasonable opportunity to review and comment on any response to such comments prior to the submission thereof. Parent and Purchaser shall give due consideration to any such comments, and, to the extent permissible and, in Parent's reasonable judgment, feasible under the circumstances, shall allow the Company and/or its counsel to participate in any meetings with the SEC or its staff with respect thereto. Each of Parent and Purchaser shall respond promptly to any comments of the SEC or its staff with respect to the Offer.

(i) For purposes of this Agreement, and the Offer, unless otherwise mutually agreed to by the Company and Purchaser, any Shares subject to notices of guaranteed delivery shall be deemed not to be validly tendered into the Offer unless and until the Shares underlying such notices of guaranteed delivery are delivered to Purchaser or to an agent of Purchaser.

1.2 Company Actions.

(a) The Company hereby approves of and consents to the Offer and represents that the Company Board, at a meeting duly called and held, has unanimously (i) determined that this Agreement and the Offer, the Merger and the other transactions contemplated herein are in the best interests of the Company's stockholders, (ii) approved and declared advisable this Agreement, the Offer, the Merger and the other transactions contemplated herein in accordance with the requirements of the DGCL and (iii) resolved to recommend that stockholders of the Company accept the Offer and tender their Shares pursuant to the Offer (the actions taken by the Company Board set forth in clauses (i), (ii) and (iii), the "Company Board Recommendation"). Subject to Section 5.4, the Company hereby consents to the inclusion of a description of the Company Board Recommendation (including the reasons therefor) in the Offer Documents.

(b) As promptly as practicable subsequent to the commencement of the Offer, following the filing of the Schedule TO (and in any event within two business days after the filing of the Schedule TO), the Company shall file with the SEC and disseminate to holders of Shares, in each case as and to the extent required by applicable Law, a Tender Offer Solicitation/Recommendation Statement on Schedule 14D-9 (together with any exhibits, amendments or supplements thereto, the "Schedule 14D-9") that, subject to Section 5.4, shall contain the Company Board Recommendation. Parent and Purchaser shall promptly furnish to the Company in writing all information concerning Parent and Purchaser required by applicable Law for inclusion in the Schedule 14D-9. The Company agrees that it will cause the Schedule 14D-9

to comply in all material respects with the Exchange Act and other applicable Laws. The Schedule 14D-9 shall also contain the notice of appraisal rights required to be delivered by the Company under Section 262(d)(2) of the DGCL in connection with a merger effected pursuant to Section 251(h) of the DGCL at the time the Company first files the Schedule 14D-9 with the SEC. The Company shall set the record date for the Company's stockholders to receive such notice of appraisal rights as the same date as the Stockholder List Date and shall disseminate the Schedule 14D-9 including such notice of appraisal rights to the Company's stockholders to the extent required by Section 262(d) of the DGCL. The Company agrees to respond promptly to any comments of the SEC or its staff with respect to the Schedule 14D-9 and each of Parent, Purchaser and the Company agrees to promptly correct any information provided by it for use in the Schedule 14D-9 if and to the extent that such information shall have become false or misleading in any material respect, and the Company further agrees to use all reasonable efforts to cause the Schedule 14D-9 as so corrected to promptly be filed with the SEC and to promptly be disseminated to holders of Shares, in each case as and to the extent required by applicable Law. Parent and its counsel shall be given reasonable opportunity to review and comment on the Schedule 14D-9 and any amendment thereto prior to the filing thereof with the SEC, and the Company shall give due consideration to any such comments. The Company agrees to provide Parent and its counsel with any comments (including a summary of any oral comments) the Company or its counsel may receive from the SEC or its staff with respect to the Schedule 14D-9 promptly after receipt of such comments (and shall give Parent prompt telephonic notice of any material discussions with the SEC or its staff), and the Parent and its counsel shall be given reasonable opportunity to review and comment on any response to such comments prior to the submission thereof. The Company shall give due consideration to any such comments, and, to the extent permissible and, in the Company's reasonable judgment, feasible under the circumstances, shall allow Parent and/or its counsel to participate in any meetings with the SEC or its staff with respect thereto.

(c) In connection with the Offer, the Company shall (or shall cause its transfer agent to) promptly (and in sufficient time to enable Purchaser to commence the Offer within the time period specified in Section 1.1(a)) furnish Parent with a list of its stockholders, mailing labels and any available listing or computer file containing the names and addresses of all record holders of Shares and lists of securities positions of Shares held in stock depositories, in each case accurate and as of the most recent practicable date, and shall provide to Parent such additional information (including updated lists of stockholders, mailing labels and lists of securities positions) and such other assistance as Parent may reasonably request in connection with the Offer (the date of the list used to determine the Persons to whom the Offer Documents and Schedule 14D-9 are first disseminated, the "Stockholder List Date"). Subject to the requirements of applicable Law and provided that (i) the Company shall not have made an Adverse Change Recommendation and (ii) this Agreement shall not otherwise have been terminated, the Company shall use its commercially reasonable efforts to make solicitations and recommendations to the record holders and beneficial owners of Shares for purposes of causing the Minimum Condition to be satisfied. Subject to the requirements of applicable Law, and except for such steps as are necessary to disseminate the Offer Documents and any other documents necessary to consummate the transactions contemplated herein, Parent and Purchaser and their agents shall hold in confidence the information contained in any such labels, listings and files, with such being subject to the Confidentiality Agreement, shall use such information only in connection with the Offer and the Merger and, if this Agreement shall be terminated, shall, upon request by the Company, deliver, and shall use their reasonable efforts to cause their agents to deliver, to the Company or destroy all copies and any extracts or summaries from such

ARTICLE II

MERGER TRANSACTION

- 2.1 **Merger of Purchaser into the Company**. Upon the terms and subject to the conditions set forth in this Agreement and in accordance with Section 251(h) of the DGCL, at the Effective Time, the Company and Purchaser shall consummate the Merger, whereby Purchaser shall be merged with and into the Company and the separate corporate existence of Purchaser shall thereupon cease. The Company shall be the surviving corporation in the Merger, and the separate corporate existence of the Company, with all its rights, privileges, powers and franchises, shall continue unaffected by the Merger, except as set forth in this <u>Article II</u>.
 - 2.2 Effect of the Merger. The Merger shall have the effects set forth in this Agreement and in the applicable provisions of the DGCL.
- 2.3 Closing; Effective Time. Unless this Agreement shall have been terminated pursuant to Article VIII, and unless otherwise mutually agreed to in writing among the Company, Parent and Purchaser, the consummation of the Merger (the "Closing") shall take place at the offices of Choate, Hall & Stewart LLP, Two International Place, Boston, Massachusetts 02110, at 9:00 a.m. local time as soon as practicable following the Offer Acceptance Time, subject to the satisfaction or, to the extent permitted by applicable Law, the waiver of the conditions set forth in Article VII by the parties entitled thereto, but in any event no later than the first (1st) business day after the satisfaction or such waiver of the last to be satisfied or waived of the conditions set forth in Article VII (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions) (the date on which the Closing occurs, the "Closing Date"). Subject to the provisions of this Agreement, a certificate of merger satisfying the applicable requirements of the DGCL shall be duly executed by the Company and, concurrently with or as soon as practicable following the consummation of the Closing, delivered by Parent or Purchaser to the Secretary of State of the State of Delaware for filing. The Merger shall become effective upon the date and time of the filing of such certificate of merger with the Secretary of State of the State of Delaware or such later date and time as is agreed upon in writing by the parties hereto and specified in the certificate of merger (such date and time, the "Effective Time"). From and after the Effective Time, the Surviving Corporation shall possess all the rights, privileges, powers and franchises and be subject to all of the restrictions, disabilities and duties of the Company and the Purchaser, all as provided in the DGCL.
- 2.4 **Merger Without Meeting of Stockholders**. The Merger shall be governed by Section 251(h) of the DGCL. The parties hereto agree to take all necessary and appropriate action to cause the Merger to become effective as soon as practicable following the consummation (within the meaning of Section 251(h) of the DGCL) of the Offer (but in any event within one (1) business day of the date on which the Offer Acceptance Time occurs), without a meeting of stockholders of the Company in accordance with Section 251(h) of the DGCL.

- 2.5 **Certificate of Incorporation and Bylaws; Directors and Officers**. The Company shall take all necessary action to cause the members of the Company Board immediately prior to the Effective Time to resign effective immediately prior to the Effective Time. Unless otherwise agreed to by the Company and Parent prior to the Effective Time, at the Effective Time:
- (a) the certificate of incorporation of the Company immediately prior to the Effective Time shall be, by virtue of the Merger and without any action on the part of Parent, Purchaser, the Company or the holders of Company Common Stock, amended and restated in its entirety as set forth in **Exhibit B** hereto and, as so amended and restated, shall be the certificate of incorporation of the Surviving Corporation until thereafter amended in accordance with such certificate of incorporation and applicable Law;
- (b) subject to <u>Section 6.3(b)</u>, the Bylaws of the Surviving Corporation shall be amended and restated as of the Effective Time to conform to the Bylaws of Purchaser as in effect immediately prior to the Effective Time, except that references to the name of Purchaser shall be replaced with references to the name reflected in **Exhibit B** hereto; and
- (c) the director of Purchaser immediately prior to the Effective Time shall be the initial director of the Surviving Corporation, to hold office in accordance with the charter and bylaws of the Surviving Corporation, and the officers of Purchaser immediately prior to the Effective Time shall be the initial officers of the Surviving Corporation, in each case until their respective successors are duly elected and qualified or until such director's or officer's earlier death, resignation or removal.

2.6 Conversion of Shares.

- (a) At the Effective Time, as a result of the Merger and without any further action on the part of the Company, Parent, Purchaser or the holder of any capital stock of Parent, Purchaser or the Company:
- (i) Each Share issued and outstanding immediately prior to the Effective Time, other than (A) Shares that are cancelled pursuant to clause (iv) or (v) below and (B) Shares that are owned by stockholders ("Dissenting Stockholders") who have properly demanded in compliance in all respects with Section 262 of the DGCL and not withdrawn a demand for, or lost their right to, appraisal pursuant to Section 262 of the DGCL with respect to such Shares (shares described in the foregoing clauses (A) and (B), collectively, "Excluded Shares"), shall be converted into the right to receive an amount in cash equal to the Offer Price (the "Merger Consideration") without interest and less any applicable Tax withholdings. At the Effective Time, all of the Shares shall cease to be outstanding, shall be cancelled and shall cease to exist, and each certificate (a "Certificate") formerly representing any of the Shares (other than Excluded Shares) and each non-certificated Share represented by book entry (a "Book Entry Share") (other than Excluded Shares) shall thereafter represent only the right to receive the Merger Consideration, without interest and less any applicable Tax withholdings, and each Certificate formerly representing Shares or Book Entry Shares owned by Dissenting Stockholders shall thereafter only represent such rights as are provided by Section 262 of the DGCL to a holder thereof.

- (ii) Each Dissenting Share shall, by virtue of the Merger and without any action on the part of the holder thereof, cease to be outstanding, shall be cancelled without payment of any consideration therefor and shall cease to exist, subject to any rights the holder of such Dissenting Share may have under **Section 2.8**.
- (iii) At the Effective Time, each share of common stock, par value \$0.01 per share, of Purchaser issued and outstanding immediately prior to the Effective Time shall be converted into one share of common stock, par value \$0.01 per share, of the Surviving Corporation.
- (iv) Any Shares then owned by the Company or any Shares owned both at the commencement of the Offer and at the Effective Time by any direct or indirect wholly owned Subsidiary of the Company shall be canceled and shall cease to exist, and no consideration shall be delivered in exchange therefor.
- (v) Any Shares (a) irrevocably accepted for purchase in the Offer or (b) owned both at the commencement of the Offer and at the Effective Time by Parent, Purchaser or any other direct or indirect wholly owned Subsidiary of Parent, shall be canceled and shall cease to exist, and no consideration shall be delivered in exchange therefor.
- (b) Without duplication of any adjustment made pursuant to <u>Section 1.1(f)</u>, the Merger Consideration shall be adjusted appropriately and proportionately to reflect the effect of any stock split, reverse stock split, stock dividend (including any dividend or distribution of securities convertible into Company Common Stock), reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to Company Common Stock occurring on or after the date hereof and at or prior to the Effective Time, and such adjustment to the Merger Consideration shall provide to the holders of Company Common Stock the same economic effect as contemplated by this Agreement prior to such action; <u>provided</u> that nothing in this <u>Section 2.6(b)</u> shall be construed to permit the Company to take any action with respect to its securities that is prohibited by the terms of this Agreement. For the avoidance of doubt, the Merger Consideration, as adjusted pursuant to this <u>Section 2.6(b)</u>, shall equal the Offer Price, as adjusted pursuant to <u>Section 1.1(f)</u>.

2.7 Surrender of Certificates; Stock Transfer Books.

(a) Prior to the Offer Acceptance Time, Parent shall designate a single bank or trust company reasonably acceptable to the Company to act both as agent (in such capacity, the "**Depository Agent**") for the holders of Shares to receive the funds to which holders of such Shares shall become entitled pursuant to **Section 1.1(b)** and to act as agent (in such capacity, the "**Paying Agent**") for the holders of Shares to receive the funds to which holders of such Shares shall become entitled pursuant to **Section 2.6**. The agreement or agreements pursuant to which Parent shall appoint the Depository Agent and Paying Agent shall be in form and substance reasonably acceptable to the Company. At or prior to the Offer

Acceptance Time, Parent shall deposit, or shall take all steps necessary to enable and cause Purchaser to deposit, with the Depository Agent cash sufficient to make payment of the cash consideration payable pursuant to Section 1.1(b) and with the Paying Agent cash sufficient to make payment of the cash consideration payable pursuant to Section 2.6 (together, the "Payment Fund"). The Payment Fund shall be invested by the Paying Agent as directed by Parent; provided, that (i) no such investment or losses thereon shall relieve Parent from making the payments required by Article I or this Article II or affect the amount of the aggregate Offer Price or the Merger Consideration payable in respect of the Shares and following any losses, or if for any other reasons the Payment Fund shall not be sufficient to make prompt payment of the Offer Price or Merger Consideration (including as a result of Dissenting Shares losing their status as such), Parent shall promptly provide additional funds to the Paying Agent in the amount of any such losses or any other amount necessary so that the Payment Fund is sufficient to make prompt payment of the aggregate Offer Price and Merger Consideration, (ii) no such investment shall have maturities that could prevent or delay payments to be made pursuant to this Agreement, and (iii) such investments shall be in short-term obligations of the United States of America with maturities of no more than three (3) months or guaranteed by, and backed by the full faith and credit of, the United States of America. Any and all interest or other amounts earned with respect to such funds shall become part of the Payment Fund, and any amounts in excess of the amounts payable in connection with the Offer or under Section 2.6 shall be promptly returned to either Parent or the Surviving Corporation. The Payment Fund shall not be used for any purpose not contemplated by this Section 2.7. The Surviving Corporation shall (and Parent shall cause the Surviving Corporation to) pay all charges and expenses, incl

(b) Promptly after the Effective Time (but in any event within five (5) business days after the Effective Time), Parent and the Surviving Corporation shall cause to be mailed to each Person who was, at the Effective Time, a holder of record of Shares entitled to receive the Merger Consideration pursuant to Section 2.6 (i) a form of letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon proper delivery of the Certificates to the Paying Agent) and (ii) instructions for use in effecting the surrender of the Certificates or non-certificated Shares represented by Book Entry Shares pursuant to such letter of transmittal. Upon surrender to the Paying Agent of Certificates or Book Entry Shares, together with such letter of transmittal, duly completed and validly executed in accordance with the instructions thereto, and such other documents as may be required pursuant to such instructions, the holder of such Certificates or Book Entry Shares shall be entitled to receive in exchange therefor the Merger Consideration for each Share formerly evidenced by such Certificates or Book Entry Shares, and such Certificates and Book Entry Shares shall then be cancelled. No interest shall accrue or be paid on the Merger Consideration payable upon the surrender of any Certificates or Book Entry Shares for the benefit of the holder thereof. If the payment of any Merger Consideration is to be made to a Person other than the Person in whose name the surrendered Certificates formerly evidencing the Shares is registered on the stock transfer books of the Company, it shall be a condition of payment that the Certificate so surrendered shall be endorsed properly or otherwise be in proper form for transfer and that the Person requesting such payment shall have paid all transfer and other similar Taxes required by reason of the payment of the Merger Consideration to a Person other than the registered holder of the Certificate surrendered, or shall have established to the satisfaction of the Surviving Corporation, Parent and the Paying Agent that such Taxes either have been paid or are not applicable. Payment of the applicable Merger Consideration with respect to Book Entry Shares shall only be made to the Person in whose name such Book Entry Shares are registered.

- (c) At any time following the one year anniversary of the Effective Time, the Surviving Corporation shall be entitled to require the Paying Agent to deliver to it any funds which had been made available to the Paying Agent and not disbursed to holders of Certificates or Book Entry Shares (including all interest and other income received by the Paying Agent in respect of all funds made available to it), and, thereafter, such holders shall be entitled to look to the Surviving Corporation (subject to abandoned property, escheat and other similar Law) only as general creditors thereof with respect to any Merger Consideration that may be payable upon due surrender of the Certificates or Book Entry Shares held by them.

 Notwithstanding the foregoing, neither Parent, Purchaser, the Surviving Corporation nor the Paying Agent shall be liable to any holder of Certificates or Book Entry Shares for any Merger Consideration delivered in respect of such share to a public official pursuant to any abandoned property, escheat or other similar Law. Any amounts remaining unclaimed by such holders immediately prior to such time at which such amounts would otherwise escheat to or become property of any Governmental Body shall become, to the extent permitted by applicable Law, the property of the Surviving Corporation or its designee, free and clear of all claims or interest of any Person previously entitled thereto.
- (d) No dividends or other distributions with respect to capital stock of the Surviving Corporation with a record date on or after the Effective Time shall be paid to the holder of any unsurrendered Certificates or Book Entry Shares.
- (e) At the close of business on the day of the Effective Time, the stock transfer books of the Company with respect to Shares shall be closed and thereafter there shall be no further registration of transfers of Shares on the records of the Company. From and after the Effective Time, the holders of Shares outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such Shares except as otherwise provided herein or by applicable Law. If, after the Effective Time, any Certificate or Book Entry Share is presented to the Surviving Corporation, Parent or the Paying Agent for transfer in accordance with this <u>Section 2.7</u>, it shall be cancelled and exchanged for the cash amount in immediately available funds to which the holder thereof is entitled pursuant to this <u>Section 2.7</u>. No Merger Consideration shall be paid to the holder of any unsurrendered Certificate or Book Entry Share until the surrender of such Certificate or Book Entry Share in accordance with this <u>Section 2.7</u>.
- (f) Notwithstanding anything to the contrary herein, the Surviving Corporation, Parent and Purchaser shall be entitled to deduct and withhold (or cause the Paying Agent to deduct and withhold) from the Offer Price or Merger Consideration payable to any holder of Shares such amounts as it is required by any Law to deduct and withhold with respect to Taxes and shall remit such amounts to the appropriate Governmental Body in accordance with applicable Law. To the extent that amounts are so withheld and paid to the appropriate Governmental Body, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the holder of Shares in respect of which such deduction and withholding was made.

(g) In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an effective affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed, which shall include an agreement to indemnify and defend and hold harmless Parent, the Surviving Corporation and the Paying Agent from and against any and all costs, claims, losses, judgments, damages, counsel fees, expenses and liabilities whatsoever which each may suffer, sustain or incur in connection with the failure of any statement, representation or warranty set forth in such affidavit, any payment for or transfer, exchange or delivery of such Certificate and such Person's inability to locate such Certificate, and, if required by Parent, the posting by such Person of a bond in customary amount as indemnity against any claim that may be made against it with respect to such Certificate, the Paying Agent will issue in exchange for such lost, stolen or destroyed Certificate a check in the amount (after giving effect to any required Tax withholdings as provided in Section 2.7(f)) equal to the number of Shares represented by such lost, stolen or destroyed Certificate that have been surrendered multiplied by the per Share Merger Consideration.

2.8 **Dissenters' Rights**. Notwithstanding anything in this Agreement to the contrary, Shares outstanding immediately prior to the Effective Time and held by a holder who is entitled to demand and properly demands and perfects appraisal rights for such Shares in accordance with, and who complies in all respects with, Section 262 of the DGCL and who, as of the Effective Time, has neither effectively withdrawn nor lost such holder's appraisal rights under the DGCL (the "**Dissenting Shares**") shall not be converted into the right to receive Merger Consideration and shall entitle such holder only to such rights as are provided by Section 262 of the DGCL, unless such holder fails to perfect or waives, withdraws or otherwise loses such holder's right to appraisal of its Shares. If after the Effective Time such holder fails to perfect or waives, withdraws or loses such holder's right to appraisal with respect to any Dissenting Shares held by such holder, each such Dissenting Share shall be treated as if it had never been a Dissenting Share and been converted as of the Effective Time into a right to receive the Merger Consideration without any interest thereon (less any amounts entitled to be deducted or withheld pursuant to Section 2.7(f)). The Company shall give Parent prompt notice of any demands received by the Company for appraisal of Shares, withdrawals of such demands and any other instruments or documents served pursuant to Section 262 of the DGCL, in each case prior to the Effective Time, and Parent shall have the right to direct and participate in all negotiations and proceedings with respect to such demands. The Company shall not, without the prior written consent of Parent, make any payment with respect to, or settle or offer to settle, any such demands for appraisal or offer to settle or settle any such demands, approve any withdrawal of such demands or agree or commit to do any of the foregoing.

2.9 Company Equity Awards.

(a) At the Effective Time, each outstanding Company Option, shall, automatically and without any required action on the part of the holder thereof, be canceled and instead represent solely the right to receive, in accordance with the terms of this Agreement and the Company Equity Plan, the applicable Option Consideration with respect to such Company Option (which may be, and shall not be less than, zero). The cash payment, if any, contemplated by this Sections 2.9(a) in cancellation and settlement of each outstanding Company Option (to the extent unexercised as of the Effective Time) shall be payable by Parent in a lump sum (net of any withholding Taxes required to be deducted and withheld by applicable Law) through the Company's payroll system or by check no later than the first payroll period after the Effective Time.

(b) At the Effective Time, each Company RSU that is outstanding and vested immediately prior to the Effective Time (after giving effect to any vesting that is contingent upon the completion of the Merger) shall, automatically and without any required action on the part of the holder thereof, be canceled and instead represent solely the right to receive, in accordance with the terms of this Agreement and the Company Equity Plan, the Merger Consideration with respect to the Shares that would otherwise have been issuable upon settlement of the vested portion of such Company RSU (less applicable Tax withholdings). The payment, if any, contemplated by this Section 2.9(b) with respect to each outstanding Company RSU to the extent vested as of the Effective Time (after giving effect to any vesting that is contingent upon the completion of the Merger) shall be payable (without any crediting of interest for the period from the Effective Time through the date of payment) by Parent in a lump sum through the Company's payroll system or by check on the first regular payroll date following the Effective Time, subject to applicable Tax withholding requirements.

(c) At the Effective Time, each Company RSU that is outstanding and unvested immediately prior to the Effective Time (after giving effect to any vesting that is contingent upon the completion of the Merger) shall be converted into and become a substitute right or award to receive (without interest), subject to the terms and conditions of this Section 2.9(c), an amount in cash (less applicable Tax withholdings), equal to the product of (i) the Offer Price and (ii) the number of Shares underlying the unvested portion of such Company RSU. The payment, if any, contemplated by this Section 2.9(c) with respect to such substitute right or award with respect to each outstanding Company RSU to the extent unvested as of the Effective Time (after giving effect to any vesting that is contingent upon the completion of the Merger) shall be payable (without any crediting of interest for the period from the Effective Time through the date of payment) by Parent in a lump sum through the Company's payroll system or by check on or after the date that such Company RSU would have otherwise vested in accordance with its terms (and in any event not later than the earlier of (1) December 31st of the year during which such vesting date occurs or (2) the first regular payroll date following the date of termination for any such employee) but only if such conditions to vesting are satisfied prior to such vesting date; provided that if the employment or service of such grantee of such Company RSU is terminated by the Surviving Corporation or its affiliates prior to such vesting date under conditions that would have accelerated the vesting of such Company RSU had it remained outstanding, then such vesting shall be accelerated as of the date of such termination of employment or service.

2.10 **Further Action**. If, at any time after the Effective Time, any further action is reasonably determined by Parent to be necessary or desirable to carry out the purposes of this Agreement or to vest the Surviving Corporation with full right, title and possession of and to all rights and property of Purchaser and the Company, the officers and directors of the Surviving Corporation and Parent shall be fully authorized (in the name of Purchaser, in the name of the Company and otherwise) to take such action.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as (a) disclosed in the Company SEC Documents filed or furnished with the SEC and publicly available since October 1, 2017 and not less than five (5) business days prior to the date of this Agreement (without giving effect to any amendment to any such Company SEC Report filed on or after the fifth (5th) business day prior to the date of this Agreement), other than information that is (i) contained in the risk factors sections of such Company SEC Documents, (ii) in any forward-looking statements in such Company SEC Documents, or (iii) cautionary, predictive or forward-looking in nature, contained in such Company SEC Documents (it being agreed and understood that nothing disclosed in the Company SEC Documents shall be deemed to modify **Section 3.3**); or (b) set forth in the Company Disclosure Letter (it being understood that any information set forth in one section or subsection of the Company Disclosure Letter shall be deemed to apply to and qualify the section or subsection of this Agreement to which it corresponds and each other section or subsection of this Agreement to the extent that it is reasonably apparent on its face that such information is relevant to such other section or subsection), the Company hereby represents and warrants to Parent and Purchaser as follows:

3.1 Due Organization; Subsidiaries, Etc.

- (a) The Company is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware and has all corporate power and authority necessary to own, lease and operate all of the properties and assets that it purports to own, lease and operate and to carry on its business as now conducted. The Company is duly licensed or qualified to do business as a foreign corporation and is in good standing in each jurisdiction where the nature of the business conducted by it or the character or location of the properties and assets owned, leased or operated by it makes such licensing or qualification necessary, except for those jurisdictions where failure to be so licensed, qualified or in good standing has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (b) <u>Section 3.1(b)</u> of the Company Disclosure Letter sets forth a true and complete list of each Subsidiary of the Company and indicates its jurisdiction of organization. No Acquired Entities own any capital stock of, or any equity interest of, or any equity interest of any nature in, any other Entity, other than the Entities identified in <u>Section 3.1(b)</u> of the Company Disclosure Letter. If any Entity is identified in <u>Section 3.1(b)</u> of the Company Disclosure Letter, <u>Section 3.1(b)</u> of the Company Disclosure Letter accurately describes the capital stock or other equity interest that the Company owns in such Entity.
- (c) Except as set forth in <u>Section 3.1(c)</u> of the Company Disclosure Letter, each Subsidiary of the Company is a corporation or other business entity duly incorporated or organized (as applicable), validly existing and, where applicable, in good standing (except for, in the case of good standing, any jurisdiction that does not recognize such concept) under the laws of its jurisdiction of incorporation or organization, has all corporate or other organizational power and authority necessary to own, lease and operate all of the properties

and assets that it purports to own, lease and operate and to carry on its business as now conducted, is duly licensed or qualified to do business as a foreign Entity and is in good standing in each jurisdiction where the nature of the business conducted by it or the character or location of the properties and assets owned, leased or operated by it makes such licensing or qualification necessary, except for such failures to be so organized, qualified, licensed or in good standing that have not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

3.2 **Certificate of Incorporation and Bylaws**. The Company has delivered or made available to Parent or Parent's Representatives prior to the date of this Agreement accurate and complete copies of the certificate of incorporation, bylaws and other charter and organizational documents of each of the Acquired Entities, including all amendments thereto, as in effect on the date hereof. The Acquired Entities' certificates of incorporation, bylaws or other charter and organizational documents so delivered are in full force and effect and no Acquired Entity is in violation of the provisions thereof.

3.3 Capitalization, Etc.

- (a) The authorized capital stock of the Company consists of: (i) 20,000,000 Shares, of which 12,736,747 Shares were issued and outstanding as of the close of business on October 31, 2018 and 14,246 Shares were held in the treasury of the Company on October 31, 2018 and (ii) 1,000,000 shares of Company Preferred Stock, of which no shares are issued or outstanding.
- (b) Section 3.3 of the Company Disclosure Letter set forth a true and accurate list of: (i) the number of shares of Company Common Stock issued under the Company Equity Plan, (ii) the number of shares of Company Common Stock subject to outstanding Company Options (with each such holder named therein), (iii) the number of shares of Company Common Stock subject to outstanding Company RSUs held by each holder of such Company RSUs (with each such holder named therein), (iv) the number of shares of Company Common Stock subject to vested outstanding Company Options held by each holder of such Company Options (with each such holder named therein) and the number of shares of Company Common Stock subject to Company Options (with each such holder named therein), (v) the number of shares of Company Common Stock subject to vested outstanding Company RSUs held by each holder of such Company RSUs (with each such holder named therein) and the number of shares of Company Common Stock subject to Company RSUs that will vest as a result of the Merger held by each holder of such Company RSUs (with each such holder named therein), (vi) the exercise price of each outstanding Company Option described therein, (vii) the number of shares of Company Common Stock reserved for future issuance under the Company Equity Plan and (viii) the number of shares of Company Common Stock option agreements evidencing outstanding Company Options and (C) all forms of agreements evidencing outstanding Company RSUs.

(c) Except as set forth in Section 3.3 of the Company Disclosure Letter or as reserved for future grants under the Company Equity Plan, as of the date of this Agreement, (i) there are no equity securities of any class of the Company, or any security exchangeable into or exercisable for such equity securities, issued, reserved for issuance or outstanding, (ii) there are no options, warrants, equity securities, calls, rights or agreements to which the Company or any of its Subsidiaries is a party or by which the Company or any of its Subsidiaries is bound obligating the Company or any of its Subsidiaries to issue, exchange, transfer, deliver or sell, or cause to be issued, exchanged, transferred, delivered or sold, additional shares of capital stock or other equity interests of the Company or any security or rights convertible into or exchangeable or exercisable for any such shares or other equity interests, or obligating the Company or any of its Subsidiaries to grant, extend, accelerate the vesting of, otherwise modify or amend or enter into any such option, warrant, equity security, call, right or agreement and (iii) there are no outstanding Company-issued contingent value rights or rights that are derivative of, or provide economic benefits based, directly or indirectly, on the value or price of, any capital stock or voting securities of the Company. The Company does not have any outstanding stock appreciation rights, phantom stock, performance based rights or similar rights or obligations. Except as contemplated by this Agreement or described in Section 3.3 of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries is a party to or is bound by any agreements with respect to the voting (including voting trusts and proxies) or sale or transfer of any shares of capital stock or other equity interests of the Company. Except as contemplated by this Agreement or described in Section 3.3 of the Company Disclosure Letter, and except to the extent arising pursuant to applicable state takeover or similar laws, there are no registration rights, and there is no rights agreement, "poison pill" anti-takeover plan or other similar agreement to which the Company or any of its Subsidiaries is a party or by which it or they are bound with respect to any equity security of any class of the Company. There are no outstanding bonds, debentures, notes or other Indebtedness of the Company having the right to vote (or convertible into or exchangeable for securities having the right to vote) on any matters on which stockholders of the Company may vote.

(d) All outstanding shares of Company Common Stock are, and all shares of Company Common Stock which may be issued prior to the Effective Time, upon issuance on the terms and conditions specified in the instruments pursuant to which they are issuable, will be, duly authorized, validly issued, fully paid and nonassessable and not subject to or issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the DGCL, the Company Charter Documents or any agreement to which the Company is a party or is otherwise bound.

(e) There are no obligations, contingent or otherwise, of the Company or any of its Subsidiaries (i) to repurchase, redeem or otherwise acquire any shares of Company Common Stock or the capital stock of the Company or any of its Subsidiaries, or (ii) to provide funds to or make any investment (in the form of a loan, capital contribution or similar financing arrangement) in the Company or any of its Subsidiaries. Except as described in Section 3.3 of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries is party to any agreement, arrangement or obligation, contingent or otherwise, restricting the transfer of, affecting the voting rights of, requiring the registration for sale of, granting any preemptive or antidilutive rights with respect to, or containing any right of first refusal with respect to, any equity securities or Indebtedness of the Company or any of its Subsidiaries.

(f) All of the outstanding shares of capital stock of and other voting or equity interests in each of the Company's Subsidiaries have been and are duly authorized and validly issued, fully paid and nonassessable and are owned beneficially and of record wholly by the Company or one of the Company's wholly-owned Subsidiaries, free and clear of any Encumbrances except for any Encumbrances under applicable securities Laws. No shares of capital stock of any of the Company's Subsidiaries are subject to or were issued in violation of the preemptive rights of any stockholder or any purchase option, call option, right of first refusal, subscription right or any similar right under any provision of the DGCL, the organizational documents of any of the Company's Subsidiaries or any agreement to which the Company or any of its Subsidiaries is a party or otherwise bound. There are no issued or outstanding (i) shares of capital stock of or other voting securities of or ownership interests in any of the Company's Subsidiaries (other than those held by the Company or any of the Company's Subsidiaries), (ii) securities of the Company or any of its Subsidiaries convertible into or exchangeable for shares of capital stock or other voting securities of or ownership interests in any Subsidiary of the Company, (iii) warrants, calls, options or other rights to acquire from any of the Company's Subsidiaries, or other obligation of any of the Company's Subsidiaries to issue, any capital stock or other voting securities or ownership interests in or any securities convertible into or exchangeable for capital stock or other voting securities or ownership interests in any of the Company's Subsidiaries, or (iv) Company Subsidiary-issued restricted shares, stock appreciation rights, performance units, contingent value rights, "phantom" stock or similar securities or rights that are derivative of, or provide economic benefits based, directly or indirectly, on the value or price of, any capital stock or voting securities of any of the Company's Subsidiaries (the items in clauses (i) through (iv) being referred to collectively as the "Subsidiary Securities"). Neither the Company nor any of its Subsidiaries are a party to any agreement (voting or otherwise) with respect to any Subsidiary Securities. There are no outstanding bonds, debentures, notes or other Indebtedness of the Company's Subsidiaries having the right to vote (or convertible into or exchangeable for securities having the right to vote) on any matters on which stockholders of the Company may vote. There are no outstanding obligations of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any Subsidiary Securities.

3.4 SEC Filings; Financial Statements.

(a) Since October 1, 2015, the Company has filed or furnished, as applicable, on a timely basis all reports, schedules, forms, proxy statements, registration statements and other documents (including exhibits and all other information incorporated therein) required to be filed or furnished, as applicable, with the SEC by the Company (such documents, the "Company SEC Documents"). None of the Company's Subsidiaries is required to file or furnish any reports with the SEC pursuant to the Exchange Act. As of their respective SEC filing dates (and in the case of any registration statements, as of the effective dates thereof), the Company SEC Documents complied, or if filed or furnished or to become effective subsequent to the date of this Agreement, will comply, as to form in all material respects with the applicable requirements of the Securities Act, the Exchange Act, and the Sarbanes-Oxley Act of 2002, as amended (together with the rules and regulations promulgated thereunder, the "Sarbanes-Oxley Act"), as the case may be. As of their respective dates (or, if amended prior to the date of this Agreement, as of the date of such amendment), none of the Company SEC Documents contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the

circumstances under which they were made, not misleading, in each case giving effect to any amendments thereto filed or furnished prior to the date hereof, and any Company SEC document filed or furnished with the SEC subsequent to the date of this Agreement will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. No executive officer of the Company has failed in any respect to make the certifications required of him or her under Section 302 or 906 of the Sarbanes-Oxley Act. Subject to any comments that the SEC staff may make subsequent to the date hereof with respect to the Offer Documents or the Schedule 14D-9 and subject to any review or investigation that the SEC may initiate subsequent to the date hereof with respect to the Offer Documents or the Schedule 14D-9, (i) there are no outstanding or unresolved comments received from the SEC staff with respect to the Company SEC Documents and (ii) to the Company's Knowledge, none of the Company SEC Documents is the subject of ongoing SEC review or investigation.

(b) The audited consolidated financial statements and unaudited consolidated interim financial statements of the Company (including, in each case, any related notes and schedules thereto) included in the Company SEC Documents (i) complied at the time they were filed, or in the case of any such financial statements filed after the date of this Agreement, will comply, as to form, as of their respective filing dates with the SEC, in all material respects with the applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, (ii) were prepared, or in the case of any such financial statements filed after the date of this Agreement, will be prepared, in accordance with United States generally accepted accounting principles ("GAAP") applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes to such financial statements or, in the case of unaudited interim financial statements, as permitted by the SEC on Form 10-Q under the Exchange Act), (iii) in the case of such audited consolidated financial statements, fairly present (except as may be indicated in the notes thereto), or in the case of any such financial statements filed after the date of this Agreement, will fairly present, in all material respects, the consolidated financial position of the Company and its consolidated Subsidiaries as of the dates thereof and their consolidated results of operations and cash flows for the periods presented therein, in conformity with GAAP, (iv) in the case of such unaudited consolidated interim financial statements, fairly present (except as may be indicated in the notes thereto), or in the case of any such financial statements filed after the date of this Agreement, will fairly present, in all material respects, the consolidated financial position of the Company and its consolidated Subsidiaries as of the dates thereof and their consolidated results of operations and cash flows for the periods presented therein (subject to normal year-end adjustments) and (v) were prepared, or in the case of any such financial statements filed after the date of this Agreement, will be prepared, in accordance with the books and records of the Company and its Subsidiaries.

(c) The Company has established and maintains disclosure controls and procedures and a system of internal control over financial reporting (as such terms are defined in paragraphs (e) and (f), respectively, of Rule 13a-15 under the Exchange Act) as required by Rule 13a-15 under the Exchange Act. The Company's system of internal control over financial reporting is sufficient to provide reasonable assurance (i) that transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP, (ii) that receipts and expenditures are executed in accordance with the authorization of

management and (iii) that any unauthorized use, acquisition or disposition of the Company's assets that would materially affect the Company's financial statements would be detected or prevented in a timely manner. Since October 1, 2015, the Company has not identified (x) any material weakness (as defined in Rule 12b-2 of the Exchange Act in the design or operation of internal controls which is reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information, (y) any significant deficiency (as defined in Rule 12b-2 of the Exchange Act) in the system of internal control over financial reporting utilized by the Company and its Subsidiaries that has not been subsequently remediated or (z) any fraud or allegation of fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal controls.

(d) The Company maintains effective disclosure controls (as defined by Rule 13a-15 or 15d-15 under the Exchange Act) designed to ensure that all information (both financial and non-financial) required to be disclosed by the Company (including information regarding the Company's Subsidiaries) in the reports that it files or submits under the Exchange Act is recorded, processed, summarized, and reported within the time periods specified in the rules and forms of the SEC, and that all such information is accumulated and communicated to the Company's management as appropriate to allow timely decisions regarding required disclosure and to make the certifications of the chief executive officer and chief financial officer of the Company required under the Exchange Act with respect to such reports. Since October 1, 2015, the Company has not received any material complaint, allegation, assertion or claim regarding deficiencies in the accounting or auditing practices, procedures, methodologies or methods of the Company or its internal accounting controls.

(e) None of the Acquired Entities is a party to or has any obligation or other commitment to become a party to any securitization transaction, joint venture, off-balance sheet partnership or any similar Contract (including any Contract relating to any transaction or relationship between or among the Acquired Entities, on the one hand, and any unconsolidated Affiliate, including any structured finance, special purpose or limited purpose Entity, on the other hand, or any "off-balance sheet arrangements" (as defined in Item 303(a) of Regulation S-K under the Exchange Act)) where the purpose or intended effect of such Contract is to avoid disclosure of any material transaction involving, or material liabilities of, any Acquired Entity in any Acquired Entity's published financial statements or other Company SEC Documents. The Company is in compliance in all material respects with the applicable listing and corporate governance rules and regulations of NASDAQ. Neither the Company nor any of its Subsidiaries has outstanding, or has arranged any outstanding, "extensions of credit" to directors or executive officers of the Company prohibited by Section 402 of the Sarbanes-Oxley Act.

3.5 Absence of Changes.

(a) Since July 1, 2018 through the date of this Agreement (and other than in relation to the negotiation and execution of this Agreement), the Acquired Entities have conducted their business in the Ordinary Course of Business, and during such period none of the Acquired Entities has undertaken or effected, or authorized or agreed to undertake or effect, any action that, if taken or effected after the date of this Agreement, would require Parent's consent pursuant to <u>Sections 5.1(a)</u>, <u>5.1(c)</u>, <u>5.1(d)</u>, <u>5.1(e)</u>, <u>5.1(g)</u>, <u>5.1(h)</u>, <u>5.1(i)</u>, <u>5.1(i)</u>, <u>5.1(l)</u>, <u>5.1(n)</u>, <u>5.1(n)</u>, <u>5.1(n)</u>, <u>5.1(q)</u>, <u>5.1(q)</u>, <u>5.1(r)</u> or <u>5.1(s)</u>.

(b) Since July 1, 2018 through the date of this Agreement, there has not been any change, event, occurrence or effect that has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

3.6 **Title to Assets**. The Acquired Entities have good and marketable title to all assets owned by them, and valid leasehold interests in all assets leased by them, that are material to the Company's business as conducted on the date hereof, including such assets (other than capitalized leases) reflected on the audited balance sheet in the Annual Report on Form 10-K for the fiscal year ended September 30, 2017 (the "**Balance Sheet**") filed by the Company with the SEC (except for assets sold or otherwise disposed of in the Ordinary Course of Business since the date of such Balance Sheet, which, for the avoidance of doubt, are not material to the conduct of the business of the Company and its Subsidiaries). Notwithstanding the foregoing, it is understood and agreed that matters regarding Intellectual Property are not addressed in this **Section 3.6**.

3.7 Real Property; Equipment.

- (a) Neither the Company nor any of its Subsidiaries owns any real property.
- (b) Section 3.7(b) of the Company Disclosure Letter sets forth (i) a true and complete list of all real property leased, subleased or otherwise occupied by the Company or any of its Subsidiaries (the "Real Property"), (ii) the address for each Real Property and (iii) current rent amounts payable by the Company or its Subsidiaries related to such Real Property. All of the leases, subleases and other occupancy agreements (including all amendments, extensions, renewals, guaranties, and other agreements with respect thereto) (each, a "Real Property Lease") to which the Company or any of its Subsidiaries is a party and which relates to the Real Property are valid, subsisting, binding, and enforceable obligations of the Company or one of its Subsidiaries and in full force and effect, except as may be limited by bankruptcy, insolvency, moratorium and other similar applicable law affecting creditors' rights generally and by general principles of equity. Neither the Company nor any of its Subsidiaries nor, to the knowledge of the Company, any other party to any Real Property Lease is in material default under any of the Real Property Leases. Neither the Company nor any of its Subsidiaries leases, subleases or licenses any real property to any Person other than the Company and its Subsidiaries. The Company has made available to Parent complete and accurate copies of all Real Property Leases.

3.8 Intellectual Property.

(a) To the Company's knowledge, the Company and its Subsidiaries own, license, sublicense or otherwise possess legally enforceable rights to use all Intellectual Property necessary and material to conduct the business of the Company and its Subsidiaries, taken as a whole, as currently conducted (in each case excluding generally commercially available, off-the-shelf software programs). Section 3.8(a) of the Company Disclosure Letter sets forth, as of the date of this Agreement, a complete and accurate list of all of the following Intellectual Property that is owned or purported to be owned by or exclusively licensed to the Company and its Subsidiaries: (i) material patents and patent applications, (ii) registered trademarks and pending applications for registration of trademarks, (iii) Internet domain names, (iv) registered copyrights and pending applications for registration of copyrights, and (v) material unregistered trademarks (for which there are no pending applications).

- (b) To the Company's knowledge, (i) all Company Intellectual Property is owned by the Company or one of its Subsidiaries, and Company and its Subsidiaries have valid and enforceable rights to use all Intellectual Property necessary and material to conduct the business of the Company and its Subsidiaries as currently conducted, free and clear of all Encumbrances other than Permitted Encumbrances; provided that the foregoing shall not be deemed to be a representation and warranty of non-infringement, and (ii) all patents and registrations for trademarks, service marks and copyrights included in the Company Intellectual Property are subsisting, in full force and effect and enforceable, except as may be limited by bankruptcy, insolvency, moratorium and other similar applicable law affecting creditors' rights generally and by general principles of equity and have not expired or been canceled. To the Company's knowledge, as of the date of this Agreement, no Company Intellectual Property is involved in any interference, reissue, reexamination, opposition, or cancellation. All necessary registration, maintenance, renewal and other relevant filing fees have been timely paid.
- (c) To the Company's knowledge, the conduct of the business of the Company and its Subsidiaries, taken as a whole, as currently conducted, does not infringe, violate or constitute a misappropriation of, and as conducted since October 1, 2015, has not infringed, violated or constituted a misappropriation of, any Intellectual Property of any third party, except for such infringements, violations and misappropriations that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect. Except as set forth in Section 3.8(c) of the Company Disclosure Letter, since October 1, 2015 through the date of this Agreement, neither the Company nor any of its Subsidiaries has received any written claim or notice from any Person (i) alleging any such infringement, violation or misappropriation or (ii) advising that such Person is challenging or threatening to challenge the ownership, use, priority, validity or enforceability of any Company Intellectual Property, except, in each case in clauses (i) and (ii), for any such infringement, violation, misappropriation or challenge that would not reasonably be expected to result in a Material Adverse Effect.
- (d) The Company and its Subsidiaries have implemented commercially reasonable measures to maintain the confidentiality of the Company Intellectual Property of a nature that the Company intends to keep confidential.
- (e) To the Company's knowledge, no third party is infringing, violating or misappropriating any Company Intellectual Property and, as of the date of this Agreement, no Company Intellectual Property is subject to any outstanding Order, stipulation, settlement, consent, covenant not to sue or other disposition of dispute that adversely restricts the use, licensing, transfer or registration of, or adversely affects the validity, scope, use, registerability or enforceability of the Company's and its Subsidiaries' rights to, any Company Intellectual Property.
- (f) <u>Section 3.8(f)</u> of the Company Disclosure Letter contains a list of Open Source Code and the license under which such Open Source Code has been incorporated into or integrated with any product or service sold, distributed or made available by the Company

and its Subsidiaries. Except as, individually or in the aggregate, would not be reasonably expected to have a Material Adverse Effect, neither the Company nor any of its Subsidiaries have used Open Source Code in such a way that creates or purports to create any obligations for the Company or any of its Subsidiaries to freely license, publicly distribute, or dedicate to the public any product or service offered by the Company or any of its Subsidiaries.

- (g) No past or present director, officer or employee of the Company or any of its Subsidiaries owns (or has any claim, or any right (whether or not currently exercisable) to any ownership interest, in or to) any Company Intellectual Property. Neither the Company nor any of its Subsidiaries utilizes any material Intellectual Property rights of any of its or their current or former employees, contractors, or consultants (or Persons it or they currently intends to hire), except for any such Intellectual Property rights that have been assigned to the Company and its Subsidiaries, except as would not reasonably be expected to result in a Material Adverse Effect. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, all current and former employees, contractors, and consultants of the Company or any of its Subsidiaries who contributed to the creation, development, invention, modification, or improvement of any material Company Intellectual Property for or on behalf of the Company or any of its Subsidiaries during the course of their employment or engagement with the Company or any of its Subsidiaries have duly executed and delivered agreements with the Company or its Subsidiaries, as applicable, assigning to the Company or its Subsidiaries all such Company Intellectual Property conceived or reduced to practice during the course of their employment or engagement by the Company or its Subsidiaries, or such Company Intellectual Property has otherwise been assigned to, or is validly owned by, the Company or its Subsidiaries by operation of law or otherwise. To the Company's knowledge, no Person who has contributed to the creation, invention, modification, or improvement of any material Company Intellectual Property purportedly owned by the Company or any of its Subsidiaries, in whole or in part, was or is under any conflicting obligation with any third party that would affect the Company's or its Subsidiaries' title to or right to use any such Com
- (h) Since October 1, 2015, to the knowledge of the Company, (i) there have been no security breaches in the information technology systems of, used by or affecting the business of the Company and its Subsidiaries and (ii) there have been no disruptions in any information technology systems that adversely affected the business of the Company and its Subsidiaries, except in each case, as has not had, or would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (i) Since October 1, 2015: (i) there have been no failures, breakdowns, continued substandard performance or other adverse events affecting any computer systems, including the software, firmware, hardware, networks, interfaces, platforms and related systems, owned, leased or licensed by the Company and its Subsidiaries (collectively, the "Company Systems") that have caused or could reasonably be expected to result in the substantial disruption or interruption in or to the use of such Company Systems or the conduct of the businesses of the Company or its Subsidiaries, and (ii) to the knowledge of the Company, there have not been any incidents of unauthorized access or other security breaches of the Company Systems, except in each case for any such failures, breakdowns, substandard performance or other events which, individually or in the aggregate, have not had and would not reasonably be expected to result in a Material Adverse Effect.

(j) To the extent that the Company or its Subsidiaries have placed Source Codes in escrow, the agreements providing for the release of such Source Codes to third-parties under enumerated circumstances (i) provide that such Source Codes may be utilized by such third-parties solely for the purpose of continuing the benefits afforded to such third-parties by the applicable license agreement between such third-parties and the Company or its Subsidiaries and (ii) prohibit such third-parties from making derivatives of such Source Codes and from commercially exploiting such Source Codes.

3.9 Contracts.

- (a) <u>Section 3.9</u> of the Company Disclosure Letter lists each of the following types of Contracts to which the Company or any of its Subsidiaries is a party as of the date hereof or by which any of their respective properties or assets is bound as of the date hereof:
- (i) each Contract or group of related Contracts with the same party for the purchase of products or services by the Company or any of its Subsidiaries with a contractually committed undelivered balance in any calendar year in excess of \$100,000, other than any such Contract that is terminable by the Company or such Subsidiary on notice of 90 days or less with an aggregate liability to the Company and its Subsidiaries of less than \$50,000;
- (ii) each Contract or group of related Contracts with the same party for the sale of products or services by the Company or any of its Subsidiaries with a contractually committed undelivered balance in any calendar year in excess of \$50,000, other than any such Contract that is terminable by the Company or such Subsidiary on notice of 90 days or less with an aggregate liability to the Company and its Subsidiaries of less than \$15,000;
 - (iii) each Contract for the sale of any capital assets with a sale price in excess of \$100,000;
 - (iv) each Contract for capital expenditures with a remaining unpaid balance in excess of \$100,000;
- (v) each Contract providing for any surety bond or letter of credit in excess of \$100,000 required to be maintained by the Company or any of its Subsidiaries;
- (vi) any non-competition or other Contract that prohibits or otherwise restricts, in any material respect, the Company or any of its Subsidiaries from freely engaging in any business material to the Company and its Subsidiaries, taken as a whole, anywhere in the world and any Contract requiring the Company or any of its Subsidiaries to conduct any business of a "most favored nations" basis with any third party or to purchase a minimum quantity of goods or services;

(vii) each Contract providing for the establishment or operation of a partnership or joint venture with any third party, excluding reseller and OEM Contracts entered into in the Ordinary Course of Business;

(viii) each license pursuant to which the Company or any of its Subsidiaries (A) grants to any third party either the exclusive right to use any Company Intellectual Property or any right to use any Source Code included in the Company Intellectual Property, or (B) is granted by any third party the right to use any material Intellectual Property included in the Company's products (excluding generally commercially available, off-the-shelf software programs);

- (ix) any Contract relating to Indebtedness (other than intercompany Indebtedness owed by the Company or any wholly owned Subsidiary to any other wholly owned Subsidiary, or by any wholly owned Subsidiary to the Company or any of its Subsidiaries having an outstanding principal amount in excess of \$250,000;
- (x) any Contract that grants any right of first refusal, right of first offer, put, call or similar right with respect to any material assets, rights or properties of the Company or its Subsidiaries;
- (xi) any Contract entered into on or after January 1, 2017 that provides for the acquisition or disposition of any assets (other than acquisitions or dispositions in the Ordinary Course of Business) or business (whether by merger, sale of stock, sale of assets or otherwise) or capital stock or other equity interests of any Person, and with any outstanding obligations as of the date hereof, in each case with a value in excess of \$150,000;
- (xii) any Contract with an affiliate or other Person that would be required to be disclosed under Item 404(a) of Regulation S-K promulgated under the Exchange Act;
- (xiii) any employment or consulting Contract with any current or former (A) executive officer of the Company or any of its Subsidiaries, (B) member of the Company's Board or any member of the board of directors of any Subsidiary or (C) any employee or consultant of the Company or any of its Subsidiaries providing for an annual base salary or payment in excess of \$150,000; provided that disclosure shall not be required of any offer letter that does not provide the recipient thereof with a contractual entitlement, other than an entitlement that has been fulfilled by the Company or any of its Subsidiaries;
 - (xiv) any employee collective bargaining agreement or other Contract with any labor union; and
 - (xv) each "material contract" as described in Item 601(b)(10) of the SEC's Regulation S-K.
- (b) Each contract, arrangement, commitment or understanding of the type described in <u>Section 3.9</u> of the Company Disclosure Letter, whether or not set forth in <u>Section 3.9</u> of the Company Disclosure Letter, is referred to herein as a "Material Contract." Except Material Contracts that have expired or terminated by their terms, all of the Material

Contracts are valid and binding on the Company or any Subsidiary of the Company, as the case may be, and, to the knowledge of the Company, each other party thereto, as applicable, and in full force and effect, except as may be limited by bankruptcy, insolvency, moratorium and other similar applicable law affecting creditors' rights generally and by general principles of equity. No Acquired Entity has, and to the knowledge of the Company, none of the other parties thereto have, violated in any material respect any provision of, or committed or failed to perform any act, and no event or condition exists, which with or without notice, lapse of time or both would constitute a material default under the provisions of any Material Contract, except in each case for those violations and defaults which, individually or in the aggregate, have not had and would not reasonably be expected to result in a Material Adverse Effect and no Acquired Entity has received written notice of any of the foregoing. The Company has made available to Parent true and complete copies of all Material Contracts, including any amendments thereto.

3.10 **Liabilities**. The Acquired Entities have no liabilities or obligations, whether or not accrued, contingent or otherwise, required by GAAP or the applicable rules and regulations of the SEC to be reflected on the consolidated balance sheet of the Company or the notes thereto, except for: (a) liabilities and obligations reflected on the Balance Sheet (including any related notes); (b) liabilities and obligations incurred in the Ordinary Course of Business since the date of the Balance Sheet; (c) liabilities and obligations that have not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect; and (d) liabilities and obligations incurred in connection with the Offer, the Merger and the other transactions contemplated herein.

3.11 Compliance with Law.

(a) Each Acquired Entity is and, since October 1, 2015, has been, in compliance in all material respects with all applicable Law. None of the Acquired Entities has received any written notice since October 1, 2015 that remains unresolved (i) of any material administrative, civil or criminal investigation or material audit by any Governmental Body relating to the Company or any of its Subsidiaries or (ii) from any Governmental Body alleging that the Company or any of its Subsidiaries are not in compliance with any applicable Law in any material respect.

(b) None of the Company, any of its Subsidiaries, nor any director, officer or, to the knowledge of the Company, employee or agent of the Company or any of its Subsidiaries has, within the past five (5) years, (i) made any unlawful contributions, gifts, entertainment or other unlawful payments relating to an act by any Governmental Body or any political activity, (ii) made any unlawful payment to any foreign or domestic government official or employee or to any foreign or domestic political party or campaign or otherwise violated any provision of the U.S. Foreign Corrupt Practices Act of 1977, as amended, (iii) made any other unlawful payment under any applicable Law relating to anti-corruption, bribery, money-laundering or similar matters, (iv) been subject to any Legal Proceedings or Order relating to the U.S. Foreign Corrupt Practices Act of 1977, as amended, or any other Law relating to anti-corruption, bribery, money-laundering or similar matters or (v) established or maintained an unlawful fund of corporate monies or other properties. Within the past five (5) years, neither the Company nor any of its Subsidiaries has disclosed to any Governmental Body that it violated or may have violated any Law relating to anti-corruption, bribery, money-laundering or similar matters. To the knowledge of the Company, no Governmental Body is investigating, examining, or reviewing the Company's compliance with any applicable provisions of any Law relating to anti-corruption, bribery, money-laundering or similar matters.

(c) Since October 1, 2015, to the knowledge of the Company, none of the Acquired Entities has been a party to any contract or bid with, and has not conducted business or participated in any transaction involving, directly or indirectly, any Prohibited Persons.

3.12 **Regulatory Matters**. The Acquired Entities have obtained all material approvals, authorizations, certificates, registrations, licenses, exemptions, variances, clearances, commissions, permits, consents, permissions, qualifications, orders and other rights from, and have made all declarations, notices, and filings with, all applicable Governmental Bodies (collectively, "<u>Authorizations</u>") necessary for them to lawfully own, lease and operate their respective properties and assets and conduct their respective businesses as presently conducted, and all such Authorizations (i) are in full force and effect, (ii) are not subject to any Legal Proceeding that could result in any modification, termination or revocation of any Authorization and, to the Company's knowledge, no suspension or cancellation of any such Authorizations is threatened in writing and (iii) are in compliance with the terms and requirements of all Authorizations, except for such Authorizations the absence or the failure of which to be in full force and effect, the modification, termination or revocation of, suspension or cancellation of, or non-compliance with, has not had, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

3.13 Privacy.

(a) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, each of the Acquired Entities has at all times complied, and as of the date hereof is in compliance, with (i) all applicable Privacy Laws with regard to the transmission, storage, collection and use of Personal Information, (ii) all obligations to third parties with respect to Personal Information, and (iii) all of its Privacy Policies. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, since October 1, 2015, each of the Acquired Entities has taken appropriate measures to ensure that the Company Data is secure and protected against loss, damage, alteration, and unauthorized disclosure, access and processing. Without limiting the foregoing: (i) to the extent required by any Privacy Law (x) each Acquired Entity has maintained and posted privacy policies or notices providing notice of the Acquired Entity's privacy, data protection and data security practices regarding the Processing of information, including Personal Information, in connection with the operation of the Acquired Entity's business (the "Privacy Policies"), and (y) each Acquired Entity has at all times posted such Privacy Policies in a clear and conspicuous location on the Acquired Entity's websites, and (ii) true and complete copies of all such Privacy Policies have been made available to Parent.

(b) Since October 1, 2015, each Acquired Entity has, to the extent required by any Privacy Law, in all material respects, implemented and maintained appropriate security measures, plans, controls and programs, including (1) written information security programs, policies and procedures designed to secure and protect all Personal Information

contained in the Company Data from loss and illegal or unauthorized Processing that (i) comply with all applicable Privacy Laws and Privacy Policies, and (ii) are consistently enforced and followed in the conduct of the business of the Company and its Subsidiaries, and (2) plans, controls, and programs that are designed to (i) identify internal and external risks to the privacy and security of Personal Information, (ii) implement, monitor and improve adequate and effective administrative, electronic and physical safeguards to eliminate those risks, (iii) maintain notification procedures in compliance with applicable Privacy Laws in the case of any breach of privacy or security compromising data, including data containing Personal Information, and (iv) provide for the prevention of data loss including loss of Personal Information.

- (c) Since October 1, 2015, except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect (i) no breach or violation of any privacy or security policy of any Acquired Entity with respect to Personal Information has occurred, is threatened, or has been communicated to any Acquired Entity in writing or otherwise, (ii) there has been no unauthorized or illegal Processing of or access to any Personal Information, (iii) no Acquired Entity has received any written notice alleging any violation of any Privacy Law, Privacy Policy, or privacy or data security obligation in any Company Contract and (iv) no claim or proceeding is pending against any Acquired Entity, or threatened in writing against any Acquired Entity, alleging that the Processing or disclosure of Personal Information by the Acquired Entity violates any Privacy Policy, or Company Contract.
- (d) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, each Acquired Entity has at all times made all required disclosures to, and obtained all required consents from, users, customers, employees, contractors, Governmental Bodies and other applicable Persons required by all applicable Privacy Laws and Privacy Policies, and as necessary for the Acquired Entity's Processing of Personal Information in connection with the conduct of its business, and have filed any and all required registrations with the applicable data protection authority.

3.14 Tax Matters.

- (a) Each of the material Tax Returns required to be filed by or with respect to the Acquired Entities has been properly prepared and timely filed, and all such Tax Returns (including information provided therewith or with respect to thereto) are true, complete and correct in all material respects.
- (b) Each of the Acquired Entities has fully and timely paid all Taxes owed by such company shown as due on any Tax Return, other than Taxes being contested in good faith as described in <u>Section 3.14(b)</u> of the Company Disclosure Letter, and has made provision adequate to cover any and all Taxes that are not yet due and payable, for all taxable periods, or portions thereof, ending on or before the date hereof.
- (c) There are no outstanding agreements extending or waiving the statutory period of limitations applicable to any claim for, or the period for the collection or assessment or reassessment of, Taxes due from any of the Acquired Entities for any taxable period and no request for any such waiver or extension is currently pending.

- (d) No audit, examination, investigation or other proceeding by any Governmental Body is pending or threatened in writing with respect to any Taxes due from or with respect to any of the Acquired Entities, no Governmental Body has given written notice of any intention to assert any deficiency or claim for additional Taxes against any of the Acquired Entities, no claim in writing has been made by any Governmental Body in a jurisdiction where an Acquired Entity does not file Tax Returns that it is or may be subject to taxation by that jurisdiction, and all deficiencies for Taxes asserted or assessed in writing against the Acquired Entities have been fully and timely paid, settled or properly reflected in the financial statements or other Company SEC Documents.
- (e) There are no Encumbrances for Taxes upon the assets or properties of any of the Acquired Entities, except for Permitted Encumbrances.
- (f) None of the Acquired Entities has (i) participated in any listed transaction within the meaning of Treasury Regulations Section 1.6011-4(b)(2) (or any similar provision of state, local or foreign Tax Law) or (ii) taken any position on a Tax Return that could reasonably be expected to result in an "underpayment" of Tax to which Code Section 6662 (or any similar provision of state, local or foreign Tax Law) applies.
- (g) None of the Acquired Entities is a party to any agreement relating to the sharing, allocation or indemnification of Taxes, or any similar agreement, contract or arrangement or has any liability for Taxes of any Person (other than members of the affiliated group, within the meaning of Section 1504(a) of the Code, filing consolidated federal income Tax Returns of which the Company is the common parent) under Treasury Regulations Section 1.1502-6 or similar provision of state, local or foreign Tax Law, as a transferee or successor, by contract, or otherwise.
- (h) Each of the Acquired Entities has withheld (or will withhold) from their respective employees, independent contractors, creditors, stockholders and third parties and timely paid (or will timely pay) to the appropriate Governmental Body proper and accurate amounts in all material respects for all periods ending on or before the Closing Date in compliance with all Tax withholding and remitting provisions of applicable Laws and has each complied in all material respects with all Tax information reporting provisions of all applicable Laws.
- (i) None of the Acquired Entities will be required to include in a taxable period ending after the Closing Date taxable income attributable to income that accrued in a taxable period prior to the Closing Date but was not recognized for Tax purposes in such prior taxable period (or to exclude from the calculation of taxable income in a taxable period ending after the Closing Date any deduction the recognition of which was accelerated from such taxable period to a taxable period prior to the Closing Date) as a result of the installment method of accounting, the completed contract method of accounting, the long-term contract method of accounting, the cash method of accounting, Section 481 of the Code, Section 965(h) of the Code or Section 108(i) of the Code or comparable provisions of state, local or foreign Tax Law, any intercompany transaction or excess loss account described in Treasury Regulations under Section 1502 of the Code or comparable provisions of state, local or foreign Tax Law, or for any other reason.

- (j) Any adjustment of Taxes of any of the Acquired Entities made by the IRS, which adjustment is required to be reported to the appropriate state, local, or foreign Governmental Bodies, has been so reported.
- (k) None of the Acquired Entities has executed or entered into a closing agreement pursuant to Section 7121 of the Code or any similar provision of state, local or foreign Tax Law, and none of the Acquired Entities has been subject to or been issued any private letter ruling or comparable ruling of the IRS or any other Governmental Body.
- (l) A true, complete and accurate copy of each U.S. federal and state income Tax Return and each Swedish income Tax Return and each Canadian federal and provincial income Tax Return filed by or on behalf of the Acquired Entities prior to the date of this Agreement with respect to a Tax period beginning on or after October 1, 2015 has been made available to Parent.
- (m) None of the Acquired Entities has constituted a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of shares qualifying for tax-free treatment under Section 355 of the Code (i) in the two years prior to the date of this Agreement or (ii) in a distribution that could otherwise constitute part of a "plan" or "series of related transactions" (within the meaning of Section 355(e) of the Code) in conjunction with the transactions contemplated hereby.
- (n) <u>Section 3.14(n)</u> of the Company Disclosure Letter sets forth a true and complete list of each Subsidiary of the Company that is a "controlled foreign corporation" as defined in Section 957 of the Code. None of the Acquired Entities (i) is a "passive foreign investment company" within the meaning of Section 1297 of the Code, or (ii) has a permanent establishment (within the meaning of an applicable Tax treaty) or otherwise has an office or fixed place of business in a country other than the country in which it is organized.

(o) Canadian Tax Matters.

- (i) The shares of Angoss Software Corporation or its predecessor have never been considered "taxable Canadian property" as defined in section 248 of the Income Tax Act (Canada) at any time, including at the time such shares were acquired by the Company.
- (ii) Solely with respect to the Angoss Acquisition, the value of the consideration paid or received between the Acquired Entities for the acquisition, sale, transfer or provision of property or the provision of services from or to any Person with which it was not dealing at arm's length at the relevant time was the fair market value of such property acquired, provided or sold or services purchased or provided.
- (iii) There are no circumstances existing prior to the date hereof which could reasonably be expected to result in the application of any of sections 80, 80.01, 80,02, 80.03 and 80.04 of the Income Tax Act (Canada) to an Acquired Entity.

3.15 Employee Benefit Plans.

- (a) Section 3.15(a) of the Company Disclosure Letter sets forth a complete and accurate list of all Employee Plans.
- (b) With respect to each Employee Plan in effect on the date of this Agreement, the Company has made available to Parent a complete and accurate copy of (i) such Employee Plan and all amendments thereto (or, if a plan is not written, a written description of such Employee Plan), (ii) the three most recent annual reports (Form 5500) filed with the IRS, if any, (iii) copies of all nondiscrimination and top-heavy testing reports for the last three plan years, if any, (iv) all material, non-routine correspondence to or from any Governmental Body, (v) the most recent determination or opinion letter from the IRS, if any, (vi) each trust agreement, funding agreement, group annuity contract and summary plan description, if any, relating to such Employee Plan, and (vi) the most recent financial statements for such Employee Plan, if any.
- (c) Each Employee Plan has been established, administered and maintained in all material respects, in accordance with ERISA, the Code and all other applicable Laws and the regulations thereunder and in accordance with its terms.
- (d) With respect to the Employee Plans, there are no material benefit obligations for which contributions have not been made or properly accrued to the extent required by GAAP and all contributions, benefits, premiums and other payments required by and due under the terms of each Employee Plan and applicable Law have, in all material respects, been timely paid.
- (e) All the Employee Plans that are intended to be qualified under Section 401(a) of the Code have received currently effective determination letters from the IRS to the effect that such Employee Plans are qualified and the plans and trusts related thereto are exempt from federal income taxes under Sections 401(a) and 501(a), respectively, of the Code, or are based on prototype or volume submitter documents that have received such letters, no such determination letter has been revoked, and to the Company's knowledge, no such revocation has been threatened, and no act or omission has occurred, that would materially and adversely affect such Employee Plan's qualification. Any non-US Employee Plans that are intended to be funded or book-reserved are funded or book-reserved, as appropriate, based on reasonable actuarial assumptions.
- (f) None of the Company, nor any of the Company's Subsidiaries or any of their ERISA Affiliates (i) has within the preceding six (6) calendar years sponsored, maintained, contributed to, been obligated to contribute to or had any liability or obligation with respect to an Employee Plan that is subject to Section 412 of the Code or Title IV of ERISA or (ii) has within the preceding six (6) calendar years sponsored, maintained, contributed to, been obligated to contribute to or had any liability or obligation with respect to a "multiemployer plan" (as defined in Section 3(37) of ERISA or Section 4001(a)(3) of ERISA).

- (g) Except as set forth in Section 3.15(g) of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries is a party to any written (i) agreement with any stockholders, director, executive officer or other key employee of the Company or any of its Subsidiaries (A) the benefits of which are contingent, or the terms of which are materially altered, upon the occurrence of a transaction involving the Company or any of its Subsidiaries of the nature of any of the transactions contemplated by this Agreement, (B) providing any term of employment or compensation guarantee or (C) providing severance benefits or other benefits after the termination of employment of such director, executive officer or key employee, or (ii) agreement or plan binding the Company or any of its Subsidiaries, including any stock option plan, stock appreciation right plan, restricted stock plan, stock purchase plan or severance benefit plan, any of the benefits of which shall be increased, or the vesting of the benefits of which shall be accelerated, by the occurrence of any of the transactions contemplated by this Agreement or the value of any of the benefits of which shall be calculated on the basis of any of the transactions contemplated by this Agreement.
- (h) None of the Employee Plans (i) promises or provides retiree medical or other retiree welfare benefits to any Person, except as required by applicable Law, or (ii) is a "multiple employer welfare arrangement" (within the meaning of Section 3(40) of ERISA).
- (i) Except as set forth in Section 3.15(i) of the Company Disclosure Letter, no amount that could be received (whether in cash or property or the vesting of any property) as a result of the consummation of the transactions contemplated by this Agreement (either alone or in combination with any other event), would not be deductible by reason of Section 280G of the Code or subject to an excise tax under Section 4999 of the Code.
- (j) Except as set forth in Section 3.15(j) of the Company Disclosure Letter, neither the execution or delivery of this Agreement, the consummation of the Offer, the Merger, nor any of the other transactions contemplated by this Agreement will (either alone or in combination with any other event) in any material respect: (i) entitle any current or former director, employee, contractor, or consultant of the Company or any Subsidiary to severance pay or any other payment, (ii) accelerate the timing of payment, funding, or vesting, or increase the amount of compensation due to any such individual, (iii) limit or restrict the right of the Company or any of the Company's Subsidiaries to merge, amend, or terminate any Employee Plan, or (iv) increase the amount payable or result in any other material obligation pursuant to any Employee Plan.
- (k) Each Employee Plan that is subject to Section 409A of the Code has been operated in compliance in all material respects with such Section 409A and all applicable regulatory guidance (including, without limitation, proposed regulations, notices, rulings and final regulations).
- (l) No Employee Plan is presently, or since October 1, 2015 has been, the subject of an examination or audit by any Governmental Body. No Employee Plan is presently, or since October 1, 2015 has been, the subject of an application or filing under, or a participant in, an amnesty, voluntary compliance, self-correction or similar program sponsored by any Governmental Body.

(m) With respect to each Employee Plan, (i) no non-exempt "prohibited transaction," within the meaning of Section 4975 of the Code or Section 406 of ERISA, has occurred, (ii) as of the date of this Agreement, there are no actions, suits or claims pending, or, to the Company's knowledge, threatened or anticipated (other than routine claims for benefits), (iii) as of the date of this Agreement, there are no audits, inquiries or proceedings pending or, to the Company's knowledge, threatened by any Governmental Body, and (iv) within the preceding six (6) calendar years, there has been no breach of fiduciary duty (including violations under Part 4 of Title I of ERISA), in each case of clauses (i) through (iv) which would reasonably be expected to result in a liability that is material to the Company and its Subsidiaries, taken as a whole.

3.16 Labor Matters. Since October 1, 2015, the Company and its Subsidiaries have complied with all applicable Laws relating to labor and employment, including those relating to employment, hiring, termination of employment, plant closing and mass layoff, wages, hours, collective bargaining, unemployment compensation, worker's compensation, equal employment opportunity, age and disability discrimination, employment discrimination, harassment, retaliation, reasonable accommodation, leave of absence, immigration control, employee classification, engagement of independent contractors, worker classification and all applicable Laws relating to the relations between it and any labor organization, trade union, work council or other body representing employees of the Company or any of its Subsidiaries, except for such failures to comply that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect. Except as set forth on Section 3.16 of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries is party to, or subject to, any collective bargaining agreement or any other agreement with any labor organization, work council, or trade union with respect to any of its or their operations. None of the current Company Associates are represented by a labor organization, trade union or work council with respect to their service to the Company or any of its Subsidiaries. As of the date of this Agreement, neither the Company nor any of its Subsidiaries is the subject of any proceeding asserting that the Company or any of its Subsidiaries has committed an unfair labor practice or is seeking to compel it to bargain with any labor union or labor organization that, individually or in the aggregate, would reasonably be expected to result in a Material Adverse Effect. As of the date of this Agreement, there are no pending or, to the Company's knowledge, threatened labor strikes, disputes, walkouts, work stoppages, slow-downs or lockouts involving the Company or any of its Subsidiaries that, individually or in the aggregate, would reasonably be expected to result in a Material Adverse Effect, and no such labor strike, dispute, walkout, work stoppage, slow-down or lockout involving the Company or any of its Subsidiaries has occurred since October 1, 2015. There are no legal actions, government investigations, or labor grievances pending, or, to the Company's knowledge, threatened relating to any employment related matter involving any current or former employee of the Company or any Subsidiary or applicant, including, but not limited to, charges of unlawful discrimination, retaliation or harassment, failure to provide reasonable accommodation, denial of a leave of absence, failure to provide compensation or benefits, unfair labor practices, or other alleged violations of Law, except for any of the foregoing that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect.

3.17 Environmental Matters.

- (a) No notice, demand, request for information, citation, summons or order has been received, no complaint has been filed, no penalty has been assessed, and no Legal Proceeding is pending and, to the knowledge of the Company, is threatened in writing by any Governmental Body or other Person relating to or arising out of any failure of the Company or any of its Subsidiaries to comply with any Environmental Law.
- (b) The Company and its Subsidiaries have complied with all Environmental Laws and have obtained and have been in compliance with all Environmental Permits required for the operations of the Company and its Subsidiaries as currently conducted, except for such failures to comply, in each case, that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect.
- (c) There has been no Release by the Company or any of its Subsidiaries, or for which the Company or any of its Subsidiaries would reasonably be expected to be liable by Company Contract or by operation of Law, of, and there has been no exposure to, any Hazardous Substance at, under, from or to any facility or real property currently or formerly owned, leased or operated by the Company or any of its Subsidiaries or any other location.
- (d) There are no liabilities of the Company or any of its Subsidiaries of any kind whatsoever, whether accrued, contingent, absolute, determined, determinable or otherwise arising under or relating to any Environmental Law or any Hazardous Substance (including liabilities that have been assumed or retained by Company Contract or operation of Law) and, to the knowledge of the Company, there is no condition, situation or set of circumstances that would reasonably be expected to result in or be the basis for any such liability.
- 3.18 **Transactions with Affiliates**. Between the date of the Balance Sheet and the date of this Agreement, no event has occurred that would be required to be reported by the Company pursuant to Item 404 of Regulation S-K (a "**Company Affiliate Transaction**"), except as disclosed in the Company SEC Documents. Any Company Affiliate Transaction since October 1, 2017 as of the time it was entered into and as of the time of any amendment or renewal thereof contained such terms, provisions and conditions as were at least as favorable to the Company or any of its Subsidiaries as would have been obtainable by the Company in a similar transaction with an unaffiliated third party. To the knowledge of the Company, no present or former director, officer, employee or Affiliate (other than Subsidiaries of the Company) of the Company or any of its Subsidiaries, nor any of such Person's Affiliates or immediate family members owns, directly or indirectly, on an individual or joint basis, any significant interest in, or serves as an officer or director or in another similar capacity of, any supplier or other independent contractor of the Company or any of its Subsidiaries, or any organization which has a Contract with the Company or any of its Subsidiaries.
- 3.19 **Legal Proceedings**; **Orders**. Except as set forth in <u>Section 3.19</u> of the Company Disclosure Letter, there is no pending or, to the knowledge of the Company, threatened Legal Proceeding against the Company or any of its Subsidiaries that seeks either damages in excess of \$200,000 or material equitable relief or challenges or seeks to prevent,

enjoin, alter or delay the transactions contemplated by this Agreement. Except as set forth in <u>Section 3.19</u> of the Company Disclosure Letter, within the past three years, there are no material Orders or settlements (including monetary settlements of more than \$200,000 paid to a plaintiff or group of plaintiffs, injunctive relief or the imposition of criminal sanctions) to which the Company or any of the Company's Subsidiaries is a party or by which any of their assets are bound.

3.20 Authority; Binding Nature of Agreement.

- (a) Assuming the accuracy of Parent's and Purchaser's representation and warranty set forth in Section 4.7, the Company has all requisite corporate power and authority and has taken all corporate action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Offer, the Merger and the other transactions contemplated herein. The execution, delivery and performance by the Company of this Agreement and the consummation by the Company of the Offer, the Merger and the other transactions contemplated herein have been duly authorized by all necessary corporate action on the part of the Company.
- (b) The Board of Directors of the Company (the "Company Board") (at a meeting duly called and held) has unanimously (i) determined that this Agreement and the Offer, the Merger and any other transaction contemplated herein are in the best interests of the Company's stockholders, (ii) approved and declared advisable this Agreement, the Offer, the Merger and the other transactions contemplated herein in accordance with the requirements of the DGCL, (iii) resolved to recommend that stockholders of the Company accept the Offer and tender their Shares pursuant to the Offer, (iv) resolved that this Agreement and the Merger shall be governed by and effected under Section 251(h) of the DGCL and the Merger shall be consummated as soon as practicable following the Offer Acceptance Time and (v) to the extent necessary, adopted a resolution having the effect (assuming the accuracy of Parent's and Purchaser's representation and warranty set forth in Section 4.7) of causing the Merger, this Agreement and the transactions contemplated by this Agreement not to be subject to (or, in the case of Section 203 of the DGCL, not to be subject to the restrictions on business combinations contained in Section 203 of the DGCL) any state Takeover Law or similar Law that might otherwise apply to the Merger or any of the other transactions contemplated by this Agreement, which resolutions, as of the date of this Agreement, have not been rescinded, modified or withdrawn in any way. This Agreement has been duly executed and delivered by the Company and constitutes the legal, valid and binding obligation of the Company and, assuming due authorization, execution and delivery by Parent and Purchaser, is enforceable against the Company in accordance with its terms, subject to (x) laws of general application relating to bankruptcy, insolvency and the relief of debtors, and (y) rules of law governing specific performance, injunctive relief and other equitable remedies.
- 3.21 **Section 203 of the DGCL, Etc.** As of the date hereof and as of the Offer Acceptance Time and assuming the accuracy of Parent's and Purchaser's representation and warranty set forth in <u>Section 4.7</u>, the Company Board has taken and will take all actions so that the restrictions (whether procedural, voting, approval, fairness or otherwise) applicable to business combinations contained in Section 203 of the DGCL are, and will be, inapplicable to the execution, delivery and performance of this Agreement and the Support Agreements, and the

timely consummation of the Offer, the Merger and any other transaction contemplated herein and will not restrict, impair or delay the ability of Parent or Purchaser to vote or otherwise exercise all rights as a stockholder of the Company. Assuming the accuracy of Parent's and Purchaser's representation and warranty set forth in Section 4.7, no "fair price," "moratorium," "control share acquisition" or other similar Law (or, in the case of Section 203 of DGCL, no restrictions on business combinations contained in Section 203 of the DGCL) or any anti-takeover provision in the Company's certificate of incorporation or bylaws is, or at the Effective Time will be, applicable to the Shares, the Offer, the Merger or the other transactions contemplated herein.

3.22 **Vote Required**. Assuming the Offer is consummated in accordance with the terms of this Agreement, and assuming the accuracy of Parent's and Purchaser's representation and warranty set forth in <u>Section 4.7</u>, no stockholder votes or consents are necessary for the Company to authorize this Agreement or to consummate the Offer, the Merger or any other transaction contemplated herein.

3.23 Non-Contravention; Consents.

(a) Assuming compliance with the applicable provisions of the DGCL, the HSR Act, the Exchange Act and the listing requirements of the NASDAQ, the execution, delivery and performance of this Agreement by the Company and the consummation by the Company of the Merger and the other transactions contemplated by this Agreement do not and will not (with or without notice or lapse of time, or both): (i) contravene, conflict with or result in any violation or breach of, or default under, any of the provisions of the Company Charter Documents or the comparable governing instruments of any of the other Acquired Entities; (ii) with or without notice or lapse of time or both, result in a breach or violation of, a termination (or right of termination) or default under, any change in or acceleration or creation of any obligations or the creation of any Encumbrance (other than Permitted Encumbrances) on any assets of any Acquired Entity or change or loss of rights pursuant to, or give a right of purchase (including under any right of first refusal or similar right) under, any Material Contract, in each case that would be binding upon any Acquired Entity or to which any of its properties are subject; or (iii) contravene, conflict with or result in a violation or breach of any Law or Order applicable to any Acquired Entity, except in each case in clauses (ii) and (iii), as, individually or in the aggregate, has not had and would not reasonably be expected to have a Material Adverse Effect.

(b) No consent, waiver, approval, license, permit, order or other authorization of, or registration, declaration or filing with, any Governmental Body or stock market or stock exchange, is required to be obtained or made by or with respect to the Company or any of its Subsidiaries in connection with the execution, delivery and performance of this Agreement or the consummation of the Offer, the Merger and the other transactions contemplated herein, other than (i) compliance with and filings under the HSR Act, (ii) the filing with the SEC of (A) the Schedule 14D-9 and (B) such reports under Section 13 of the Exchange Act as may be required in connection with this Agreement, the Offer, the Merger and the other transactions contemplated herein, (iii) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware and (iv) such filings as may be required under the rules and regulations of NASDAQ in connection with this Agreement, the Offer, the Merger and the other transactions contemplated herein.

- 3.24 **Fairness Opinion**. The Company Board has received the written opinion (the "**Fairness Opinion**") of GCA Advisors, LLC as financial advisor to the Company (the "**Company Financial Advisor**"), dated the date of this Agreement, to the effect that, as of the date of such opinion, and subject to the assumptions, qualifications and limitations set forth therein, the Offer Price is fair, from a financial point of view, to the Company's stockholders (other than Parent and its Affiliates).
- 3.25 **Financial Advisor**. Except as set forth on <u>Section 3.25</u> of the Company Disclosure Letter with respect to the retention of the Company Financial Advisor, neither the Company nor any of its officers, directors or employees has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finder's fees in connection with the Offer, the Merger or the other transactions contemplated herein.
- 3.26 **Disclosure**. None of the information supplied or to be supplied by or on behalf of the Company for inclusion or incorporation by reference in the Offer Documents or the Schedule 14D-9 will, at the time such document is filed with the SEC, at any time it is amended or supplemented or at the time it is disseminated to the Company's stockholders, or at the Offer Acceptance Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they are made, not misleading. The Schedule 14D-9 will comply as to form in all material respects with the provisions of applicable federal securities Laws and, on the date filed with the SEC and on the date first published, sent or given to the Company's stockholders and at the Offer Acceptance Time, shall not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading, except that no representation is made by the Company with respect to information supplied by Parent or Purchaser in writing for inclusion in the Schedule 14D-9. All documents that the Company is responsible for filing with the SEC in connection with the transactions contemplated hereby will comply as to form and substance in all material respects with the applicable requirements of the Securities Act and the Exchange Act.

3.27 Rule 14d-10 Matters. All amounts payable to holders of Shares and other equity interests of the Company ("Covered Securityholders") pursuant to the Employee Plans are being paid or granted as compensation for past services performed, future services to be performed or future services to be refrained from performing by the Covered Securityholders (and matters incidental thereto) and are not calculated based on the number of Shares tendered or to be tendered into the Offer by the applicable Covered Securityholder. The Compensation Committee of the Company (each member of which the Company Board determined is an "independent director" within the meaning of the applicable NASDAQ rules and is an "independent director" in accordance with the requirements of Rule 14d-10(d)(2) under the Exchange Act) (i) at a meeting duly called and held at which all members of the Compensation Committee were present, duly and unanimously adopted resolutions approving as an "employment compensation, severance or other employee benefit arrangement" within the meaning of Rule 14d-10(d)(1) under the Exchange Act (an "Employment Compensation Arrangement") (A) each Company Equity Plan and (B) the treatment of the Company Options and Company RSUs in accordance with the terms set forth herein, the applicable Company Equity Plan and any applicable Employee Plans, which resolutions have not been rescinded, modified or withdrawn in any way, and (ii) has taken all other actions necessary to satisfy the requirements of the non-exclusive safe harbor under Rule 14d-10(d) under the Exchange Act with respect to the foregoing arrangements.

- 3.28 **Material Customers and Suppliers**. Section 3.28 of the Company Disclosure Letter sets forth the names of (a) the Company's ten (10) largest customers as determined by revenues received from such Persons in the twelve (12) month period ended September 30, 2018 (each, a "Material Customer") and (b) the ten (10) largest suppliers as determined by expenditures in the twelve (12) month period ended September 30, 2018 (each, a "Material Supplier"). To the knowledge of the Company, since October 1, 2017, the Company has not received any written notice from any Material Supplier or Material Customer indicating that it will, or intends to, cease doing business with the Company or materially decrease the volume of business that it is presently conducting with the Company.
- 3.29 **Insurance**. Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, as of the date of this Agreement, all general liability, business interruption, product liability, property liability and any other material insurance policies maintained by the Company or any of its Subsidiaries ("**Insurance Policies**") are in full force and effect and all premiums due with respect to all Insurance Policies have been paid as of the date of this Agreement, and, to the extent applicable, neither the Company nor any of its Subsidiaries has received any notice of cancellation or non-renewal with respect to any such policy or coverage under the Insurance Policies.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF PARENT AND PURCHASER

Parent and Purchaser hereby represent and warrant to the Company as follows:

- 4.1 **Due Organization**. Each of Parent and Purchaser is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware and has all corporate power and authority necessary to own, lease and operate all of its properties and assets and to carry on its business as now conducted. Each of Parent and Purchaser is duly licensed or qualified to do business as a foreign corporation and is in good standing in each jurisdiction where the nature of the business conducted by it or the character or location of the properties and assets owned, leased or operated by it makes such licensing or qualification necessary, except for those jurisdictions where failure to be so licensed, qualified or in good standing would not reasonably be expected to, individually or in the aggregate, materially and adversely affect Parent's or Purchaser's ability to consummate the Offer, the Merger or the other transactions contemplated herein.
- 4.2 **Purchaser**. Purchaser was formed solely for the purpose of engaging in the transactions contemplated hereby and has not engaged in any business activities or conducted any operations other than in connection with the transactions contemplated hereby. Parent owns beneficially and of record all of the outstanding voting securities of Purchaser.

4.3 **Authority; Binding Nature of Agreement**. Parent and Purchaser have all requisite corporate power and authority to enter into this Agreement and to consummate the Merger and the other transactions contemplated hereby. The execution, delivery and performance by Parent and Purchaser of this Agreement and the consummation by the Parent and Purchaser of the Offer, the Merger and the other transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of Parent and Purchaser (subject to adoption of this Agreement by Parent as the sole stockholder of Purchaser, which shall occur immediately after the execution and delivery of this Agreement). This Agreement constitutes a valid and binding agreement of Parent and Purchaser and, assuming due authorization, execution and delivery by the Company, is enforceable against Parent and Purchaser in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium and other similar applicable Law affecting creditors' rights generally and by general principles of equity.

4.4 Non-Contravention; Consents. Assuming compliance with the applicable provisions of the DGCL, the HSR Act, the Exchange Act and the listing requirements of the NASDAQ, the execution, delivery and performance by Parent and Purchaser of this Agreement and the consummation by Parent and Purchaser of the Merger and the other transactions contemplated hereby do not and will not (with or without notice or lapse of time, or both): (a) contravene, conflict with, or result in any violation or breach of or default under any provisions of the certificate of incorporation or bylaws of Parent or Purchaser, (b) contravene, conflict with or result in a violation or breach of any provision of any applicable Law or Order; or (c) with or without notice or lapse of time or both, result in a breach or violation of, a termination (or right of termination) or default under, any change in or acceleration or creation of any obligations or the creation of any Encumbrance (other than Permitted Encumbrances) on any assets of the Parent or Purchaser or change or loss of rights pursuant to, or give a right of purchase, including any right of first refusal or similar right under, any Contract, in each case that would be binding upon Parent or Purchaser or to which any of its properties are subject, with such exceptions, in the case of each of clauses (b) and (c), as would not reasonably be expected to, individually or in the aggregate, materially and adversely affect Parent's or Purchaser's ability to consummate the Offer, the Merger or the other transactions contemplated herein. No vote of Parent's stockholders is necessary to approve this Agreement or any of the transactions contemplated by this Agreement.

4.5 Disclosure.

(a) The Offer Documents will comply in all material respects as to form with the provisions of applicable federal securities Laws and, on the date filed with the SEC, on the date first published, sent or given to the Company's stockholders and at the Offer Acceptance Time, shall not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading, except that no representation is made by Parent or Purchaser with respect to information supplied by the Company in writing for inclusion in the Offer Documents. The information to be supplied in writing by or on behalf of Parent for inclusion in the Schedule 14D-9, on the date the Schedule 14D-9 is filed with the SEC, on the date the Schedule 14D-9 is first published, sent or given to the Company's stockholders and at the Offer Acceptance Time, shall not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

- (b) None of the information provided by Parent for inclusion in the Offer Documents or any amendment or supplement thereto, at the time any Offer Document or any amendment or supplement thereto is distributed or disseminated and at the time of the consummation of the Offer, will contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading.
- 4.6 **Funds**. Parent has sufficient funds on hand or available through existing credit facilities to perform and cause Purchaser to perform all of their respective obligations pursuant to this Agreement, including without limitation payment in cash of the aggregate Offer Price at the Offer Acceptance Time and the aggregate Merger Consideration on the Closing Date and to pay all related fees and expenses.
- 4.7 **Ownership of Company Common Stock**. Other than as a result of this Agreement, the Support Agreements, or any agreement, arrangement or understanding between Parent and/or Purchaser, on the one hand, and any stockholder party to the Support Agreements, on the other, neither Parent nor Purchaser is, nor at any time for the past three (3) years has been, an "interested stockholder" of the Company as defined in Section 203 of the DGCL.
- 4.8 **Litigation**. There is no Legal Proceeding pending against or, to the knowledge of Parent, threatened against Parent or Purchaser, that would materially and adversely affect Parent's or Purchaser's ability to consummate the Offer, the Merger and any other transactions contemplated herein. Neither Parent nor Purchaser is subject to any continuing Order, consent decree, settlement agreement or similar written agreement with, or, to the knowledge of Parent, continuing investigation by, any Governmental Body, or any order, writ, judgment, injunction, decree, determination or award of any Governmental Body that would materially and adversely affect Parent's or Purchaser's ability to consummate the Offer, the Merger and any other transactions contemplated herein.
- 4.9 **Management Agreements.** As of the date of this Agreement, except for the Support Agreements and the matters set forth in a letter from Parent and Purchaser to the Company dated the date hereof and referencing this **Section 4.9**, none of Parent or Purchaser, or their respective executive officers, directors of Affiliates, has entered into any agreement, arrangement or understanding: (a) with any of the executive officers, directors, Affiliates or other stockholders of the Company that is currently in effect or would become effective in the future (upon consummation of the Offer or the Merger or otherwise) and that would be required to be disclosed under Item 1005(d) of Regulation M-A under the Exchange Act or (b) assuming the accuracy of **Section 3.27**, pursuant to which any stockholder of the Company would be entitled to receive consideration of a different amount or nature than the Offer Price or Merger Consideration or pursuant to which any stockholder of the Company agrees to tender such stockholder's shares in the Offer or agrees to vote against or not otherwise support any Superior Proposal.

- 4.10 **Financial Advisor**. Except for RBC Capital Markets, LLC, neither Parent, Purchaser nor any of their officers, directors or employees has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finder's fees in connection with the Offer, the Merger or the other transactions contemplated herein.
- 4.11 **Independent Investigation**. Each of Parent and Purchaser acknowledges that it has conducted to its satisfaction its own independent investigation and analysis of the business, operations, assets, liabilities, results of operations and condition (financial or otherwise) of the Company and the Subsidiaries and that each of Parent and Purchaser and their respective Representatives have received access to such books and records, facilities, equipment, contracts and other assets of the Company and the Subsidiaries that Parent, Purchaser and their respective Representatives have desired or requested to review for such purpose, and that Parent, Purchaser and their respective Representatives have had full opportunity to meet with the management of the Company and the Subsidiaries and to discuss the business, operations, assets, liabilities, results of operations and condition (financial or otherwise) of the Company and the Subsidiaries.

ARTICLE V

CERTAIN COVENANTS OF THE COMPANY

5.1 **Covenants of the Company**. Except as provided or permitted herein, set forth in **Section 5.1** of the Company Disclosure Letter or as consented to in writing by Parent (which consent shall not be unreasonably withheld, conditioned or delayed), during the period commencing on the date of this Agreement and ending at the Effective Time or such earlier date as this Agreement may be terminated in accordance with its terms (the "**Pre-Closing Period**"), the Company shall, and shall cause each of its Subsidiaries to, act and carry on its business in the Ordinary Course of Business and use commercially reasonable efforts to maintain and preserve its and each of its Subsidiary's business organization, assets and properties, preserve its business relationships with customers, strategic partners, suppliers, distributors and others having business dealings with it and timely file all material Tax Returns and pay all Taxes shown as due on such Tax Returns or that are otherwise required to be paid by or on behalf of the Company or any of its Subsidiaries, other than those contested in good faith by the Company with the prior written consent of Parent, which consent shall not be unreasonably withheld. Without limiting the generality of the foregoing, except as expressly provided or permitted herein or as set forth in **Section 5.1** of the Company Disclosure Letter, during the Pre-Closing Period the Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, do any of the following without the prior written consent of Parent (which consent shall not be unreasonably withheld, conditioned or delayed):

(a) (i) declare, set aside or pay any dividends (whether in cash, securities or other property) on, or make any other distributions (whether in cash, securities or other property) in respect of, any of its capital stock (other than dividends and distributions by a direct or indirect wholly-owned Subsidiary of the Company to its parent); (ii) split, combine or reclassify any of its capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock or any of its other securities; or (iii) purchase, redeem or otherwise acquire any shares of its capital stock or any other of its

securities or any rights, warrants or options to acquire any such shares or other securities, except, in the case of this clause (iii), for the acquisition of shares of Company Common Stock from former employees, directors and consultants in accordance with agreements in effect on the date hereof and set forth in **Section 5.1(a)** of the Company Disclosure Letter providing for the repurchase of shares at their original issuance price in connection with any termination of services to the Company or any of its Subsidiaries;

- (b) except as permitted by <u>Section 5.1(j)</u>, issue, deliver, sell, grant, pledge or otherwise dispose of or encumber any shares of its capital stock, any other voting securities or any securities convertible into or exchangeable for, or any rights, warrants or options to acquire, any such shares, voting securities or convertible or exchangeable securities (other than the issuance of shares of Company Common Stock upon the exercise of Company Stock Options or the vesting of Company RSUs outstanding on the date of this Agreement in accordance with the terms of equity award agreements in effect on the date hereof, access to which has been provided to Parent);
- (c) amend or propose to amend the Company Charter Documents or, with respect to any Subsidiary, its certificate of incorporation, by-laws or other comparable charter or organizational documents;
- (d) acquire (i) by merging or consolidating with, or by purchasing all or a substantial portion of the assets or any stock of, or by any other manner, any business or any corporation, partnership, joint venture, limited liability company, association or other business organization or division thereof or (ii) except in the Ordinary Course of Business, any assets that are material, in the aggregate, to the Company and its Subsidiaries, taken as a whole;
- (e) sell, lease, sublease, license, sublicense, mortgage, pledge, surrender, encumber, divest, cancel, waive, abandon or allow to lapse or expire, assign, transfer or dispose of, covenant not to assert, or create or incur any Encumbrance (other than a Permitted Encumbrance) on, any properties, assets, interests or rights (including Company Intellectual Property) of the Company or any of its Subsidiaries, other than (i) pursuant to non-exclusive licenses in the Ordinary Course of Business, (ii) pursuant to Contracts existing as of the date of this Agreement and disclosed in Section 5.1(e) of the Company Disclosure Letter or (iii) in the Ordinary Course of Business;
- (f) other than in the Ordinary Course of Business, terminate, cancel or require any material change in, or agree to any material change in, any Material Contract or enter into or amend any contract, agreement, commitment or arrangement that would have constituted a Material Contract if such had existed as of the date hereof;
 - (g) adopt or implement any stockholder rights plan;
- (h) (i) incur any Indebtedness or guarantee any Indebtedness of another person (other than up to \$200,000 pursuant to existing credit facilities in the Ordinary Course of Business), (ii) issue, sell or amend any debt securities or warrants or other rights to acquire any debt securities of the Company or any of its Subsidiaries, guarantee any debt securities of another person, enter into any "keep well" or other agreement to maintain any

financial statement condition of another person or enter into any arrangement having the economic effect of any of the foregoing, (iii) make any loans, advances (other than routine advances to employees of the Company and its Subsidiaries in the Ordinary Course of Business, or pursuant to the advancement provisions contained in the Company Charter Documents as in effect on the date hereof or pursuant to any of those certain indemnification agreements between the Company and each of its non-employee directors as in effect on the date hereof) or capital contributions to, or investment in, any other Person, other than the Company or any of its direct or indirect wholly-owned Subsidiaries, provided, however, that the Company may continue to make investments in accordance with its investment policy as in effect on the date hereof (a copy of which has been made available to Parent), or (iv) other than in the Ordinary Course of Business, enter into any hedging agreement or other financial agreement or arrangement designed to protect the Company or its Subsidiaries against fluctuations in commodities prices or exchange rates;

(i) make any material changes in accounting methods, principles or practices, except insofar as may have been required by a change in GAAP or, except as so required, change any assumption underlying, or method of calculating, any bad debt, contingency or other reserve or estimate;

(j) except as required to comply with applicable Law or agreements, plans or arrangements existing on the date hereof, (i) establish, adopt, enter into, terminate, amend or accelerate any rights under (x) any employment, change in control, severance or similar agreement or Employee Plan (or any plan, agreement, program, or arrangement that would be an Employee Plan if it were in existence as of the date of this Agreement) for the benefit or welfare of any current or former director, officer, employee or consultant or (y) any collective bargaining agreement, (ii) increase the compensation or benefits of, or pay any bonus (including without limitation any transaction-related bonus) to, any directors or officers of the Company or its Subsidiaries, (iii) with respect to non-director and non-officer employees of the Company or its Subsidiaries, (A) increase the compensation to any such Person other than in the Ordinary Course of Business, (B) materially increase any such Person's benefits or (C) pay any bonus to any such Person (including without limitation any transaction-related bonus), (iv) accelerate the payment, right to payment or vesting of any compensation or benefits, including any outstanding options or restricted stock units other than as contemplated by this Agreement, (v) grant any awards under any bonus, incentive, performance or other compensation plan or arrangement or benefit plan, including the grant of stock options, restricted stock units, stock appreciation rights, stock based or stock related awards, performance units or restricted stock, (vi) take any action other than in the Ordinary Course of Business to fund or in any other way secure the payment of compensation or benefits under any Company Employee Plan, (vii) promote any officer or employee, except in the Ordinary Course of Business or as a direct result of the termination or resignation of any officer or employee, or (viii) hire any new employee of the Company or any Subsidiary outside the Ordinary Course of Business;

(k) other than in the Ordinary Course of Business or as required by Applicable Law, (i) make any change (or file any such change) in any method of Tax accounting or any annual Tax accounting period, (ii) make, change or rescind any Tax election, (iii) settle or compromise any Tax Liability or consent to any claim or assessment relating to Taxes, in each case, that require payment by the Company or any of its Subsidiaries in settlement or

compromise of amounts that, individually or in the aggregate, exceed, when coupled with amounts paid in settlement pursuant to the exception set forth in Section 5.1(!)(ii)(A), \$250,000, (iv) file any amended Tax Return or claim for refund, (v) enter into any closing agreement relating to Taxes or (vi) waive or extend the statute of limitations in respect of Taxes, in each case, to the extent that doing so would reasonably be expected to result in a material incremental cost to Parent, the Company or any of their respective Subsidiaries;

- (l) initiate, compromise or settle any material litigation or arbitration proceeding (other than (i) in connection with the enforcement of the Company's rights under this Agreement or (ii) settlements or compromises that (A) require payment by the Company or any of its Subsidiaries in settlement or compromise of amounts that do not, individually or in the aggregate, exceed, when coupled with amounts paid in settlement pursuant to the exception set forth in Section 5.1(k)(iii), \$250,000, and (B) do not (i) involve any material injunctive or equitable relief, (ii) impose restrictions that are material to the business of the Company and its Subsidiaries, taken as a whole, (iii) involve Intellectual Property owned, leased or licensed by the Company or its Subsidiaries or (iv) involve any acquisition or disposition of any business or assets outside the Ordinary Course of Business);
- (m) with respect to Company Intellectual Property, (i) sell, assign, license, sublicense, encumber, impair, abandon, fail to diligently maintain, covenant not to assert, transfer or otherwise dispose of any material right, title or interest of the Company or any of its Subsidiaries in any material Company Intellectual Property, other than granting non-exclusive licenses to Company Intellectual Property in the Ordinary Course of Business, (ii) extend, amend, waive, cancel or modify any rights in or to the material Company Intellectual Property in a manner that is adverse to the Company or its Subsidiaries or other than in the Ordinary Course of Business, (iii) fail to diligently prosecute the material patent applications owned by the Company or any of its Subsidiaries or exclusively licensed to the Company or any of its Subsidiaries and for which the Company or any of its Subsidiaries controls the prosecution thereof as of the date of this Agreement or (iv) divulge, furnish to or make accessible any trade secrets or other confidential information within Company Intellectual Property other than in the Ordinary Course of Business to any third party who is subject to an enforceable written agreement to maintain the confidentiality of such trade secrets or other confidential information;
- (n) fail to use commercially reasonable efforts to maintain (with insurance companies substantially as financially responsible as their existing insurers) insurance in at least such amounts and against at least such risks and losses as are consistent in all material respects with the past practice of the businesses of the Company and its Subsidiaries;
- (o) make any capital expenditures or other expenditures with respect to property, plant or equipment, other than in the Ordinary Course of Business and not exceeding \$200,000 per calendar month;
- (p) engage in any transaction or series of transactions with any Affiliate that would be required to be disclosed under Item 404 of Regulation S-K under the Securities Act, without regard to any monetary thresholds therein;

- (q) adopt a plan of complete or partial liquidation (or resolutions providing for or authorizing such liquidation), dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of the Company or any of its Subsidiaries (other than the Merger) or otherwise enter into any agreements or arrangements imposing material changes or restrictions on the assets, operations or businesses of the Company or its Subsidiaries;
- (r) enter into any agreement, arrangement or commitment that (i) limits or otherwise restricts the Company or any of its Subsidiaries, or that would, after the Effective Time, limit or otherwise restrict Parent or any of its Subsidiaries, from engaging or competing in any line of business or product market or in any geographic area or (ii) obligates the Company or any of its Subsidiaries, or that would, after the Effective Time, obligate Parent or any of its Subsidiaries to provide "most favored customer" pricing of their respective products or services;
- (s) take any action or omit to take any action that is reasonably likely to result in any of the conditions to the Merger set forth in <u>Article VII</u> or any of the conditions to the Offer set forth in <u>Annex I</u> not being satisfied; or
 - (t) authorize any of, or commit or agree, in writing or otherwise, to take any of, the foregoing actions.
- 5.2 **Confidentiality**. The parties acknowledge that Parent and the Company have previously executed the Confidentiality Agreement, which Confidentiality Agreement shall continue in full force and effect in accordance with its terms, except as expressly waived or modified as provided herein or therein.
- 5.3 Access and Investigation. During the Pre-Closing Period, upon reasonable advance written notice to the Company, the Company shall, and shall cause its Representatives, its Subsidiaries and its Subsidiaries' Representatives to: (i) provide Parent and Parent's Representatives with reasonable access during the normal business hours of the Company to the Company's Representatives, personnel, Real Property, officers, employees, agents, offices and other properties, and assets and to all existing books, records, Contracts, Tax Returns, work papers and other documents and information relating to the Company and its Subsidiaries, and (ii) promptly provide Parent and Parent's Representatives with all reasonably requested information regarding the business of the Company and its Subsidiaries, including copies of the existing books, records, Tax Returns, work papers and other documents and information relating to the Company and its Subsidiaries, and with any available or reasonably producible additional financial, operating and other data and information regarding the Company and its Subsidiaries, as Parent may reasonably request; provided, however, that any such access shall be conducted at Parent's expense, at a reasonable time, under the supervision of appropriate personnel of the Company and in such a manner as not to unreasonably interfere with the normal operation of the business of the Company and its Subsidiaries. Nothing herein shall require the Company to disclose (A) any information concerning Acquisition Proposals, which shall be governed by Section 5.4. (B) any information regarding the deliberations of the Company Board with respect to this Agreement or the transactions contemplated hereby or any similar transaction or transactions with any other Person, or any materials provided to the Company Board in connection therewith, or (C) any information, that in the reasonable good faith judgement of the

Company would (i) jeopardize any attorney-client or other legal privilege (so long as the Company has reasonably cooperated with Parent to permit such inspection of or to disclose such information on a basis that does not waive such privilege with respect thereto), (ii) contravene any applicable Law, fiduciary duty or Contract entered into prior to the date of this Agreement (including any confidentiality agreement to which the Company or its Affiliates is a party, so long as the Company has used commercially reasonable efforts to make appropriate substitute arrangements to permit reasonable disclosure not in violation of such Law, agreement or duty), or (iii) unreasonably disrupt the operations of the Company and its Subsidiaries; provided, further, that information described in the foregoing clause (C)(i), shall be disclosed subject to execution of a joint defense agreement in customary form, and disclosure may be limited to external counsel for Parent, to the extent the Company determines doing so may be reasonably required for the purpose of complying with applicable Antitrust Laws. With respect to the information disclosed pursuant to this Section 5.3 to Parent and its Representatives, Parent shall comply with, and shall cause Parent's Representatives to comply with, all of its obligations under the Confidentiality Agreement. All requests for information made pursuant to this Section 5.3 shall be directed to the executive officer or other Person designated by the Company. No investigation shall affect the Company's representations and warranties contained herein, or limit or otherwise affect the remedies available to Parent or Purchaser pursuant to this Agreement.

5.4 No Solicitation.

(a) No Solicitation.

- (i) Neither the Company nor any of its Subsidiaries shall, and the Company shall cause its and its Subsidiaries' Representatives not to, directly or indirectly:
- (A) solicit, initiate or knowingly encourage any inquiries or the making of any proposal or offer that constitutes, or could reasonably be expected to lead to, any Acquisition Proposal; or
- (B) other than directing a Person or "group" (as defined under Section 13(d) of the Exchange Act) of Persons that has made an Acquisition Proposal to a copy of this Agreement filed by Parent or the Company with the SEC, enter into, continue or otherwise participate in any discussions or negotiations regarding, or furnish to any person any non-public information that encourages or facilitates, or could reasonably be expected to lead to, any Acquisition Proposal or otherwise knowingly facilitate any effort or attempt to make any Acquisition Proposal, it being agreed that the furnishing to any person of non-public information unrelated to an Acquisition Proposal in the Ordinary Course of Business shall not be a violation of this Section 5.4(a)(i)(B).
- (ii) Notwithstanding anything to the contrary set forth in this Agreement, at any time prior to the Offer Acceptance Time, the Company may, in response to an Acquisition Proposal that (i) is made or received after the date hereof, (ii) did not arise or result from a breach of this **Section 5.4** and (iii) the Company Board determines, in good faith after consultation with outside counsel and its financial advisor,

is a Superior Proposal or an Acquisition Proposal that is reasonably likely to lead to a Superior Proposal, if the Company Board determines, in good faith, that to act otherwise would be inconsistent with its fiduciary duties, after providing Parent with at least concurrent written notice that it is taking any action in the following clauses (x), (y) or (z), (x) furnish information with respect to the Company and its Subsidiaries to the person or "group" (as defined under Section 13(d) of the Exchange Act) of persons making such Acquisition Proposal and its Representatives pursuant to a confidentiality agreement that is not materially less restrictive of the other party than the Confidentiality Agreement and does not restrict the Company from performing its obligations to Purchaser and Parent hereunder (provided, however, the Company shall provide or make such information available to Parent prior to or at the same time as the furnishing information pursuant hereto unless previously provided or made available to Parent), (y) participate in discussions or negotiations (including solicitation of a revised Acquisition Proposal) with such person or group and its Representatives regarding any Acquisition Proposal and (z) amend, or grant a waiver or release under, any standstill or similar agreement with respect to any Company Common Stock with the person making such Acquisition Proposal. Upon execution of this Agreement, the Company shall immediately (A) cease and cause to be terminated any negotiations with any Persons (other than Parent and Purchaser and their Representatives) that may be ongoing with respect to an Acquisition Proposal, (B) cease providing any information to any such Person or its Representatives and terminate all access granted to any such Person and its Representatives to any physical or electronic dataroom and (C) request that any such Person and its Representatives promptly return to the Company or destroy any non-public information concerning the Company or its Subsidiaries that was previousl

(b) No Change in Recommendation or Alternative Acquisition Agreement.

(i) During the Pre-Closing Period, none of the Company, the Company Board or any committee thereof shall (A) withhold, fail to include in (or remove from) the Schedule 14D-9, withdraw, adversely qualify or modify (or resolve, determine or publicly propose to do any of the foregoing) the Company Board Recommendation, (B) following the date any Acquisition Proposal (or any material modification thereto) is made or commenced, or an intention to make or commence any Acquisition Proposal (or any material modification thereto) is publicly proposed or announced, fail to reaffirm (publicly, if so requested) the Company Board Recommendation, within five (5) business days after a request by Parent to do so (or, if earlier, by the second business day prior to the then-scheduled Expiration Date of the Offer), (C) fail to publicly recommend against acceptance of any tender offer or exchange offer (other than the Offer or any other tender offer or exchange offer by Parent or Purchaser or any of their Affiliates) for the Shares within ten (10) business days after the commencement of such offer, (D) adopt, approve, recommend, submit to the vote of securityholders or declare advisable any Acquisition Proposal or the entry into any Company Acquisition Agreement (or resolve, determine or publicly propose to do any of the foregoing), or (E) approve any transaction under, or any transaction resulting in any Third Party becoming an "interested stockholder" under, Section 203 of the DGCL (or resolve, determine or publicly propose to do any of the foregoing) (any action described in clauses (A) through (E) being referred to as an "Adverse Change Recommendation"), except as expressly permitted by this Section 5.4(b).

(ii) Notwithstanding anything to the contrary contained in this Agreement, at any time prior to the Offer Acceptance Time, the Company Board may make an Adverse Change Recommendation (other than as reflected in clause (E) of the definition thereof, unless such Adverse Change Recommendation referred to in such clause (E) is made with respect to an Acquisition Proposal that is reasonably likely to lead to a Superior Proposal) or, subject to the prior or concurrent payment of the Termination Fee in accordance with <u>Section 8.3(b)</u>, terminate this Agreement pursuant to <u>Section 8.1(e)</u> to enter into a Company Acquisition Agreement in respect of a Superior Proposal, if and only if:

(1) such Superior Proposal did not arise or result from a breach of this **Section 5.4**;

(2) the Company Board determines in good faith, after consultation with the Company's financial advisors and outside legal counsel, that the failure to make such Adverse Change Recommendation or terminate this Agreement to enter into a Company Acquisition Agreement would be inconsistent with the fiduciary duties of the Company Board under applicable Law;

(3) Parent shall have received from the Company prior written notice of the Company's intention to make such an Adverse Change Recommendation or to terminate this Agreement to enter into any such Company Acquisition Agreement at least three (3) business days prior to making any such Adverse Change Recommendation or terminating this Agreement to enter into a Company Acquisition Agreement (a "Change of Status Notice"), which Change of Status Notice shall be accompanied by unredacted copies of all draft agreements embodying such Superior Proposal (other than financing-related documents that may be provided in redacted form) and, unless reflected in such draft agreements, a description of the terms and conditions of such Superior Proposal, it being understood that such Change of Status Notice shall not, in and of itself, constitute an Adverse Change Recommendation; and

(4) the Company shall have complied with clauses (A) through (C) (as applicable) as follows and the conditions described in clause (C) shall have been satisfied:

(A) prior to giving effect to clauses (B) and (C), the Company Board shall have determined in good faith, after consultation with the Company's financial advisors and outside legal counsel, that such Acquisition Proposal constitutes a Superior Proposal and that failure to terminate this Agreement or to make an Adverse Change Recommendation would be inconsistent with its fiduciary duties;

(B) the Company shall have complied with this <u>Section 5.4</u> with respect to such Acquisition Proposal (other than non-compliance that is immaterial in significance and inadvertent); and

(C) Parent shall not have made, within three (3) business days of its receipt of the Change of Status Notice (the "Notice Period"), a written offer or proposal that remains outstanding and unmodified as of the end of the Notice Period and that, upon binding acceptance by the Company, is binding on Parent to revise the terms of this Agreement (any such offer, a "Revised Transaction Proposal") in a manner that the Company Board determines in good faith, after consulting with its financial advisors and outside legal counsel, is reasonably likely to be consummated if accepted on the terms thereof and is at least as favorable to the Company's stockholders from a financial perspective as such Superior Proposal; provided, however, that, during the Notice Period, the Company shall (and shall cause its Representatives to) negotiate in good faith with Parent (to the extent Parent desires to negotiate and makes itself reasonably available to negotiate) regarding any Revised Transaction Proposal; provided, further, that any amendment to the financial terms of such Superior Proposal or any other material amendment to the terms of such Superior Proposal during the Notice Period shall require the Company to deliver a new Change of Status Notice with respect to such amended Superior Proposal and the Notice Period shall expire three (3) business days subsequent to delivery of such new Change of Status Notice, including with respect to the Company's obligations to negotiate in good faith with Parent and to cause its Representatives to negotiate in good faith with Parent.

(iii) Notwithstanding anything to the contrary herein, at any time prior to the Offer Acceptance Time, the Company Board may make an Adverse Change Recommendation (as set forth in clause (A) thereof) with respect to an Intervening Event, if and only if:

(1) the Company Board determines in good faith, after consultation with the Company's outside legal counsel, that the failure to make such Adverse Change Recommendation would be inconsistent with the fiduciary duties of the Company Board under applicable Law;

(2) Parent shall have received from the Company a Change of Status Notice at least three (3) business days prior to making any such Adverse Change Recommendation (the "Intervening Event Notice Period"), which Change of Status Notice shall be accompanied by a reasonably detailed description of the Intervening Event, a description of the bases for the conclusion that the events described in such Change of Status Notice constitute an Intervening Event and unredacted copies of any material documentation relating thereto, it being understood that such Change of Status Notice shall not, in and of itself, constitute an Adverse Change Recommendation;

(3) during the Intervening Event Notice Period, the Company shall have negotiated in good faith with Parent (and caused its Representatives to negotiate with Parent), to the extent Parent desires to negotiate and makes itself reasonably available to negotiate, with respect to any proposed revisions to this Agreement or other proposals made by Parent, if any, that would obviate the need to make such an Adverse Change Recommendation; and

(4) after considering the results of negotiations with Parent and taking into account the proposals made by Parent, if any, after consultation with its outside legal counsel and the Company's financial advisors, the Company Board shall have determined in good faith that the failure to make such Adverse Change Recommendation would be inconsistent with the fiduciary duties of the Company Board under applicable Law.

In the event of a material change to the facts and circumstances relating to an Intervening Event in respect of which the Company shall have previously provided to Parent a Change of Status Notice, the provisions of this <u>Section 5.4(b)(iii)</u> shall also apply to each subsequent material change to the facts and circumstances relating to an Intervening Event (in which case each such material change shall require a new Change of Status Notice and the Company shall be required to comply again with the provisions of this <u>Section 5.4(b)(iii)</u>), provided that the Intervening Event Notice Period shall expire three (3) business days subsequent to delivery of such new Change of Status Notice to the extent the Intervening Event Notice Period would otherwise have expired prior to three (3) business days subsequent to delivery of such new Change of Status Notice.

(c) Notices to Parent. The Company shall as promptly as practicable (and in any event within 48 hours) notify Parent in writing of receipt by the Company of any Acquisition Proposal or any request for nonpublic information, in each case on or subsequent to the date hereof, in connection with, or that could reasonably be expected to lead to, any Acquisition Proposal, the material terms and conditions of any such Acquisition Proposal or request and the identity of the person making any such Acquisition Proposal or request. Such notice shall be accompanied by unredacted copies of any draft agreements embodying any such Acquisition Proposal. In addition to any other notifications required by this Section 5.4, the Company shall endeavor in good faith to provide Parent with updates regarding any material developments (and unredacted copies of any draft agreements relating thereto, other than financing-related documents that may be provided in redacted form) with respect to any Acquisition Proposal identified by the Company in accordance with this Section 5.4(c). The Company shall promptly (and in any event within two business days) notify Parent of the receipt by the Company of any material modification to any such Acquisition Proposal, unless prohibited by any Antitrust Laws or comparable Laws.

(d) <u>Certain Permitted Disclosure</u>. Nothing contained in this <u>Section 5.4</u> or elsewhere in this Agreement shall prohibit the Company from (i) taking and disclosing a position contemplated by Rule 14d-9, Rule 14e-2(a) or Item 1012(a) of Regulation M-A promulgated under the Exchange Act, (ii) making any disclosure to the Company's stockholders if, in the good faith judgment of the Company Board or any committee thereof, after consultation with outside legal counsel, the failure to do so would be inconsistent with the fiduciary duties of the Company Board under applicable Law or any disclosure requirements under applicable Law, or (iii) making any disclosure that constitutes a stop, look and listen communication or similar communication of the type contemplated by Section 14d-9(f) promulgated under the Exchange Act; provided, however, that (1) in no event shall this <u>Section 5.4(d)</u> affect the Company's obligations specified in <u>Sections 5.4(a)</u> and <u>Section 5.4(b)</u> and (2) any such disclosure (other than issuance by the Company of a "stop, look and listen" or similar communication of the type contemplated by Rule 14d-9(f) under the Exchange Act) that addresses or relates to the approval, recommendation or declaration of advisability by the

Company Board with respect to this Agreement or an Acquisition Proposal shall be deemed to be an Adverse Change Recommendation unless the Company Board in connection with such communication publicly states that its recommendation with respect to this Agreement has not changed or refers to the prior recommendation of the Company Board, without disclosing any Adverse Change Recommendation.

(e) <u>Breach</u>. The Company agrees that any breach of this <u>Section 5.4</u> by any of its Subsidiaries or its or their respective Representatives shall be deemed to be a breach of this <u>Section 5.4</u> by the Company.

ARTICLE VI ADDITIONAL COVENANTS OF THE PARTIES

6.1 Filings and Approvals.

(a) Subject to the terms and conditions of this Agreement, Parent, Purchaser and the Company shall cooperate with each other and use (and shall cause their respective Subsidiaries to use) reasonable best efforts to take, or cause to be taken, all actions, and do, or cause to be done, all things, reasonably necessary, proper or advisable on its part under this Agreement and applicable Law to consummate and make effective the Offer, the Merger and the other transactions contemplated hereby as promptly as practicable (and in any event no later than the End Date). Without limiting the generality of the foregoing, each party to this Agreement shall (and shall cause their respective Subsidiaries to) (i) use reasonable best efforts to prepare and make all filings and obtain as soon as practicable all Consents, waivers and expirations of waiting periods (if any) from any Governmental Body and make all registrations, declarations and filings with, or notices to, Governmental Bodies, in each case, in connection with the Offer and the Merger, that are necessary or required to be obtained in order to consummate the transactions contemplated hereby, and (ii) use reasonable best efforts to give any notices to third parties specified on Schedule 6.1(a)(ii) and to obtain as soon as practicable each other Consent or waiver (if any) of any third party (other than from any Governmental Body, which are addressed in the preceding clause (i)) specified on Schedule 6.1(a)(ii); provided, that in no event will the Company, Parent or any of their respective Subsidiaries be obligated to (and in no event shall the Company or any of its Subsidiaries, without Parent's prior written consent) pay or commit to pay, whether prior to or after the Offer Acceptance Time, any fee, penalty or other consideration to any third party to obtain any Consent or waiver pursuant to this **Section 6.1**. The Company and Parent will consult and cooperate with one another, and will consider in good faith the views of one another, in connection with any actions described in this Section 6.1. Subject to applicable Laws relating to the exchange of information, outside legal counsel for each party shall have the right to review in advance and, to the extent practicable, outside legal counsel for each party will consult with the other party's outside legal counsel on and consider in good faith the views of the other party's outside legal counsel in connection with, all of the information relating to such party, and any of their respective Subsidiaries, that appears in any filing made with or written materials submitted to, any third party and/or any Governmental Body in connection with the Offer, the Merger and the other transactions contemplated hereby; provided however that such sharing of information with respect to the HSR Notification Form shall be governed by Section 6.1(c). In exercising the foregoing rights, each of the Company and Parent and their respective outside legal counsel shall act reasonably and as promptly as practicable.

(b) Without limiting the generality of Section 6.1(a), each of Parent and the Company shall (i) file as promptly as practicable after the date of this Agreement (but in no event later than the tenth (10th) business day after the date hereof) and not withdraw (other than in connection with the termination of this Agreement) any notification and report forms and related material required to be filed by it with the FTC and the DOJ, as applicable, pursuant to the HSR Act, together with a request for early termination of the applicable waiting period under the HSR Act and (ii) as promptly as practicable after the date of this Agreement, make any filings required to be made by it with any Governmental Body under any other applicable Antitrust Laws, and, in any such case, shall promptly make any further filings pursuant thereto that may be necessary.

(c) Subject to applicable Laws, each of Parent and the Company will promptly furnish outside legal counsel for the other party (i) all necessary information concerning itself, its Subsidiaries, directors, officers and stockholders and such other matters as may be reasonably necessary in connection with any statement, filing, notice or application made by or on behalf of Parent, the Company or any of their respective Subsidiaries to any third party and/or Governmental Body in connection with the Offer or the Merger, including in connection with the preparation of any filing or submission pursuant to the HSR Act or other Antitrust Laws, (ii) copies of any such filings or submissions, but not including the HSR Notification Form or the Item 4(c) or Item 4(d) documents filed therewith, and (iii) copies of all substantive written communications (and shall inform the other of any oral communications) between them or any of their Representatives, on the one hand, and any Governmental Body with jurisdiction over enforcement of any applicable Antitrust Laws (a "Governmental Antitrust Authority") or members of its staff, on the other hand, in connection with this Agreement and the transactions contemplated hereby; provided, however, that materials may be redacted as necessary to comply with applicable Laws. Prior to making any written communications to a Governmental Antitrust Authority in connection with this Agreement and the transactions contemplated hereby, the communicating party shall provide outside legal counsel for the other party with a copy of the intended communication, outside legal counsel for the other party shall have a reasonable period of time to review and comment on the communication, and the communicating party shall consider any such comments in good faith. Each party shall promptly provide to each Governmental Antitrust Authority all non-privileged information and documents that are necessary, proper or advisable to permit consummation of the transactions contemplated hereby and if Parent or the Company (or any of their respective Affiliates) receives a request for additional non-privileged information or documentary material from a Governmental Antitrust Authority that is related to the transactions contemplated hereby, then such party shall endeavor in good faith to make, or cause to be made, after consultation with the other party, an appropriate response to such request and shall supply as promptly as reasonably practicable any additional information or documentary material that may be requested pursuant to applicable Law. No party shall participate in any meeting or engage in any substantive conversation with any Governmental Antitrust Authority in respect of any filings, investigation (including any settlement of the investigation), litigation, proceeding or other inquiry relating to the transactions contemplated hereby without consulting with the other party in advance, giving the other party prior notice of the meeting or conversation and, unless prohibited by such Governmental

Antitrust Authority or applicable Law, the opportunity to attend or participate. In furtherance and not in limitation of the foregoing, each of Parent and the Company shall use reasonable best efforts to resolve any objections that may be asserted with respect to the transactions contemplated hereby under any Antitrust Law as promptly as practicable.

- (d) Notwithstanding anything to the contrary in this Agreement, Parent shall not have any obligation to proffer to, or agree or consent to (or to cause any of its Subsidiaries or Affiliates or the Company (including, after the Effective Time, the Surviving Corporation) or its Subsidiaries or Affiliates to proffer to, or agree or consent to) (i) sell, license, lease, transfer, divest, dispose of or otherwise encumber or hold separate any entities, assets, licenses, operations, rights, product lines, Intellectual Property or businesses or interests of Parent, the Company (including, after the Effective Time, the Surviving Corporation), or any of their respective Subsidiaries or Affiliates, (ii) terminate, amend or assign any existing relationships or contractual rights or obligations of Parent, the Company (including, after the Effective Time, the Surviving Corporation), or any of their respective Subsidiaries or Affiliates, (iii) change or modify any course of conduct regarding future operations or agree to any changes (including through a licensing arrangement) or restriction on, or other impairment of Parent's ability to own or operate, any of the assets, licenses, operations, rights, product lines, Intellectual Property, or businesses or interests of Parent, the Company (including, after the Effective Time, the Surviving Corporation), or any of their respective Subsidiaries or Affiliates, or Parent's ability to vote, transfer, receive dividends or otherwise exercise full ownership rights with respect to the stock of the Company (including, after the Effective Time, the Surviving Corporation), (iv) otherwise take any action that would limit the freedom of action of Parent, the Company (including, after the Effective Time, the Surviving Corporation), or any of their respective Subsidiaries or Affiliates with respect to, or the ability of Parent, the Company (including, after the Effective Time, the Surviving Corporation), or any of their respective Subsidiaries or Affiliates to retain, one or more of the businesses, assets or rights or interests of Parent, the Company (including, after the Effective Time, the Surviving Corporation), or any of their respective Subsidiaries or Affiliates; or (v) commit or agree to take any such action in the foregoing clauses (i), (ii), (iii) or (iv), or consent to the Company's or any of the Company's Subsidiaries or Affiliates taking any of the foregoing actions, whether before or after the Offer Acceptance Time. For the avoidance of doubt, Parent shall not require the Company to, and the Company shall not be required to, take any action with respect to any Order or any applicable Law unless contingent upon the occurrence of the transactions contemplated hereby. With respect to any Governmental Body in connection with the transactions contemplated hereby, neither the Company nor its Subsidiaries shall, without Parent's prior written consent in Parent's sole discretion, discuss or commit to any action referred to in clauses (i) through (iv) above.
- (e) Nothing in this Agreement shall require, or be construed to require, Parent or any of its Affiliates to take any other action under this <u>Section 6.1</u> if the DOJ or the FTC authorizes its staff to seek a preliminary injunction or restraining order to enjoin the consummation of the Transactions.
- (f) No party hereto shall take any action that is intended to or would reasonably be expected to adversely affect or materially delay its ability or the ability of any other party hereto to perform its covenants and agreements under this Agreement or to consummate the Offer, the Merger and the other transactions contemplated by this Agreement.

6.2 Employee Benefits.

(a) For the six (6) month period immediately following the Effective Time, Parent will use commercially reasonable efforts to provide to employees of the Surviving Corporation or their respective Subsidiaries who shall have been employees of the Company or any of its Subsidiaries immediately prior to the Effective Time and who shall continue to be employed by the Surviving Corporation ("Continuing Employees"), base salary or wage level and annual target bonus opportunities (excluding equity-based compensation and, for the avoidance of doubt, excluding any retiree healthcare or defined benefit retirement benefits) that, in the aggregate, are substantially similar to the annual base salary or wage level and annual target bonus opportunities (excluding equity-based compensation and, for the avoidance of doubt, excluding any retiree healthcare or defined benefit retirement benefits) provided to those employees immediately prior to the execution of this Agreement.

(b) With respect to any accrued but unused personal, sick or vacation time to which any Continuing Employee is entitled pursuant to the personal, sick or vacation policies applicable to such Continuing Employee immediately prior to the Effective Time (the "PSV Policies"), to the extent accrued in the most recent consolidated financial statements of the Company filed with the SEC and since the date of such financial statements in the Ordinary Course of Business, Parent shall assume the liability for such accrued personal, sick or vacation time and allow such Continuing Employee to use such accrued personal, sick or vacation time, subject to the terms and conditions of such PSV Policies (as such policies are in effect on the date hereof or are amended or modified with Parent's consent). Parent shall be permitted to terminate, amend or modify the PSV Policies at any time, in its sole discretion.

(c) Following the Effective Time, Parent will use commercially reasonable efforts to give each Continuing Employee credit for prior service with the Company or its Subsidiaries under Parent Employee Plans (as defined below) in which such Continuing Employees are eligible to participate, if any, for purposes of eligibility and vesting (excluding defined benefit plans and retiree plans), but not for purposes of benefit accrual (except for paid time off, to the extent applicable) to the same extent as such Continuing Employee was entitled immediately prior to the Effective Time to credit for such service under any similar Employee Plan or if such service is recognized under a Parent Employee Plan providing severance or retirement benefits for similarly situated employees of Parent; provided, however, such service shall not be recognized to the extent that (x) such recognition would result in a duplication of benefits with respect to the same period of service, (y) such service was not recognized under the corresponding Employee Plan or (z) such service with respect to a newly established Parent Employee Plan for which similarly-situated employees of Parent do not receive past service credit. In addition, Parent shall use commercially reasonable efforts to waive, or cause to be waived, any limitations on benefits relating to pre-existing conditions to the same extent such limitations are waived under any comparable plan of the Parent and recognize for purposes of annual deductible and out-of-pocket limits under its medical and dental plans, deductible and out-of-pocket expenses paid by Continuing Employees in the calendar year in which the Effective Time occurs, in each case, to the extent Continuing Employees participate in any such comparable plan of the Parent. For purposes of this Agreement, the term "Parent Employee Plan" means any "employee pension benefit plan" (as defined in Section 3(2) of ERISA), and any "employee welfare benefit plan" (as defined in Section 3(1) of ERISA), for the benefit of, or relating

(d) This <u>Section 6.2</u> shall be binding upon and inure solely to the benefit of each of the parties to this Agreement, and nothing in this <u>Section 6.2</u> or implied, shall confer upon any other Person any third-party beneficiary rights or remedies of any nature whatsoever under or by reason of this <u>Section 6.2</u> or inure to the benefit of or be enforceable by any employee, director or independent consultant of the Company, Parent and their Subsidiaries, of any entity or any Person representing the interest of any employees, directors or independent consultants or of any Person whose rights are derivative of any such employee (including a family member or estate of the employee). Nothing contained herein, express or implied shall (i) be construed to establish, amend or modify any Employee Plan or other benefit plan, program, agreement or arrangement or any of their respective Affiliates to amend, modify or terminate any benefit plan, program, agreement or arrangement at any time assumed, established, sponsored or maintained by any of them. The parties hereto acknowledge and agree that the terms set forth in this <u>Section 6.2</u> shall not create any right in any Company employee or any other Person to any continued employment with the Surviving Corporation, Parent or any of their respective Subsidiaries or compensation or benefits of any nature or kind whatsoever.

(e) Notwithstanding the foregoing, nothing contained in this Agreement will be treated as an amendment of any particular Employee Plan.

6.3 Indemnification of Officers and Directors.

(a) From the Effective Time, each of Parent and the Surviving Corporation shall, jointly and severally, indemnify and hold harmless each person who is now, or has been at any time prior to the date hereof, or who becomes prior to the Effective Time, a director or officer of the Company or any of its Subsidiaries (the "Indemnified Parties"), against all claims, losses, liabilities, damages, judgments, fines and reasonable fees, costs and expenses, including attorneys' fees and disbursements (collectively, "Costs"), incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to the fact that the Indemnified Party is or was an officer or director of the Company or any of its Subsidiaries, whether asserted or claimed prior to, at or after the Effective Time, to the fullest extent permitted under the DGCL for officers and directors of Delaware corporations. Each Indemnified Party will be entitled, subject to applicable law, to advancement of expenses incurred in the defense of any such claim, action, suit, proceeding or investigation from each of Parent and the Surviving Corporation within ten (10) Business Days of receipt by Parent or the Surviving Corporation from the Indemnified Party of a request therefor; provided that any person to whom expenses are advanced provides a written personal undertaking to repay such advances if it is determined that such person is not entitled to indemnification. In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination shall presume that each Indemnified Party is entitled to indemnification and the other benefits of this Section 6.3.

- (b) The Certificate of Incorporation and By-laws, or corresponding governing documents, of the Surviving Corporation and its Subsidiaries shall contain, and Parent shall cause the Certificate of Incorporation and By-laws, or corresponding governing documents, of the Surviving Corporation and its Subsidiaries to so contain, provisions no less favorable with respect to indemnification, advancement of expenses and exculpation of present and former directors and officers of the Company and its Subsidiaries than are presently set forth in the certificate of incorporation and by-laws of the Company and its Subsidiaries, as applicable, which provisions shall not be amended, modified or repealed for a period of six (6) years from the Effective Time in a manner that would adversely affect the rights thereunder of individuals who, at or prior to the Effective Time, were officers or directors of the Company or any of its Subsidiaries.
- (c) Prior to the Effective Time, the Company shall (in consultation with Parent) purchase, as of the Effective Time, tail insurance, at commercially reasonable rates and containing a customary straddle claim provision, for (i) the directors' and officers' liability coverage of the Company's existing directors' and officers' liability insurance policy covering acts, errors or omissions occurring at or prior to the Effective Time and (ii) the Company's existing fiduciary liability insurance policies (collectively, "D&O Insurance"), for a claims reporting or discovery period of six (6) years after the Effective Time (the "Tail Period"), with respect to those Persons who are covered by the D&O Insurance as of the Effective Time, with terms and conditions, retentions and limits of liability under the existing D&O Insurance with respect to acts, errors or omissions occurring at or prior to the Effective Time.
- (d) Parent shall pay all expenses, including reasonable attorneys' fees, that may be incurred by the Indemnified Parties in connection with their enforcement of their rights provided in this **Section 6.3**.
- (e) The provisions of this <u>Section 6.3</u> are intended to be in addition to the rights otherwise available to the Indemnified Parties by law, certificate of incorporation, statute, by-law or agreement, and shall operate for the benefit of, and shall be enforceable by, each of the Indemnified Parties, their heirs and their representatives. Notwithstanding any other rights an Indemnified Party may have, the obligations of Parent, the Surviving Corporation and its Subsidiaries specified herein are to be the primary source of indemnification and advancement of expenses for such Indemnified Party.
- (f) In the event Parent, the Surviving Corporation or a Subsidiary of the Surviving Corporation or any of their respective successors or assigns (i) consolidates with or merges into any other person and shall not be the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfers all or substantially all of its properties and assets to any person, then, and in each such case, proper provision shall be made so that the successors and assigns of Parent, the Surviving Corporation or the Subsidiary of the Surviving Corporation, as the case may be, shall succeed to the obligations set forth in this Section 6.3.
- (g) Parent shall cause the Surviving Corporation and its Subsidiaries to perform all of the obligations of the Surviving Corporation or the Subsidiary, as applicable, under this **Section 6.3**.

- 6.4 **Notification of Certain Matters**. During the Pre-Closing Period, Parent shall give prompt notice to the Company, and the Company shall give prompt notice to Parent, of (a) the occurrence, or failure to occur, of any event, which occurrence or failure to occur is reasonably likely to cause any representation or warranty of such party contained in this Agreement to be untrue or inaccurate in any material respect, in each case at any time from and after the date of this Agreement until the Effective Time, or (b) any material failure of Parent and Purchaser or the Company, as the case may be, or of any officer, director, employee or agent thereof, to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it under this Agreement. Notwithstanding the above, the delivery of any notice pursuant to this Section will not limit or otherwise affect the remedies available hereunder to the party receiving such notice or the conditions to such party's obligation to consummate the Merger.
- 6.5 **Stockholder Litigation**. The Company shall provide Parent with prompt notice of any Stockholder Litigation brought, asserted or commenced by, on behalf of or in the name of, against or otherwise involving the Company or any of its Subsidiaries, the Company Board or any of its Subsidiaries, any committee thereof and/or any of the Company's directors or officers relating to the Offer, the Merger, this Agreement or any of the transactions contemplated by this Agreement, and shall keep Parent informed on a reasonably prompt basis with respect to the status thereof. The Company shall give Parent the opportunity to participate (at Parent's expense) in the defense or settlement of any such litigation, and except for settlements that are (x) solely for monetary damages entirely covered within the limits of the Company's and the Company Subsidiaries' insurance policies (other than the deductible under such insurance policies) or (y) to provide additional disclosure in the Schedule 14d-9, no such settlement shall be agreed to without Parent's prior written consent which may be withheld in Parent's discretion.
- 6.6 **Disclosure**. Parent and the Company have agreed upon and approved their respective press releases with respect to the execution of this Agreement. Thereafter Parent and the Company shall consult with each other before issuing any further press release(s) or otherwise making any public statement (to the extent not previously issued or made in accordance with this Agreement) with respect to the Offer, the Merger, this Agreement or any of the other transactions contemplated by this Agreement and shall not issue any such press release or public statement without the other party's written consent, which shall not be unreasonably withheld, delayed or conditioned. Notwithstanding the foregoing: (a) each of the Company and Parent may, without such consultation or consent, make any public statement in response to questions from the press, analysts, investors or those attending industry conferences, make internal announcements to employees and make disclosures in Company SEC Documents or in documents filed by Parent with the SEC, so long as such statements are consistent with previous press releases, public disclosures or public statements made jointly by the parties (or individually, if approved by the other party); (b) each of the Company and Parent may, without the prior consent of the other party but subject to giving advance notice to the other party, issue any such press release or make any such public announcement or statement as may be required by Law (including applicable stock exchange rules); and (c) the Company need not consult with Parent in connection with such portion of any press release, public statement or filing to be issued or made with respect to any Adverse Change Recommendation effected in accordance with the provisions of this Agreement.
- 6.7 **Section 16 Matters**. Promptly after the date hereof and prior to the Expiration Date, the Company shall take all such steps as may be required to cause any dispositions of Shares resulting from the transactions contemplated by this Agreement by each individual who is subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to the Company to be exempt under Rule 16b-3 promulgated under the Exchange Act, to the extent permitted by applicable Law.

- 6.8 **Stock Exchange Delisting; Deregistration**. Prior to the Effective Time, the Company shall cooperate with Parent and use commercially reasonable efforts to take, or cause to be taken, all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under applicable Law and rules and policies of NASDAQ to enable the delisting by the Surviving Corporation of the Shares from NASDAQ and the deregistration of the Shares under the Exchange Act as promptly as practicable after the Effective Time.
- 6.9 **Transfer Taxes**. Except as otherwise provided in **Section 2.7(b)**, all stock transfer, real estate transfer, documentary, stamp, recording and other similar Taxes (including interest, penalties and additions to any such Taxes) ("**Transfer Taxes**") incurred in connection with the transactions contemplated by this Agreement shall be paid by either Purchaser or the Surviving Corporation, and the Company shall cooperate with Purchaser and Parent in preparing, executing and timely filing any Tax Returns with respect to such Transfer Taxes.
- 6.10 **Rule 14d-10 Matters**. Prior to the Offer Acceptance Time and to the extent permitted by applicable Law, the Compensation and Stock Committee of the Company Board (the "**Compensation Committee**") will cause each Employee Plan and Company Employee Agreement pursuant to which consideration is payable to any officer, director or employee who is a holder of any security of the Company to be approved by the Compensation Committee (comprised solely of "independent directors") in accordance with the requirements of Rule 14d-10(d)(2) under the Exchange Act and the instructions thereto as an "employment compensation, severance or other employee benefit arrangement" within the meaning of Rule 14d-10(d)(2) under the Exchange Act and satisfy the requirements of the non-exclusive safe harbor set forth in Rule 14d-10(d) of the Exchange Act.
- 6.11 **Takeover Statutes**. If any state takeover Law or other state Law that purports to limit or restrict business combinations or the ability to acquire or vote shares of Company Common Stock (including any "control share acquisition," "fair price," "business combination" or other similar takeover Applicable Law ("**Takeover Law**"), including the restrictions on business combinations in Section 203 of the DGCL, becomes or is deemed to be applicable to the Company, Parent or Purchaser, the Merger or any other transaction contemplated hereby, and the representation and warranty of Parent and Purchaser in <u>Section 4.7</u> is true and correct, then the Company and the Company Board, as applicable, shall use their reasonable best efforts to take such actions within their control as permitted under applicable Law to (i) ensure that the transaction contemplated hereby, including the Merger, may be consummated as promptly as practicable on the terms contemplated herein and (ii) otherwise act to eliminate, or if not possible minimize to the maximum extent possible, the effects of such Law on this Agreement and the transactions contemplated herein.
- 6.12 **Control of the Company's Business**. Nothing contained in this Agreement shall give Parent or Purchaser, directly or indirectly, the right to control or direct any of the operations of the Company or its Subsidiaries prior to the Closing. Prior to the Closing, the Company shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its and its Subsidiaries' respective operations.

6.13 **Non-USRPHC Certificate and Notice.** The Company shall furnish to Parent on or before the Closing Date a certification dated as of the Closing Date, sworn under penalty of perjury and in form and substance required under the Treasury Regulations issued pursuant to Section 1445(b) (3) of the Code, stating that the Company is not and has not been a "United States real property holding corporation" (as defined in Section 897(c)(2) of the Code) during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code and that no interest in the Company is a "United States real property interest" within the meaning of Section 897(c) of the Code and proof reasonably satisfactory to Parent that the Company has provided notice of such certificate to the IRS in accordance with the provisions of Treasury Regulations Section 1.897-2(h)(2).

ARTICLE VII

CONDITIONS PRECEDENT TO THE MERGER

The obligations of the parties to effect the Merger are subject to the satisfaction, at or prior to the Closing, of each of the following conditions:

- 7.1 **No Restraints**. No temporary restraining order, preliminary or permanent injunction or other Order preventing, restraining, enjoining or otherwise prohibiting the consummation of the Offer or the Merger shall have been issued by any Governmental Body of competent jurisdiction and remain in effect, and there shall not be any applicable Law enacted or deemed applicable to the Merger that makes consummation of the Offer or the Merger illegal.
- 7.2 **Consummation of Offer**. Purchaser shall, in accordance with <u>Section 1.1(b)</u>, have irrevocably accepted for payment all of the Shares validly tendered pursuant to the Offer and not properly withdrawn.

ARTICLE VIII

TERMINATION

- 8.1 **Termination**. This Agreement may be terminated and the transactions contemplated herein may be abandoned at any time prior to the Offer Acceptance Time, as follows:
 - (a) by mutual written consent of Parent and the Company;
- (b) by either Parent or the Company by written notice to the other if (i) a court of competent jurisdiction or other Governmental Body shall have issued a final and nonappealable Order permanently restraining, enjoining or otherwise prohibiting the making or consummation of the Offer, the Merger or the other transactions contemplated by this Agreement or (ii) there shall be enacted after the date of this Agreement any applicable Law in any jurisdiction in which any Acquired Entity or Parent has a permanent establishment making the commencement or consummation of the Offer, the Merger or the other transactions contemplated by this Agreement illegal;

(c) by Parent by written notice to the Company if prior to the Offer Acceptance Time the Company Board shall have effected an Adverse Change Recommendation;

(d) by either the Company or Parent by written notice if the Offer Acceptance Time shall not have occurred by the End Date; provided, that if as of the End Date the Regulatory Condition (as defined in Annex I) has not been satisfied, the End Date may be extended on one occasion by Parent or the Company for a period of 90 days by written notice to the Company or Parent, as applicable, and such date, as so extended, shall be the End Date; provided, however, that the right to terminate this Agreement pursuant to this Section 8.1(d) shall not be available to any party whose breach of any covenant or agreement set forth in this Agreement has been the principal cause of, or resulted in, the failure of the Offer Acceptance Time to have occurred on or before the End Date;

(e) by the Company by written notice to Parent at any time prior to the Offer Acceptance Time in order to accept a Superior Proposal and enter into a Company Acquisition Agreement relating to such Superior Proposal, if (i) such Superior Proposal shall not have resulted from any breach of Section 5.4, (ii) the Company has complied in all respects (other than non-compliance that is immaterial in significance and inadvertent) with its obligations under Section 5.4 with respect to such Superior Proposal and the Company Board, after satisfying all of the applicable requirements set forth in Section 5.4 with respect to the Superior Proposal, shall have authorized the Company to enter into a Company Acquisition Agreement with respect to such Superior Proposal and (iii) the Company enters into such Company Acquisition Agreement concurrently with the termination of this Agreement;

(f) by Parent by written notice to the Company at any time prior to the Offer Acceptance Time if a breach of any representation or warranty or failure to perform any covenant or obligation contained in this Agreement on the part of the Company shall have occurred that would cause a failure of the conditions in clauses (2)(c) or (d) of Annex I to occur; provided, however, that, for purposes of this Section 8.1(f), if such a breach is curable by the Company within the earlier of the End Date and twenty (20) business days of the date Parent gives the Company written notice of such breach referencing this Section 8.1(f) and the Company is continuing to use its reasonable best efforts to cure such breach, then Parent may not terminate this Agreement under this Section 8.1(f) on account of such breach unless such breach shall remain uncured upon the earlier of the End Date and the expiration of such twenty (20) business day period; provided further, however, that Parent shall not be entitled to terminate this Agreement pursuant to this Section 8.1(f) if either Parent or Purchaser is in breach of its obligations under this Agreement such that the Company would be entitled to terminate this Agreement pursuant to Section 8.1(g);

(g) by the Company by written notice to Parent at any time prior to the Offer Acceptance Time, if a breach in any respect of any representation or warranty or failure to perform in any respect any covenant or obligation contained in this Agreement on the part of Parent shall have occurred, in each case if such breach or failure has prevented or could reasonably be expected to prevent Parent or Purchaser from consummating the Offer, the Merger or any other transaction contemplated herein; provided, however, that, for purposes of this Section 8.1(g), if such a breach is curable by Parent within the earlier of the End Date and twenty (20) business days of the date the Company gives Parent written notice of such breach referencing this Section 8.1(g) and Parent is continuing to use its reasonable best efforts to cure such breach, then the Company may not terminate this Agreement under this Section 8.1(g) on account of such breach unless such breach shall remain uncured upon the earlier of the End Date and the expiration of such twenty (20) business day period; provided further, however, that the Company shall not be entitled to terminate this Agreement pursuant to this Section 8.1(g) if the Company is in breach of its obligations under this Agreement such that Parent would be entitled to terminate this Agreement pursuant to Section 8.1(f); or

(h) by either Parent or the Company, if the Offer shall have expired (after giving effect to any extensions thereof in accordance with this Agreement) or been terminated, in each case, in accordance with the terms of this Agreement and the Offer Acceptance Time shall not have occurred solely as a result of the Minimum Condition not being satisfied; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this <u>Section 8.1(h)</u> shall not be available to a party that has breached its (or, in the case of Parent, Purchaser breached its) obligations in any material respect under this Agreement in any manner that shall have been the primary cause of the Minimum Condition not being satisfied.

8.2 **Effect of Termination**. If this Agreement is terminated pursuant to <u>Section 8.1</u>, this Agreement shall be of no further force or effect without liability of any party (or any Representative or securityholders of such party) to each other party hereto; <u>provided</u>, <u>however</u>, that the provisions of <u>Section 1.1(g)</u>, the last sentence of <u>Section 1.2(c)</u>, <u>Section 5.2</u>, this <u>Section 8.2</u>, <u>Section 8.3</u> and <u>Article IX</u> shall survive any termination hereof pursuant to <u>Section 8.1</u>. Notwithstanding the foregoing or any other provision of this Agreement to the contrary, none of Parent, Purchaser or the Company shall be relieved or released from any liabilities or damages arising out of any Willful Breach of any provision of this Agreement or any other agreement delivered in connection herewith or any intentional fraud. The Confidentiality Agreement shall survive the termination of this Agreement and shall remain in full force and effect in accordance with its terms.

8.3 Expenses; Termination Fees.

(a) Except as set forth in Section 6.9 and this Section 8.3, all fees and expenses incurred in connection with this Agreement and the Offer, the Merger and the other transactions contemplated herein shall be paid by the party incurring such expenses, whether or not the Offer and Merger are consummated. Parent and the Company shall each pay fifty percent (50%) of (i) the filing fees, mailing expenses and printing expenses associated with the Offer and (ii) the filing fees payable pursuant to the HSR Act and any other applicable Antitrust Laws in connection with the Offer, the Merger and the other transactions contemplated by this Agreement.

- (b) In the event that:
 - (i) this Agreement is terminated pursuant to **Section 8.1(c)**;
 - (ii) this Agreement is terminated pursuant to **Section 8.1(e)**; or
- (iii) (A) this Agreement is terminated by Parent or the Company pursuant to <u>Section 8.1(d)</u> or <u>Section 8.1(h)</u> or by Parent pursuant to <u>Section 8.1(f)</u>, (B) after the date of this Agreement and at or prior to the time of the termination of this Agreement an Acquisition Proposal shall have been made or commenced, or an intention to make or commence any Acquisition Proposal shall have been publicly proposed or announced, and (C) the Company consummates any Acquisition Proposal within twelve (12) months after such termination or the Company enters into a definitive agreement within twelve (12) months after such termination to effect any Acquisition Proposal that is subsequently consummated (<u>provided</u> that, for purposes of this <u>Section 8.3(b)(iii)</u>, all percentages in the definition of Acquisition Proposal shall be replaced with 50%);

then the Company shall pay Parent a fee in an amount equal to six million one hundred eighteen thousand eight hundred ten dollars (\$6,118,810) (the "Termination Fee") by wire transfer of same-day funds (x) in the case of Section 8.3(b)(i), within two (2) business days after such termination, (y) in the case of Section 8.3(b)(ii), prior to or concurrently with the termination of this Agreement pursuant to Section 8.1(e), and (z) in the case of Section 8.3(b)(iii), concurrently with the consummation of the transactions contemplated by such Acquisition Proposal. For the avoidance of doubt, any payment made by the Company under this Section 8.3(b) shall be payable only once with respect to this Section 8.3(b) and not in duplication even though such payment may be payable under one or more provisions hereof.

(c) The Company and Parent acknowledge and agree that the agreements contained in this Section 8.3 are an integral part of the transactions contemplated by this Agreement, and that, without these agreements, the Company and Parent would not enter into this Agreement. Subject to Section 8.2 and Section 9.6, this Section 8.3 shall be the sole and exclusive remedy of the Company, Parent and Purchaser in the event of termination of this Agreement. In the event that the Company shall fail to pay the Termination Fee when due, the term "Termination Fee" shall be deemed to include the reasonable costs and expenses actually incurred or accrued by the other party (including reasonable fees and expenses of counsel) in connection with the collection under and enforcement of this Section 8.3, together with interest on such unpaid Termination Fee, commencing on the date that the Termination Fee became due, at a rate equal to the "prime rate" as published in The Wall Street Journal, Eastern Edition, in effect on the date such payment was required to be made through the date of payment (calculated daily on the basis of a year of 365 days and the actual number of days elapsed, without compounding).

ARTICLE IX

MISCELLANEOUS PROVISIONS

9.1 **Notices**. Any notices or other communications required or permitted under, or otherwise given in connection with, this Agreement shall be in writing and shall be deemed to have been duly given (i) when delivered or sent if delivered in person or sent by facsimile transmission (<u>provided</u> confirmation of facsimile transmission is obtained), (ii) on the fifth (5th) business day after dispatch by registered or certified mail, (iii) on the next business day if transmitted by national overnight courier or (iv) on the date delivered if sent by e-mail (<u>provided</u> confirmation of email receipt is obtained), in each case as follows:

if to Parent or Purchaser:

Altair Engineering Inc. 1820 E. Big Beaver Road Troy, Michigan 48083 Fax: 248-614-6197

Attention: Mr. James R. Scapa

with a copy to (which shall not constitute notice):

Steven M. Rivkin, Esq. Raoul Maitra, Esq. Altair Engineering Inc. 1820 E. Big Beaver Road Troy, Michigan 48083 Fax: 248-614-6197

and

Lowenstein Sandler LLP 1251 Avenue of the Americas New York, New York 10020

Fax: 973-597-2351

Attention: Peter H. Ehrenberg, Esq.

if to the Company:

Datawatch Corporation 4 Crosby Drive Bedford, Massachusetts 01730

Fax: 978-441-1114

Attention: Mr. Michael A. Morrison

with a copy to (which shall not constitute notice):

Choate, Hall & Stewart LLP Two International Place Boston, MA 02110 Fax: 617-248-4000

Attention: Robert V. Jahrling, Esq.

- 9.2 **Amendment**. Prior to the Offer Acceptance Time, this Agreement may be amended with the mutual agreement of the Company, Purchaser and Parent at any time. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- 9.3 **Waiver**. No failure on the part of any party to exercise any power, right, privilege or remedy under this Agreement, and no delay on the part of any party in exercising any power, right, privilege or remedy under this Agreement, shall operate as a waiver of such power, right, privilege or remedy; and no single or partial exercise of any such power, right, privilege or remedy shall preclude any other or further exercise thereof or of any other power, right, privilege or remedy. No party shall be deemed to have waived any claim arising out of this Agreement, or any power, right, privilege or remedy under this Agreement, unless the waiver of such claim, power, right, privilege or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such party; and any such waiver shall not be applicable or have any effect except in the specific instance in which it is given.
- 9.4 **No Survival of Representations and Warranties**. None of the representations and warranties contained in this Agreement or in any certificate or schedule or other document delivered pursuant to this Agreement shall survive the Effective Time.

9.5 Entire Agreement; No Reliance; Counterparts.

- (a) This Agreement, the Confidentiality Agreement (to the extent consistent with the terms of this Agreement), the Support Agreements and the Company Disclosure Letter constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof.
- (b) Each party hereto agrees that, except for the representations and warranties contained in <u>Article III</u> (including the Company Disclosure Letter), and <u>Article IV</u> of this Agreement, neither the Company, Parent or Purchaser makes any other representations or warranties and each hereby disclaims any other representations or warranties made by itself or any of its Representatives, with respect to the execution and delivery of this Agreement or the transactions contemplated hereby, notwithstanding the delivery or disclosure to any other party or any other party's Representatives of any document or other information with respect to any one or more of the foregoing. Without limiting the generality of the foregoing, each of Parent and Purchaser agrees that none of the Acquired Entities makes or has made any representation or warranty with respect to (i) any projections, forecasts, estimates, plans or budgets or future revenues, expenses or expenditures, future results of operations (or any component thereof),

future cash flows (or any component thereof) or future financial condition (or any component thereof) of the Company or any Company Subsidiary or the future business, operations or affairs of the Company or any Company Subsidiary heretofore or hereafter delivered to or made available to it, or (ii) any other information, statements or documents heretofore or hereafter delivered to or made available to it, including the information in the electronic data room of the Company, with respect to the Company or any Company Subsidiary or the business, operations or affairs of the Company or any Company Subsidiary, in each case except to the extent expressly covered by a representation and warranty made in this Agreement.

(c) This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by facsimile or PDF shall be sufficient to bind the parties to the terms and conditions of this Agreement.

9.6 Applicable Law; Jurisdiction; Specific Performance; Remedies; Waiver of Jury Trial.

(a) This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without regard to the conflicts of law rules of such State or of any other jurisdiction that would cause the application of the laws of any jurisdiction other than the State of Delaware. The parties hereto agree that any Legal Proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby shall be brought in the Delaware Court of Chancery or, if that court does not have jurisdiction, a federal court sitting in the State of Delaware (the "Delaware Courts"). Each party hereby irrevocably submits to the exclusive jurisdiction of such court in respect of any legal or equitable action, suit or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby, or relating to enforcement of any of the terms of this Agreement, and hereby waives, and agrees not to assert, as a defense in any such action, suit or proceeding, any claim that it is not subject personally to the jurisdiction of such court, that the action, suit or proceeding is brought in an inconvenient forum, that the venue of the action, suit or proceeding is improper or that this Agreement or the transactions contemplated hereby may not be enforced in or by such courts. Each party agrees that notice or the service of process in any action, suit or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby shall be properly served or delivered if delivered in the manner contemplated by Section 9.1 or in any other manner permitted by Law.

(b) The parties agree that irreparable damage would occur and that the parties would not have any adequate remedy at law in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that each of the parties shall be entitled to an injunction or injunctions to prevent breaches or threatened breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in the Delaware Courts and, in any action for specific performance, each party waives the defense of adequacy of a remedy at law and waives any requirement for the securing or posting of any bond in connection with such remedy or any requirement to prove damages, this being in addition to any other remedy to which they are entitled at law or in equity (subject to the limitations set forth in this Agreement). The parties hereto further agree that (i) by seeking the remedies provided for in this Section 9.6(b), a party

shall not in any respect waive its right to seek any other form of relief that may be available to a party under this Agreement (including monetary damages) for breach of any of the provisions of this Agreement or in the event that this Agreement has been terminated or in the event that the remedies provided for in this **Section 9.6(b)** are not available or otherwise are not granted, and (ii) nothing set forth in this **Section 9.6(b)** shall require any party hereto to institute any proceeding for (or limit any party's right to institute any proceeding for) specific performance under this **Section 9.6(b)** prior or as a condition to exercising any termination right under **Article VIII** (and pursuing damages after such termination), nor shall the commencement of any Legal Proceeding pursuant to this **Section 9.6(b)** or anything set forth in this **Section 9.6(b)** restrict or limit any party's right to terminate this Agreement in accordance with the terms of **Article VIII** or pursue any other remedies under this Agreement that may be available at any time.

(c) EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT, ANY OTHER AGREEMENTS OR DOCUMENTS EXECUTED OR PREPARED IN CONNECTION HEREWITH OR THE TRANSACTIONS CONTEMPLATED HEREBY.

- 9.7 **Third Party Beneficiaries**. This Agreement is for the sole benefit of the parties hereto and their permitted assigns and nothing herein, express or implied, is intended to or shall confer upon any other Person or entity any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement, except from and after the Effective Time, as specifically provided in <u>Section 6.3</u> hereof (which shall be to the benefit of the Persons referred to in such section).
- 9.8 **Assignability**. Neither this Agreement nor any of the rights, interests or obligations under this Agreement shall be assigned, in whole or in part, by operation of law or otherwise by any of the parties without the prior written consent of the other parties, except that either Parent or Purchaser may assign, in its sole discretion, any of or all its rights, interests and obligations under this Agreement to Parent (in the case of Purchaser) or to any direct or indirect Subsidiary of Parent, but no such assignment shall relieve Parent or Purchaser, as applicable, of any of its obligations under this Agreement. Any purported assignment without such consent shall be null and void. Subject to the preceding sentences, this Agreement will be binding upon, inure to the benefit of, and be enforceable by, the parties and their respective successors and assigns.
- 9.9 **Cooperation**. The Company agrees to reasonably cooperate with Parent and to execute and deliver such further documents, certificates, agreements and instruments and to take such other actions as may be reasonably requested by Parent to evidence or reflect the transactions contemplated by this Agreement and to carry out the intent and purposes of this Agreement, in each case to the extent not inconsistent with any other provision of this Agreement.

9.10 **Severability**. Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective only to the extent of such prohibition or unenforceability without invalidating the remaining provisions of this Agreement or affecting the validity or enforceability of such provision in any other jurisdiction. The application of such invalid or unenforceable provision to Persons or circumstances other than those as to which it is held invalid or unenforceable shall be valid and be enforced to the fullest extent permitted by applicable Law. To the extent any provision of this Agreement is determined to be prohibited or unenforceable in any jurisdiction, or determined to be impermissible by any Governmental Body, each of the parties agrees to use reasonable commercial efforts to substitute one or more valid, legal and enforceable provisions that, insofar as practicable, implement the purposes and intent of the prohibited, unenforceable, or impermissible provision.

9.11 **Obligation of Parent**. Parent shall cause Purchaser to comply in all respects with each of the representations, warranties, covenants, obligations, agreements and undertakings made or required to be performed by Purchaser in accordance with the terms of this Agreement, the Offer, the Merger, and the other transactions contemplated hereby. As a material inducement to the Company's willingness to enter into this Agreement and perform its obligations hereunder, Parent hereby unconditionally guarantees full performance and payment by Purchaser of each of the covenants, obligations and undertakings required to be performed by Purchaser under this Agreement and the transactions contemplated hereby at or prior to the Effective Time, subject to all terms, conditions and limitations contained in this Agreement, and hereby represents, acknowledges and agrees that any such breach of any such representation and warranty or default in the performance of any such covenant, obligation, agreement or undertaking of Purchaser at or prior to the Effective Time shall also be deemed to be a breach or default of Parent, and the Company shall have the right, exercisable in its sole discretion, to pursue any and all available remedies it may have arising out of any such breach or nonperformance directly against either or both of Parent and Purchaser in the first instance.

9.12 Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement or any other agreements or documents executed or prepared in connection herewith.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits," "Annexes" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits, Annexes or Schedules to this Agreement.

- (e) The bold-faced headings and the table of contents contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.
 - (f) The phrase "date hereof" or "date of this Agreement" shall be deemed to refer to November 5, 2018.
- (g) Unless the context requires otherwise (i) any definition of or reference to any Contract, instrument or other document or any Law herein shall be construed as referring to such Contract, instrument or other document or Law as from time to time amended, supplemented or otherwise modified, (ii) any reference herein to any Person shall be construed to include such Person's successors and assigns and (iii) the words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this Agreement in its entirety and not to any particular provision hereof.
- 9.13 **Non-Parties**. Notwithstanding anything to the contrary in this Agreement, it is hereby agreed and acknowledged that this Agreement may only be enforced against, and any claims of action that may be based upon, arise out of, or relate to, this Agreement, or the negotiation, execution or performance of this Agreement, may only be made against the parties hereto, and no former, current or future Affiliates, officers, directors, managers, employees, equityholders, lenders, managers, agents or Representatives of any party hereto, in each case, who is not a party to this Agreement, shall have any liability for any obligations of the parties hereto or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby (whether at law or in equity, in contract, tort or otherwise).

[Signature page follows]

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed as of the date first above written.

DATAWATCH CORPORATION

By: /s/ Michael A. Morrison

Name: Michael A. Morrison Title: Chief Executive Officer

[Signature Page to Agreement and Plan of Merger]

ALTAIR ENGINEERING INC.

By: /s/ James R. Scapa

Name: James R. Scapa Title: Chief Executive Officer

DALLAS MERGER SUB, INC.

By: /s/ James R. Scapa

Name: James R. Scapa Title: President

[Signature Page to Agreement and Plan of Merger]

Exhibit A

Certain Definitions

For purposes of the Agreement (including this Exhibit A and Annex I):

"Acquired Entities" shall mean the Company and each of its Subsidiaries, collectively.

"Acquisition Proposal" shall mean any inquiry (in writing or otherwise), offer, proposal or indication of interest from any Third Party relating to any transaction or series of related transactions involving (a) any acquisition or purchase by any Third Party, directly or indirectly, of (i) assets (including capital stock of Subsidiaries of the Company) representing (x) 15% or more of the consolidated net revenues of the Company and its Subsidiaries immediately prior to such transaction or series of transactions or (y) 15% or more of the fair market value of the assets of the Company and its Subsidiaries, taken as a whole, or (ii) 15% or more of any class of outstanding voting or equity securities of the Company, (b) any tender offer (including a self-tender) or exchange offer that, if consummated, would result, directly or indirectly, in any Third Party (or the shareholders thereof) beneficially owning 15% or more of any class of outstanding voting or equity securities of the Company or the surviving entity or (c) any merger, amalgamation, consolidation, share exchange, business combination, joint venture or other similar transaction involving the Company or any of its Subsidiaries, or liquidation, dissolution, recapitalization, extraordinary dividend, other significant corporate reorganization or other similar transaction involving the Company or any of its Subsidiaries, (i) pursuant to which 15% or more of any class of outstanding voting or equity securities of the Company or the resulting entity would be beneficially owned by any Third Party (or the direct or indirect parent entity of such Third Party or the shareholders of such Third Party) or (ii) which would result in assets (including capital stock of Subsidiaries of the Company) representing (x) 15% or more of the fair market value of the assets of the Company and its Subsidiaries immediately prior to such transaction, or series of transactions or (y) 15% or more of the fair market value of the assets of the Company and its Subsidiaries, taken as a whol

"Adverse Change Recommendation" is defined in Section 5.4(b) of the Agreement.

"Affiliate" shall mean, as to any Person, any other Person that, directly or indirectly, controls, or is controlled by, or is under common control with, such Person. For this purpose, "control" (including, with its correlative meanings, "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of management or policies of a Person, whether through the ownership of securities or partnership or other ownership interests, by contract or otherwise.

"Agreement" shall mean the agreement and plan of merger to which this Exhibit A is attached, as such agreement and plan of merger may be amended from time to time in accordance with the terms thereof.

"Angoss Acquisition" means the transactions pursuant to that certain Share Purchase Agreement, dated as of January 30, 2018, by and among the Company, 2617421 Ontario Inc. and the shareholders of Angoss Software Corporation.

- "Antitrust Laws" shall mean the Sherman Act, as amended, the Clayton Act, as amended, the HSR Act, the Federal Trade Commission Act, as amended, all applicable foreign anti-trust laws and all other applicable Laws and Orders issued by a Governmental Body that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening of competition through merger or acquisition.
 - "Authorizations" is defined in Section 3.12 of the Agreement.
 - "Balance Sheet" is defined in Section 3.6 of the Agreement.
 - "Book Entry Share" is defined in Section 2.6(a)(i) of the Agreement.
- "business day" means any day except a Saturday, Sunday or other day on which the SEC or banks in the City of New York are authorized or required by Law to be closed in New York City.
 - "Certificate" is defined in Section 2.6(a)(i) of the Agreement.
 - "Change of Status Notice" is defined in Section 5.4(b) of the Agreement.
 - "Closing" is defined in Section 2.3 of the Agreement.
 - "Closing Date" is defined in Section 2.3 of the Agreement.
 - "Code" shall mean the Internal Revenue Code of 1986, as amended.
 - "Company" is defined in the preamble to the Agreement.
- "Company Acquisition Agreement" means any binding letter of intent, memorandum of understanding, acquisition agreement, merger agreement or similar binding agreement providing for the consummation of a transaction contemplated by any Acquisition Proposal (other than a confidentiality agreement referred to in Section 5.4(a)).
 - "Company Affiliate Transaction" is defined in Section 3.18 of the Agreement.
- "Company Associate" shall mean any current or former director, officer, employee, independent contractor or other service provider of the Company or any Company Subsidiary.
 - "Company Board" is defined in Section 3.20(b) of the Agreement.
 - "Company Board Recommendation" is defined in Section 1.2(a) of the Agreement.
- "Company Charter Documents" shall mean the Company's amended and restated certificate of incorporation and amended and restated bylaws, as in effect on the date of execution of the Agreement, access to which has been provided to Purchaser.
 - "Company Common Stock" shall mean the common stock, \$0.01 par value per share, of the Company.

- "Company Contract" shall mean any Contract to which any of the Acquired Entities is a party.
- "Company Data" means any and all data and information (including Personal Information) processed by or on behalf of any of the Acquired Entities in connection with the operation of the business of the Company and its Subsidiaries.
- "Company Disclosure Letter" shall mean the disclosure letter that has been prepared by the Company in accordance with the requirements of the Agreement and that has been delivered by the Company to Parent immediately prior to or concurrently with the execution of the Agreement.
- "Company Employee Agreement" shall mean each management, employment, severance, termination pay, stay-bonus, retention, transaction bonus, change in control, consulting, relocation, repatriation or expatriation agreement or other Contract between: (a) any of the Acquired Entities and (b) any Company Associate.
 - "Company Equity Plan" shall mean the Company's Third Amended and Restated 2011 Equity Compensation and Incentive Plan, as amended.
 - "Company Financial Advisor" is defined in Section 3.24 of the Agreement.
- "Company Intellectual Property" shall mean any Intellectual Property owned by the Company or its Subsidiaries as of the date hereof that is material to the business of the Company and its Subsidiaries, taken as a whole, as conducted as of the date of this Agreement.
 - "Company Options" shall mean all options to purchase shares of Company Common Stock issued pursuant to the Company Equity Plan.
 - "Company Preferred Stock" shall mean the preferred stock, \$0.01 par value per share, of the Company.
 - "Company RSUs" shall mean any restricted stock units of the Company issued pursuant to the Company Equity Plan.
 - "Company SEC Documents" is defined in Section 3.4(a) of the Agreement.
 - "Compensation Committee" is defined in Section 6.10 of the Agreement.
 - "Company Systems" is defined in Section 3.8(i) of the Agreement
- "Confidentiality Agreement" shall mean that certain Mutual Non-Disclosure Agreement, dated as of May 14, 2018, by and between the Company and Parent.
 - "Consent" shall mean any consent, approval, order or authorization, or registration, declaration or filing.
 - "Continuing Employees" is defined in Section 6.2(a) of the Agreement.

"Contract" shall mean any legally binding written or oral contract, agreement, note, bond, indenture, mortgage, guarantee, option, lease (or sublease), license (or sublicense), sales or purchase order, warranty, commitment, or other instrument, obligation, arrangement or understanding of any kind, including all amendments, supplements or modifications thereto.

"Costs" is defined in Section 6.3(a) of the Agreement.

"Covered Securityholders" is defined in Section 3.27 of the Agreement.

"Delaware Courts" is defined in Section 9.6(a) of the Agreement.

"Depository Agent" is defined in Section 2.7(a) of the Agreement.

"DGCL" shall mean the Delaware General Corporation Law, as amended.

"Dissenting Shares" is defined in Section 2.8 of the Agreement.

"Dissenting Stockholders" is defined in Section 2.6(a) of the Agreement.

"DOJ" shall mean the U.S. Department of Justice.

"D&O Insurance" is defined in Section 6.3(c) of the Agreement.

"Effective Time" is defined in Section 2.3 of the Agreement.

"Employee Plan" shall mean any "employee benefit plan" within the meaning of Section 3(3) of ERISA, whether or not subject to ERISA, and any change in control, bonus, deferred compensation, incentive compensation, stock purchase, stock option, stock-based, severance pay, termination pay, death and disability benefits, employment, consulting, hospitalization, medical, life or other insurance, flexible benefits, supplemental unemployment benefits, vacation or paid-time off, profit-sharing, pension or retirement plan, fringe benefit, plan, policy, program, agreement or arrangement and each other employee benefit plan, program, policy or arrangement that is or has been (a) sponsored, maintained, contributed to or required to be contributed to by any of the Acquired Entities for the benefit of any Company Associate or (b) with respect to which any Acquired Entity or any ERISA Affiliate has or may have any current or contingent liability.

"Employment Compensation Arrangement" is defined in Section 3.27 of the Agreement.

"Encumbrance" shall mean any lien, pledge, hypothecation, charge, deed of trust, mortgage, security interest, encumbrance, defect in title, lease, sublease, option, right of first refusal, preemptive right, community property interest or restriction of any nature (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, any restriction on the use of any asset and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

"End Date" shall mean February 28, 2019.

"Entity" shall mean any corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any company limited by shares, limited liability company or joint stock company), firm, society or other enterprise, association, organization or entity.

"Environmental Law" shall mean any applicable Law or any agreement with any Governmental Body or other Person, relating to human health and safety, the environment or any Hazardous Substance.

"Environmental Permits" shall mean, with respect to any Person, any permit, certificate, registration, notice, approval, identification number, license, or other authorization relating to or required by Environmental Law and affecting, or relating in any way to, the business of such Person or any of its Subsidiaries.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" shall mean any person or entity, whether or not incorporated, that would be treated together with any Acquired Entity as a single employer within the meaning of Section 414 of the Code.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Excluded Shares" is defined in Section 2.6(a)(i) of the Agreement.

"Expiration Date" is defined in Section 1.1(d) of the Agreement.

"Extension Deadline" is defined in Section 1.1(e) of the Agreement.

"Fairness Opinion" is defined in Section 3.24 of the Agreement.

"FTC" shall mean the U.S. Federal Trade Commission.

"GAAP" is defined in Section 3.4(b) of the Agreement.

"Governmental Antitrust Authority" is defined in Section 6.1(c) of the Agreement.

"Governmental Body" shall mean (a) any government or any state, department, local authority or other political subdivision thereof, or (b) any governmental or quasi-governmental body, agency, authority (including any central bank, taxing authority or transgovernmental or supranational entity or authority), minister or instrumentality (including any court or tribunal) exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

"Hazardous Substance" shall mean any pollutant, contaminant, waste or chemical or any toxic, radioactive, ignitable, corrosive, reactive or otherwise hazardous substance, waste or material, or any substance, waste or material having any constituent elements displaying any of the foregoing characteristics, including any substance, waste or material regulated under any Environmental Law.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indebtedness" shall mean, with respect to any Person, all obligations (including all obligations in respect of principal, accrued interest, penalties, fees and premiums) of such Person: (a) for borrowed money (including obligations in respect of drawings under overdraft facilities), (b) evidenced by notes, bonds, debentures, loan agreements or similar contracts or agreements, (c) for the deferred purchase price of property, goods or services or other conditional sale or title retention agreements (other than trade payables or accruals incurred in the Ordinary Course of Business which are not overdue), including all seller notes and "earn-out" payments, (d) under capital leases (in accordance with GAAP), (e) the maximum amount (after giving effect to any prior drawings and reductions which have been reimbursed) in respect of outstanding letters of credit, bank guaranties, surety bonds, performance bonds and bankers' acceptances and similar instruments issued or created by or for the account of such Person or pursuant to securitization, factoring programs or arrangements or similar arrangements and all obligations of such Person under or in respect of a synthetic lease, tax retention operating lease, off-balance sheet loan or other off-balance sheet financing product, (f) for contracts or agreements relating to interest rate or currency rate protection, swap agreements, collar agreements or any other hedging agreements, (g) guarantees with respect to any indebtedness of any other Person of a type described in clauses (a) through (e) above and (h) for all Indebtedness referred to in clauses (a) through (g) above, all accrued interest thereon and all Indebtedness secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Encumbrances on property (including accounts and contract rights) owned by such Person, even though such Person has not assumed or become liable for the payment of such Indebtedness.

"Indemnified Parties" is defined in <u>Section 6.3(a)</u> of the Agreement.

"Initial Expiration Date" is defined in Section 1.1(d) of the Agreement.

"Intellectual Property" shall mean (a) patents, trademarks, service marks, trade names, domain names, designs and trade secrets, (b) applications for, registrations of, and related continuations, divisionals, reissues, re-examinations, and extensions of such patents, trademarks, service marks, trade names, domain names, copyrights and designs, (c) processes, formulae, methods, schematics, technology, know-how, computer software programs (including Source Code, object, executable, or binary code, and all documentation related thereto), and applications and (d) other tangible proprietary or confidential information and materials.

"Intervening Event" shall mean any material fact, event, change, development or circumstance that is not known to or reasonably foreseeable by the Company Board as of the date hereof, not relating to any Acquisition Proposal and becomes known by the Company Board

between the date hereof and the Offer Acceptance Time; <u>provided</u> that in no event shall any of the following constitute or be deemed to be an Intervening Event: (i) the receipt of, existence of or terms of an Acquisition Proposal, or any inquiry, indication of interest, proposal or offer that could reasonably be expected to lead to an Acquisition Proposal, or any matter relating thereto or consequence thereof or (ii) changes in the market price or trading volume of the Company Common Stock (it being understood that the underlying causes of such changes may be taken into account in determining whether there has been or there exists an Intervening Event unless such underlying cause would otherwise be excepted by this definition).

"Intervening Event Notice Period" is defined in Section 5.4(b)(iii)(2).

"IRS" shall mean the Internal Revenue Service.

"knowledge" shall mean knowledge, after reasonable inquiry to the direct reports of such Person, of each of the executive officers of the Person.

"Law" shall mean, with respect to any Person, any international, national, federal, state or local law (statutory, common or otherwise), constitution, treaty, convention, ordinance, code, rule, regulation or other similar requirement enacted, adopted, promulgated or applied by a Governmental Body that is binding upon or applicable to such Person, as amended unless expressly specified otherwise.

"Legal Proceeding" shall mean any action, suit, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation commenced, brought, conducted or heard by or before, or otherwise involving, any court or other Governmental Body or any arbitrator or arbitration panel.

"Material Adverse Effect" shall mean any change, event, occurrence or effect, individually or in the aggregate, that has or would reasonably be expected to have a material adverse effect on (a) the assets, liabilities, business, financial condition or results of operations of the Company and its Subsidiaries, taken as a whole, or (b) the ability of the Company to consummate the Offer, the Merger and the other transactions contemplated by this Agreement, excluding, with respect to clause (a) only, any such adverse effect resulting from or arising out of: (i) the announcement of the Offer or the Merger; (ii) the identity of Parent or any of its Affiliates as the acquirer of the Company; (iii) general economic, financial market or political conditions; (iv) general conditions in the industry in which the Company and its Subsidiaries operate; (v) any changes (after the date hereof) in GAAP or applicable Law (or interpretations thereof); (vi) any geopolitical conditions, the outbreak or escalation of hostilities, any acts of war, sabotage, terrorism or military actions, or any escalation or worsening of any such hostilities, acts of war, sabotage, terrorism or military actions threatened or underway as of the date of this Agreement; (vii) any failure by the Company to meet internal or analysts' estimates or projections, performance measures, operating statistics or revenue or earnings predictions for any period or a decline in the price or trading volume of Shares (provided, however, that, except as otherwise provided in this definition, the underlying causes of such failure or decline may be taken into account in determining whether a Material Adverse Effect has occurred); or (viii) the taking of any specific action, or refraining from taking any specific action, in each case at the

written request or with the written approval of Parent following the date hereof and not otherwise expressly required in this Agreement; provided, further, in the case of clauses (iii), (iv), (v) and (vi) such effect may be taken into account in determining whether or not there has been a Material Adverse Effect to the extent such effect has had or would reasonably be expected to have a disproportionate adverse effect on the Company and its Subsidiaries, taken as a whole, as compared to other participants in the industry in which the Acquired Entities operate, in which case the incremental disproportionate impact or impacts may be taken into account in determining whether there has been, or would reasonably be expected to be, a Material Adverse Effect.

- "Material Contract" is defined in Section 3.9(b) of the Agreement.
- "Maximum Amount" is defined in Section 6.3(c) of the Agreement.
- "Merger" is defined in the recitals to the Agreement.
- "Merger Consideration" is defined in <u>Section 2.6(a)(i)</u> of the Agreement.
- "Minimum Condition" is defined in <u>Section 1</u> of <u>Annex I</u> to the Agreement.
- "NASDAQ" is defined in Section 1.1(e) of the Agreement.
- "Notice Period" is defined in Section 5.4(b)(ii)(4)(C).
- "OFAC" shall mean the Office of Foreign Assets Control, within the U.S. Department of Treasury.
- "Offer" is defined in the recitals to the Agreement.
- "Offer Acceptance Time" is defined in Section 1.1(b) of the Agreement.
- "Offer Conditions" is defined in Section 1.1(b) of the Agreement.
- "Offer Documents" is defined in Section 1.1(h) of the Agreement.
- "Offer Price" is defined in the recitals to the Agreement.
- "Offer to Purchase" is defined in Section 1.1(c) of the Agreement.
- "Open Source Code" shall mean all software that is distributed as "free software," "open source software," or under a similar licensing distribution model, including the GNU General Public License, GNU Lesser General Public License, Mozilla Public License, BSD license, the Artistic License, the Netscape Public License, the Sun Community Source License, the Sun Industry Standards License and the Apache License.
- "**Option Consideration**" shall mean the excess, if any, of the Offer Price over the exercise price per share of the applicable Company Option, without any interest thereon.

"**Order**" shall mean, with respect to any Person, any order, injunction, judgment, decree, ruling or other similar requirement enacted, adopted, promulgated or applied by a Governmental Body or arbitrator that is binding upon or applicable to such Person or its property.

"Ordinary Course of Business" shall mean the ordinary course of business of the Company and its Subsidiaries, in all material respects consistent with past practices (including, without limitation, if the context so requires, past practice with respect to nature, scope, quality, quantity, frequency and magnitude).

"Parent" is defined in the preamble to the Agreement.

"Paying Agent" is defined in Section 2.7(a) of the Agreement.

"Parent Employee Plan" is defined in Section 6.2(c) of the Agreement.

"Payment Fund" is defined in Section 2.7(a) of the Agreement.

"Permitted Encumbrance" shall mean (a) Encumbrances for existing Indebtedness reflected in Section 3.9(a)(ix) of the Company Disclosure Letter or on the Balance Sheet or in Note 6 to the Company's unaudited consolidated financial statements for the nine months ended June 30, 2018; (b) any Encumbrance that arises out of Taxes not in default and payable without penalty or interest or the validity of which is being contested in good faith by appropriate proceedings promptly initiated and diligently conducted, and as to which adequate reserves have been set aside in accordance with GAAP; (c) Encumbrances in favor of carriers, warehousemen, repairmen, mechanics, workmen, materialmen or similar liens or encumbrances imposed under Law in the Ordinary Course of Business consistent with past practice (i) that are for amounts not yet due and payable or that are being contested in good faith by appropriate proceedings promptly initiated and diligently conducted and for which adequate reserves have been specifically established on the Balance Sheet and (ii) that are not, individually or in the aggregate, material to the business of the Company and its Subsidiaries; (d) Encumbrances arising from transfer restrictions under securities Laws or related Laws of any jurisdiction; (e) Encumbrances discharged at or prior to the Effective Time; (f) with respect to real property, easements, rights of way, zoning ordinances, and other similar land use regulations affecting real property that do not (i) secure obligations for the payment of money or (ii) materially impair the value of such property or its use by the Company and its Subsidiaries taken as a whole in the normal conduct of the business of the Company and its Subsidiaries; or (g) non-exclusive licenses of Company Intellectual Property granted in the Ordinary Course of Business; provided, however, that except in the case of clause (a), "Permitted Encumbrances" shall not include any Encumbrances that secure the payment of money.

"Person" shall mean any individual, Entity or Governmental Body.

"Personal Information" means any information from or about an individual that alone or in combination with other information could be used to identify that individual, directly or indirectly, in particular by reference to an identifier such as a name, an identification number, location data, an online identifier, or to one or more factors specific to the physical, physiological, genetic, mental, economic, cultural or social identity of that individual, in each case, to the extent and as defined under applicable Privacy Laws.

"Pre-Closing Period" is defined in Section 5.1 of the Agreement.

"Privacy Laws" means (i) any applicable Laws relating to the Processing, privacy, security or breach of any Personal Information including, but not limited to, the Fair Credit Reporting Act, 15 U.S.C. § 1681, the General Data Protection Regulation (GDPR) (EU) 2016/679, and each European Union Member States' national implementation thereof, the Canadian Personal Information Protection and Electronic Documents Act, SC 2000, c 5 (PIPEDA), (ii) any such Laws requiring a Person to be notified of any situation where there is, or reason to believe there has been a loss, misuse, modification, unauthorized access, disclosure or acquisition of Personal Information, (iii) any successor Laws to any of the foregoing, and (iv) any applicable generally-accepted industry standards including, but not limited to, PCI-DSS related to payment processing.

"Processing" means any operation or set of operations which is performed on Personal Information or on sets of Personal Information, whether or not by automated means, such as collection, recording, organization, structuring, storage, adaptation or alteration, retrieval, consultation, use, disclosure by transmission, dissemination or otherwise making available, alignment or combination, restriction, erasure, destruction or transfer (including cross-border transfer).

"Prohibited Person" shall mean (a) any Person identified on OFAC's list of Specially Designated Nationals and Blocked Persons or targeted by an OFAC Sanctions Program; (b) the government, including any political subdivision, agency, instrumentality, or national thereof, of any country with respect to which the United States or any jurisdiction in which any Acquired Entity is operating, located or incorporated or organized administers or imposes economic or trade sanctions or embargoes; (c) any Person acting, directly or indirectly, on behalf of, or an entity that is owned or controlled by, a Specially Designated National and Blocked Person or by a government or Person identified in clause (b) above; or (d) a Person on any other similar export control, terrorism, money laundering or drug trafficking related list administered by any Governmental Body either within or outside the U.S. with whom it is illegal to conduct business pursuant to applicable Law.

"PSV Policies" is defined in Section 6.2(b) of the Agreement.

"Purchaser" is defined in the preamble to the Agreement.

"Real Property" is defined in Section 3.7(b) of the Agreement.

"Real Property Lease" is defined in Section 3.7(b) of the Agreement.

"Regulatory Condition" is defined in Annex I to this Agreement.

"Release" shall mean any actual or threatened release, spill, emission, leaking, dumping, injection, pouring, deposit, disposal, discharge, dispersal, leaching or migration into or through the environment or within any building, structure, facility or fixture.

"Representatives" shall mean with respect to any Person, the officers, directors, employees, attorneys, accountants, investment bankers, consultants, agents, professional advisors and other representatives of such Person, acting in such capacity.

"Revised Transaction Proposal" is defined in Section 5.4(b)(ii)(4)(C).

"Sarbanes-Oxley Act" is defined in Section 3.4(a) of the Agreement.

"Schedule 14D-9" is defined in Section 1.2(b) of the Agreement.

"Schedule TO" is defined in Section 1.1(h) of the Agreement.

"SEC" shall mean the United States Securities and Exchange Commission.

"Securities Act" shall mean the Securities Act of 1933, as amended.

"Shares" is defined in the recitals to the Agreement.

"Source Code" shall mean any computer instruction (whether written in a programming language, markup or script) displayed in a human readable form, and including all related annotations and software data exchange specifications, comments, documentation, instructions and procedural information.

"Stockholder List Date" is defined in Section 1.2(c) of the Agreement.

"Stockholder Litigation" shall mean any claim or Legal Proceeding (including any class action or derivative litigation) asserted or commenced by, on behalf of or in the name of, against or otherwise involving the Company, the Company Board, any committee thereof and/or any of the Company's directors or officers relating directly or indirectly to the Agreement, the Offer, the Merger or any related transaction (including any such claim or Legal Proceeding based on allegations that the Company's entry into the Agreement or the terms and conditions of the Agreement or any related transaction constituted a breach of the fiduciary duties of any member of the Company Board, any member of the Board of Directors of any of the Company's Subsidiaries or any officer of the Company or any of its Subsidiaries).

An Entity shall be deemed to be a "**Subsidiary**" of another Person if such Person directly or indirectly owns, beneficially or of record, (a) an amount of voting securities or other interests in such Entity that is sufficient to enable such Person to elect at least a majority of the members of such Entity's Board of Directors or other governing body or (b) a majority of the outstanding equity or economic interests of such Entity.

"Subsidiary Securities" is defined in Section 3.3(f) of the Agreement.

"Superior Proposal" shall mean any bona fide and written Acquisition Proposal that is binding on the Third Party making the Acquisition Proposal that the Company Board determines in good faith (after consultation with the Company's financial advisors and outside legal counsel), taking into account those matters that the Company Board in good faith deems relevant, is reasonably likely to be consummated if accepted on the terms thereof and is more

favorable from a financial perspective to the Company's stockholders than the Offer and the Merger (taking into account all adjustments to the terms of this Agreement that are offered by Parent pursuant to <u>Section 5.4(b)(ii)</u>); provided, however, that, for purposes of this definition of "Superior Proposal," references in the term "Acquisition Proposal" to "15% or more" shall be deemed to be references to "more than 80%".

"Support Agreement" and "Support Agreements" are defined in the recitals to the Agreement.

"Surviving Corporation" is defined in the recitals to the Agreement.

"Tail Period" is defined in Section 6.3(c) of the Agreement.

"Takeover Law" in Section 6.11 of the Agreement.

"Tax" shall mean (a) any and all federal, state, provincial, local, foreign and other taxes, levies, fees, imposts, duties, assessments and similar governmental charges (including any interest, fines, penalties or additions to tax imposed in connection therewith or with respect thereto, whether disputed on not) including, without limitation (i) taxes imposed on, or measured by, income, franchise, profits or gross receipts, and (ii) ad valorem, value added, capital gains, sales, goods and services, use, real or personal property, capital stock, license, branch, payroll, estimated withholding, employment, social security (or similar), unemployment, compensation, utility, severance, production, excise, stamp, occupation, premium, windfall profits, transfer, estimated and gains taxes, and customs duties, (b) any and all liability for the payment of any items described in clause (a) above as a result of being (or ceasing to be) a member of an affiliated, consolidated, combined, unitary or aggregate group (or being included (or being required to be included) in any Tax Return related to such group, and (c) any and all liability for the payment of any amounts as a result of any obligation to indemnify any other person, or any successor or transferee liability, in respect of any items described in clause (a) or (b) above.

"Tax Return" shall mean any and all reports, returns, declarations, claims for refund, elections, disclosures, estimates, information reports, returns or statements or other information or filing supplied to or required to be supplied to a Governmental Body in connection with Taxes, including any schedule or attachment thereto or amendment thereof.

"Termination Fee" is defined in Section 8.3(b) of the Agreement.

"Third Party" shall mean any Person or "group" (as defined under Section 13(d) of the Exchange Act) of Persons, other than Parent or any of its Affiliates or Representatives.

"Transfer Taxes" is defined in Section 6.9 of the Agreement.

"Treasury Regulations" shall mean the Treasury regulations promulgated under the Code.

"Willful Breach" shall mean a material breach that is caused by an action or omission to act where both of the following conditions exist: (a) the action or omission to act was itself deliberate; and (b) such deliberate action or omission to act was taken or omitted to be taken by a

party with knowledge that such act or omission to act would or would reasonably be expected to constitute a breach of this Agreement. For purposes of clarity and without limitation of what other acts or omissions may constitute a Willful Breach, subsequent to satisfaction or waiver (to the extent permitted by this Agreement) of the Offer Conditions as of the Expiration Date, it shall be a Willful Breach by Purchaser if Purchaser fails to accept for payment and to pay for all Shares validly tendered and not subsequently withdrawn pursuant to the Offer that Purchaser becomes obligated to purchase pursuant to the Offer.

EXHIBIT B

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF DATAWATCH CORPORATION

ARTICLE I

The name of the corporation is Datawatch Corporation (the "Corporation").

ARTICLE II

The registered office of the Corporation in the State of Delaware is located at 108 West 13th Street, Wilmington, New Castle County, Delaware 19801. The name of the Corporation's registered agent in the State of Delaware at such address is Business Filings Incorporated.

ARTICLE III

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (the "<u>DGCL</u>").

ARTICLE IV

The total number of shares of stock which the Corporation shall have authority to issue is one thousand (1,000) shares of Common Stock. All shares shall be Common Stock with a par value of \$0.0001 per share.

ARTICLE V

The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors, the number of members of which shall be fixed by, or in the manner provided in, the Bylaws of the Corporation. Election of directors need not be by ballot unless the Bylaws of the Corporation shall so provide.

ARTICLE VI

In furtherance and not in limitation of the power conferred upon the Board of Directors by law, the Board of Directors is expressly authorized to make, adopt, alter, amend and repeal, from time to time, the Bylaws of the Corporation, subject to the right of the stockholders entitled to vote with respect thereto to alter and repeal Bylaws made by the directors.

ARTICLE VII PERSONAL LIABILITY OF DIRECTORS.

The Corporation eliminates the personal liability of each member of the Board of Directors to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, provided that the foregoing shall not eliminate the liability of a director (i) for

any breach of such director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of Title 8 of the DGCL or (iv) for any transaction from which such director derived an improper personal benefit.

ARTICLE VIII

The Corporation shall have the right, subject to any express provisions or restrictions contained in this Certificate of Incorporation or the Bylaws of the Corporation, from time to time, to amend, alter or repeal any provision of this Certificate of Incorporation in any manner now or hereafter provided by law, and all rights and powers of any kind conferred upon a director, officer or stockholder of the Corporation by this Certificate of Incorporation or any amendment thereof are conferred subject to such right.

ARTICLE IX

- 1. Actions other than by or in the Right of the Corporation. The Corporation shall indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Corporation) by reason of the fact that he is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys' fees), judgements, fines and amounts paid in settlement actually and reasonably incurred by him in connection with such action, suit or proceeding if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation and, with respect to any criminal action or proceedings, had no reasonable cause to believe his conduct was unlawful. The termination of any action, suit or proceeding by judgement, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which he reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his conduct was unlawful.
- 2. Actions by or in the Right of the Corporation. The Corporation shall indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Corporation to procure a judgement in its favor by reason of the fact that he is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against expenses (including attorneys' fees) actually and reasonably incurred by him in connection with the defense or settlement of such action or suit if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation and except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable unless and only to the extent that the Court of Chancery of the State of Delaware or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the

circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery of the State of Delaware or such other court shall deem proper.

- 3. <u>Success on the Merits</u>. To the extent that any person described in Section 1 or 2 of this Article IX has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in said Sections, or in defense of any claim, issue or matter therein, he shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by him in connection therewith.
- 4. <u>Specific Authorization</u>. Any indemnification under Section 1 or Section 2 of this Article IX (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that indemnification of the present or former director, officer, employee or agent is proper in the circumstances because he has met the applicable standard of conduct set forth in said Sections. Such determination shall be made, with respect to a person who is a director or officer of the Corporation at the time of such determination: (1) by a majority vote of the directors who were not parties to such action, suit or proceeding, even though less than a quorum; or (2) by a committee of such directors designated by majority vote of such directors, even though less than a quorum; or (3) if there are no such directors, or if such directors so direct, by independent legal counsel in a written opinion, or (4) by the stockholders of the Corporation.
- 5. <u>Advance Payment</u>. Expenses incurred in defending a civil or criminal action, suit or proceeding shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of any person described in said Section to repay such amount if it shall ultimately be determined that he is not entitled to indemnification by the Corporation as authorized in this Article IX.
- 6. <u>Claims</u>. If a claim for indemnification under this Article IX (following the final disposition of such proceeding) is not paid in full within sixty days after the Corporation has received a claim therefor, or if a claim for any advancement of expenses under this Article IX is not paid in full within thirty days after the Corporation has received a statement or statements requesting such amounts to be advanced, the person claiming entitlement to such indemnification or advancement of expenses shall thereupon (but not before) be entitled to file suit to recover the unpaid amount of such claim. If successful in whole or in part, such person shall be entitled to be paid the expense of prosecuting such claim to the fullest extent permitted by law. In any such action, the Corporation shall have the burden of proving that such person is not entitled to the requested indemnification or advancement of expenses under applicable law.
- 7. Non-Exclusivity. The indemnification and advancement of expenses provided by, or granted pursuant to, the other Sections of this Article IX shall not be deemed exclusive of any other rights to which those provided indemnification or advancement of expenses may be entitled under any by-law, agreement, vote of stockholders or disinterested directors otherwise, both as to action in his official capacity and as to action in another capacity while holding such office.

- 8. <u>Insurance</u>. The Board of Directors may authorize, by a vote of the majority of the full board, the Corporation to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any liability asserted against him and incurred by him in any such capacity, or arising out of his status as such, whether or not the Corporation would have the power to indemnify him against such liability under the provisions of this Article IX.
- 9. <u>Continuation of Indemnification and Advancement of Expenses</u>. The indemnification and advancement of expenses provided by or granted pursuant to, this Article IX shall continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a person.
- 10. <u>Severability</u>. If any word, clause or provision of this Article IX or any award made hereunder shall for any reason be determined to be invalid, the provisions hereof shall not otherwise be affected thereby but shall remain in full force and effect.
- 11. <u>Intent of Article</u>. The intent of this Article IX is to provide for indemnification and advancement of expenses to the fullest extent permitted by Section 145 of the DGCL. To the extent that such Section or any successor section may be amended or supplemented from time to time, this Article IX shall be amended automatically and construed so as to require indemnification and advancement of expenses to the fullest extent from time to time permitted by law.

Annex I

Conditions Of The Offer

The capitalized terms used in this Annex I have the meanings set forth in the Agreement and Plan of Merger (including Exhibit A) to which this Annex I is annexed, except that the term "Agreement" shall be deemed to refer to the Agreement and Plan of Merger to which this Annex I is annexed.

- 1. Notwithstanding any other provisions of the Offer or the Agreement to the contrary, Purchaser shall not be required to, and Parent shall not be required to cause Purchaser to, accept for payment or, subject to any applicable rules and regulations of the SEC, including Rule 14e-1(c) under the Exchange Act (relating to Purchaser's obligation to pay for or return tendered Shares promptly after termination or withdrawal of the Offer), pay for any Shares validly tendered (and not validly withdrawn) pursuant to the Offer (and not theretofore accepted for payment or paid for) unless (a) there shall have been validly tendered and not validly withdrawn prior to the Expiration Date that number of Shares that, considered together with all other Shares (if any) owned by Purchaser and its "affiliates" (as such term is defined in Section 251(h)(6)(a) of the DGCL), but excluding any Shares tendered pursuant to guaranteed delivery procedures that have not yet been "received" (as such term is defined in Section 251(h)(6)(f) of the DGCL) by the "depository" (as such term is defined in Section 251(h)(6)(c) of the DGCL), represent not less than one share more than 50% of the total number of Shares outstanding at the Offer Acceptance Time (such condition, the "Minimum Condition") and (b) the waiting period (and any extension thereof) applicable to the consummation of the Offer and the Merger under the HSR Act shall have expired or been terminated (the "Regulatory Condition").
- 2. Notwithstanding any other provisions of the Offer or the Agreement to the contrary, Purchaser shall not be required to, and Parent shall not be required to cause Purchaser to, accept for payment or, subject to any applicable rules and regulations of the SEC, including Rule 14e-1(c) under the Exchange Act (relating to Purchaser's obligation to pay for or return tendered Shares promptly after termination or withdrawal of the Offer), pay for any Shares validly tendered (and not validly withdrawn) pursuant to the Offer (and not theretofore accepted for payment or paid for) if at any time on or after the date of the commencement of the Offer and prior to the Expiration Date, any of the following events shall occur and be continuing at the Expiration Date:
 - a. the Agreement shall have been terminated in accordance with its terms;
- b. any Governmental Body of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any Law that is in effect, or there shall be any temporary restraining order, preliminary or permanent injunction or other Order, in any such case preventing, restraining, enjoining, prohibiting or otherwise making illegal the consummation of the Offer, the Merger or any of the other transactions contemplated hereby;
- c. (i) the representations and warranties of the Company set forth in <u>Section 3.5(b)</u> of the Agreement, and the representations and warranties of the Company set forth elsewhere in the Agreement that are qualified by reference to Material Adverse Effect, shall

not have been true and correct as of the date of the Agreement and as of the Expiration Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall not have been true and correct as of such earlier date); (ii) the representations and warranties of the Company set forth in the Agreement that are not set forth in Section 3.5(b) of the Agreement and are not qualified by reference to Material Adverse Effect (other than the representations and warranties set forth in the first sentence of Section 3.1(a), Section 3.2, Section 3.3, Section 3.20, Section 3.21, Section 3.22, and Section 3.25 of the Agreement) shall not have been true and correct as of the date of the Agreement and as of the Expiration Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall not have been true and correct as of such earlier date) other than for any failures of such representations and warranties of the Company to be so true and correct that, individually or in the aggregate, have not resulted, and would not reasonably be expected to result, in a Material Adverse Effect (for purposes of determining the satisfaction of this condition, without regard to any "materiality" or similar qualifications or exceptions therein); (iii) any representation and warranty of the Company set forth in Section 3.3 of the Agreement shall not have been true and correct as of the date of the Agreement and as of the Expiration Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall not have been true and correct as of such earlier date), except for any *de minimis* inaccuracies (so long as the sum of all such inaccuracies does not increase the aggregate amount of the Merger Consideration by more than \$375,000); and (iv) any representation and warranty of the Company set forth in the first sentence of Section 3.1(a), Section 3.2, Section 3.20, Section 3.21, Section 3.22, and Section 3.25 of the Agreement shall not have been true and correct in any material respect as of the date of this Agreement and as of the Expiration Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall not have been true and correct as of such earlier date) (for purposes of determining the satisfaction of this condition, without regard to any "materiality" or similar qualifications and exceptions contained therein);

- d. the Company shall not have performed or complied in all material respects with the obligations, agreements and covenants required to be performed by it under the Agreement;
- e. Parent and Purchaser shall not have received a certificate of the Company signed by a duly authorized representative thereof certifying the matters set forth in clauses (c) and (d) above;
- f. since the date of this Agreement there shall have occurred any event, change, effect or development that, individually or in the aggregate, has had or would reasonably be expected to have, a Material Adverse Effect; or
- g. the SEC staff has advised, or has confirmed in response to a telephonic or written inquiry made jointly by counsel to the Company and counsel to the Purchaser and Parent, that there are outstanding or unresolved comments received from the SEC staff with respect to the Schedule TO or the related 14D-9.

The foregoing conditions shall be for the sole benefit of Parent and Purchaser and may be asserted or waived by Parent or Purchaser in whole or in part at any time and from time to time, in each case, except for the Minimum Condition and the Regulatory Condition, subject to the terms of this Agreement and applicable Law. The foregoing conditions shall be in addition to, and not in limitation of, the rights and obligations of Parent and Purchaser to extend, terminate or modify the Offer subject to, and in accordance with, the terms and conditions of this Agreement. Any Shares subject to notices of guaranteed delivery shall be deemed not to be validly tendered for purposes of satisfying the Minimum Condition unless the Shares underlying such notices of guaranteed delivery are "received" (as such term is defined in Section 251(h)(6)(f) of the DGCL) by the "depository" (as such term is defined in Section 251(h)(6)(c) of the DGCL) prior to the Expiration Date. Any reference in this Annex I or elsewhere in the Agreement to a condition or requirement being satisfied shall be deemed to be satisfied if such condition or requirement is so waived (provided, however, that the Purchaser may not waive the Minimum Condition).

TENDER AND SUPPORT AGREEMENT

This TENDER AND SUPPORT AGREEMENT (this "Agreement"), dated as of November 5, 2018, is entered into by and among Altair Engineering Inc., a Delaware corporation ("Parent"), Dallas Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("Purchaser"), and each of the Persons set forth on Schedule A hereto (each, a "Stockholder"). All capitalized terms used but not otherwise defined in this Agreement shall have the respective meanings ascribed to such terms in the Merger Agreement (as defined below). Additionally, the term "business day" shall have the meaning ascribed to such term in the Merger Agreement.

WHEREAS, as of the date hereof, each Stockholder is the record or beneficial owner (as defined in Rule 13d-3 under the Exchange Act) of the number of shares of common stock ("*Common Stock*"), par value \$0.01, of Datawatch Corporation, a Delaware corporation (the "*Company*" and together with Parent, Purchaser and each Stockholder, the "*Parties*") set forth opposite such Stockholder's name on <u>Schedule A</u> hereto (all such shares set forth on <u>Schedule A</u>, together with any shares of Common Stock of the Company that are hereafter issued to or otherwise acquired or beneficially owned by any Stockholder prior to the termination of this Agreement, but excluding any shares of Common Stock that are hereafter validly transferred under Section 4.2(d), being referred to herein as the "*Subject Shares*");

WHEREAS, concurrently with the execution and delivery of this Agreement, Parent, Purchaser and the Company are entering into an Agreement and Plan of Merger, dated as of the date hereof (as it may be amended from time to time, the "*Merger Agreement*"), which provides, among other things, for Purchaser to commence an offer to purchase all of the issued and outstanding Common Stock of the Company and the Merger of the Company and Purchaser, upon the terms and subject to the conditions set forth in the Merger Agreement;

WHEREAS, as a condition to their willingness to enter into the Merger Agreement, Parent and Purchaser have required that each Stockholder enter into this Agreement, and each Stockholder has agreed to do so in order to induce Parent and Purchaser to enter into, and in consideration of their entering into, the Merger Agreement; and

WHEREAS, the Board of Directors of the Company has adopted the Merger Agreement and approved the transactions contemplated thereby, and has approved the execution and delivery of this Agreement in connection therewith, understanding that the execution and delivery of this Agreement is a material inducement and condition to Parent's and Purchaser's willingness to enter into the Merger Agreement.

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth below and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, do hereby agree as follows:

ARTICLE 1

AGREEMENT TO TENDER

1.1 Agreement to Tender.

(a) Each Stockholder agrees to validly tender or cause to be tendered in the Offer all of such Stockholder's Subject Shares pursuant to and in accordance with the terms of the Offer, free and clear of all Encumbrances (as defined below) (other than Permitted Encumbrances (as defined below)). Without limiting the generality of the foregoing, as promptly as practicable after, but in no event later than ten business days after, the receipt of Offer documents in the Offer, each Stockholder shall deliver pursuant to the terms of the Offer all of the Subject Shares owned by the Stockholder as of the date of such tender (the "Tender Date") and all other documents or instruments required to be delivered by other Company stockholders pursuant to the terms of the Offer, including (A) a properly completed letter of transmittal with respect to such Stockholder's Subject Shares and (B) a certificate representing such Stockholder's Subject Shares or, in the case of Subject Shares that are book-entry shares or uncertificated shares, written instructions to Stockholder's broker, dealer or other nominee that such Subject Shares be tendered into the Offer, subject to the terms of this Agreement, and requesting delivery of an "agent's message" (or such other evidence, if any, of transfer as the Depository Agent may reasonably request). If any Stockholder acquires any Subject Shares after the Tender Date, such Stockholder shall tender into the Offer such Subject Shares prior to the earlier of (x) five business days following the date that the Stockholder shall acquire such Subject Shares and (y) the Expiration Date. Each Stockholder agrees that, once such Stockholder's Subject Shares are tendered, such Stockholder shall not withdraw any of such Subject Shares from the Offer, unless and until (A) the Merger Agreement is terminated prior to the purchase of the Subject Shares in the Offer, or (B) this Agreement shall have been terminated in accordance with Section 5.2. Notwithstanding anything to the contrary herein, (x) nothing herein shall require any Stockholder to exercise or tender any unexercised Company Options, or prohibit Stockholder from exercising any Company Options, and (y) no Stockholder shall have any obligation under this Agreement to tender (or cause to be tendered) any Subject Shares to the extent such tender would reasonably be expected to cause the Stockholder to incur liability under Section 16(b) of the Exchange Act. Each Stockholder agrees that (i) on and after the date hereof, such Stockholder shall not take any action, or omit to take any action, that would cause the tender of such Stockholder's Subject Shares to result in liability for such Stockholder under Section 16(b) of the Exchange Act and (ii) such Stockholder has identified on Schedule A hereto any purchases of Company Common Stock by such Stockholder during the prior six months that would be required to be matched with the tender of such Stockholder's Subject Shares for purposes of determining liability under Section 16(b) of the Exchange Act.

(b) Upon receipt of payment in full for each Stockholder's Subject Shares pursuant to the Merger Agreement, each Stockholder agrees that any and all rights incident to its ownership of Subject Shares (including any rights to recover amounts, if any, that may be determined to be due to any stockholder or former stockholder of the Company), including but not limited to rights arising out of such Stockholder's ownership of Subject Shares prior to the transfer of such Subject Shares to Purchaser or Parent pursuant to the Merger Agreement, shall be transferred to Purchaser and Parent upon the transfer to Purchaser or Parent of such Stockholder's Subject Shares.

ARTICLE 2

REPRESENTATIONS AND WARRANTIES OF THE STOCKHOLDERS

Each Stockholder represents and warrants to Parent and Purchaser as to such Stockholder on a several basis, that as of the date hereof:

- **2.1** <u>Authorization; Binding Agreement</u>. Such Stockholder has full legal capacity, right and authority to execute, deliver and perform this Agreement and to consummate the transactions contemplated hereby. This Agreement has been duly and validly authorized, executed and delivered by such Stockholder. This Agreement constitutes the legal, valid and binding obligation of such Stockholder, and assuming due authorization, execution and delivery by each of Parent and Purchaser, is enforceable against such Stockholder in accordance with its terms, subject to (a) laws of general application relating to bankruptcy, insolvency and the relief of debtors and (b) rules of law governing specific performance, injunctive relief and other equitable remedies.
- 2.2 Non-Contravention. The execution and delivery of this Agreement by such Stockholder does not, and the performance by such Stockholder of such Stockholder's obligations hereunder and the consummation by such Stockholder of the transactions contemplated hereby will not (i) except for applicable requirements under federal securities laws, the securities laws of any state or other jurisdiction, the rules of any applicable securities exchange, Takeover Laws, the pre-merger notification requirements of the HSR Act, and filings and recordation of appropriate merger documents as required by the DGCL or any other applicable Law, require any consent, approval, order, authorization, permit or other action by, or filing with or notice to, any Person (including any Governmental Body) under, constitute a default (with or without the giving of notice or the lapse of time or both) under, or give rise to any right of termination, cancellation or acceleration under, or result in the creation of any Encumbrances on any of the Subject Shares pursuant to, any contract, order or other instrument binding on such Stockholder or any applicable Law or (ii) violate any order, writ, injunction, decree, judgment, statute, rule or regulation applicable to such Stockholder, except for matters that, individually or in the aggregate, would not reasonably be expected to prevent or materially delay or impair the consummation by such Stockholder of the transactions contemplated by this Agreement or otherwise adversely impact in any material respect such Stockholder's ability to perform its obligations hereunder.
- **2.3** Ownership of Subject Shares; Total Shares. Except as set forth on Schedule A, such Stockholder is the record and beneficial owner (as defined in Rule 13d-3 under the Exchange Act) of such Stockholder's Subject Shares and has good and marketable title to such Subject Shares free and clear of any liens, claims, proxies, voting trusts or agreements, options, rights, understandings or arrangements or any other encumbrances or restrictions whatsoever on title, transfer or exercise of any rights of a stockholder in respect of such Subject Shares (collectively, "Encumbrances"), except (i) as provided hereunder or (ii) pursuant to any applicable restrictions on transfer under the Securities Act (collectively, "Permitted")

Encumbrances"). The Subject Shares listed on Schedule A opposite such Stockholder's name constitute all of the shares of Common Stock of the Company owned (both beneficially and of record) by such Stockholder as of the date hereof. Except pursuant to the Merger Agreement, no Person has any contractual or other right or obligation to purchase or otherwise acquire any of such Stockholder's Subject Shares.

- **2.4 Voting and Disposition Power**. Except as set forth on Schedule A, such Stockholder has full and sole voting power with respect to such Stockholder's Subject Shares, and full and sole power of disposition, full and sole power to issue instructions with respect to the matters set forth herein and full and sole power to agree to all of the matters set forth in this Agreement, in each case with respect to all of such Stockholder's Subject Shares. None of such Stockholder's Subject Shares are subject to any stockholders' agreement, proxy, voting trust or other agreement or arrangement with respect to the voting of such Subject Shares, except as provided hereunder.
- **2.5** <u>Reliance</u>. Such Stockholder understands and acknowledges that Parent and Purchaser are entering into the Merger Agreement in reliance upon such Stockholder's execution, delivery and performance of this Agreement.
- **2.6** <u>Absence of Litigation</u>. With respect to such Stockholder, as of the date hereof, there is no action pending against, or, to the actual knowledge of such Stockholder, threatened in writing against such Stockholder or any of such Stockholder's properties or assets (including the Subject Shares) before or by any Governmental Body that could reasonably be expected to prevent or materially delay or impair the consummation by such Stockholder of the transactions contemplated by this Agreement or otherwise materially impair such Stockholder's ability to perform its obligations hereunder; provided, however, that such Stockholder makes no representations or warranties regarding any action involving the Company or relating to the Merger Agreement.
- **2.7 Brokers**. No broker, finder, financial advisor, investment banker or other Person is entitled to any brokerage, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated hereby based upon arrangements made by or on behalf of such Stockholder.

ARTICLE 3

REPRESENTATIONS AND WARRANTIES OF PARENT AND PURCHASER

Each of Parent and Purchaser jointly and severally represents and warrants to each Stockholder that as of the date hereof:

3.1 <u>Organization; Authorization</u>. Each of Parent and Purchaser is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and has all necessary power and authority to enter into this Agreement. Parent and Purchaser have all requisite corporate power and authority, and have taken all corporate action necessary, to execute and deliver and perform their respective obligations under this Agreement; and the execution, delivery and performance by Parent and Purchaser of this Agreement have been duly authorized by all necessary action on the part of Parent and Purchaser and their respective boards of directors.

- **3.2** <u>Binding Agreement</u>. This Agreement constitutes the legal, valid and binding obligation of Parent and Purchaser, and assuming due authorization, execution and delivery by each Stockholder, is enforceable against them in accordance with its terms, subject to (a) laws of general application relating to bankruptcy, insolvency and the relief of debtors and (b) rules of law governing specific performance, injunctive relief and other equitable remedies.
- 3.3 Non-Contravention. The execution and delivery of this Agreement by each of Parent and Purchaser does not, and the performance by each of them of its obligations hereunder and the consummation by each of them of the transactions contemplated hereby will not (i) except for applicable requirements under federal securities laws, the securities laws of any state or other jurisdiction, the rules of any applicable securities exchange, Takeover Laws, the pre-merger notification requirements of the HSR Act, and filings and recordation of appropriate merger documents as required by the DGCL or any other applicable Law, require any consent, approval, order, authorization or other action by, or filing with or notice to, any Person (including any Governmental Body) under, constitute a default (with or without the giving of notice or the lapse of time or both) under, or give rise to any right of termination, cancellation or acceleration under, or result in the creation of any Encumbrances on any of the properties or assets of Parent or Purchaser pursuant to, any contract, order or other instrument binding on Parent or Purchaser or any applicable Law, or (ii) violate any order, writ, injunction, decree, judgment, statute, rule or regulation applicable to Parent or Purchaser, except, in each case, for matters that, individually or in the aggregate, would not reasonably be expected to prevent or materially delay or impair the consummation by Parent and Purchaser of the transactions contemplated by this Agreement or otherwise adversely impact in any material respect Parent's and Purchaser's ability to perform their respective obligations hereunder.

ARTICLE 4

ADDITIONAL COVENANTS OF THE STOCKHOLDERS

Each Stockholder hereby covenants and agrees that until the termination of this Agreement:

4.1 Voting of Subject Shares; Proxy.

(a) At any meeting of the stockholders of the Company held while this Agreement is in effect, however called, and at every adjournment or postponement thereof, or in connection with any written consent of the stockholders of the Company or in any other circumstances upon which a vote, consent or other approval of all or some of the stockholders of the Company is sought, at any time prior to the termination of this Agreement, such Stockholder shall, or shall cause the holder of record of its Subject Shares on any applicable record date to, appear at such meeting or otherwise cause such Stockholder's Subject Shares to be counted as present thereat for purposes of establishing a quorum at any such meeting and vote (or cause to be voted) all of such Stockholder's Subject Shares to the extent that any of the Subject Shares are entitled to vote at such meeting or in such written consent (the "Vote Shares") (i) against (A) any

action or agreement which is intended or would reasonably be expected to impede, delay, postpone, interfere with, nullify or prevent, in each case in any material respect, the Offer or the Merger, including, but not limited to, any other extraordinary corporate transaction, including a merger, acquisition, sale, consolidation, reorganization, recapitalization, extraordinary dividend or liquidation involving the Company and any Person (other than Parent, Purchaser or their Affiliates), or any other proposal of any Person (other than Parent, Purchaser or their Affiliates) to acquire the Company or all or substantially all of the assets thereof, (B) any Acquisition Proposal and any action in furtherance of any Acquisition Proposal or (C) any action, proposal, transaction or agreement that would reasonably be expected to result in the occurrence of any condition to the Offer to not be satisfied or result in a breach of any covenant, representation or warranty or any other obligation of the Company under the Merger Agreement or of such Stockholder under this Agreement and/or (ii) in favor of any other matter necessary or in furtherance of the Merger Agreement or the transactions contemplated by the Merger Agreement.

(b) Such Stockholder hereby revokes (and agrees to cause to be revoked) any proxies that such Stockholder has heretofore granted with respect to the Subject Shares. Solely with respect to the matters described in Section 4.1(a), while this Agreement is in effect, such Stockholder hereby irrevocably grants to, and appoints, Parent and any designee thereof, as such Stockholder's proxy and attorney-in-fact, for and in the name, place and stead of such Stockholder, to (i) attend any meeting of the stockholders of the Company on behalf of such Stockholder with respect to the matters set forth in Section 4.1(a), (ii) cause such Stockholder's Subject Shares to be counted as present for purposes of establishing a quorum at any such meeting and (iii) vote all Vote Shares, or grant or withhold a consent or approval in respect of all Vote Shares, or issue instructions to the record holder of such Stockholder's Vote Shares to do any of the foregoing, in connection with any meeting of the stockholders of the Company or any action by written consent in lieu of a meeting of the stockholders of the Company with respect to the matters set forth in Section 4.1(a), in a manner consistent with the provisions of Section 4.1(a). Such Stockholder authorizes such proxy and attorney-in-fact to substitute any other Person to act hereunder, to revoke any substitution and to file the proxy and any substitution or revocation with the Secretary of the Company. Such Stockholder hereby affirms that the irrevocable proxy set forth in this Section 4.1(b) is given in connection with the execution of the Merger Agreement and granted in consideration of and as an inducement to Parent and Purchaser to enter into the Merger Agreement, and that such irrevocable proxy is given to secure the performance of the duties of such Stockholder under this Agreement, subject to the termination of this Agreement pursuant to Section 5.2. Such Stockholder hereby further affirms that the proxy set forth in this Section 4.1(b) is coupled with an interest, is intended to be irrevocable (and as such shall survive and shall not be affected by the death, incapacity, mental illness or insanity of such Stockholder, as applicable), subject, however, to its automatic termination upon the termination of this Agreement pursuant to Section 5.2, and shall not be terminated by operation of law or upon the occurrence of any other event other than the termination of this Agreement pursuant to Section 5.2. Parent agrees not to exercise the proxy granted herein for any purpose other than the purposes described in this Agreement.

4.2 No Transfer; No Inconsistent Arrangements.

(a) Except as provided hereunder (including pursuant to Section 1.1 or Section 4.1) or under the Merger Agreement, from and after the date hereof and until this Agreement is terminated, such Stockholder shall not, directly or indirectly, voluntarily or involuntarily, (i) create or permit to exist any Encumbrance, other than Permitted Encumbrances, on any of such Stockholder's Subject Shares, (ii) transfer, sell, assign, gift, hedge, mortgage, pledge or otherwise dispose of (including by merger, by tendering into any tender or exchange offer, by testamentary disposition, by operation of law or otherwise), or enter into any derivative arrangement with respect to (collectively, "Transfer"), any or all of such Stockholder's Subject Shares, or any right or interest therein (or consent to any of the foregoing), (iii) enter into any contract, option or other agreement, arrangement or understanding with respect to any Transfer of such Subject Shares or any interest therein, (iv) grant or permit the grant of any proxy, power-of-attorney or other authorization or consent in or with respect to any such Stockholder's Subject Shares, (v) deposit or permit the deposit of any of such Stockholder's Subject Shares, into a voting trust or enter into a voting agreement or arrangement with respect to any of such Stockholder's Subject Shares or (vi) directly or indirectly, take or permit any other action that would reasonably be expected to in any way materially restrict, limit or interfere with the performance of such Stockholder's obligations hereunder or otherwise make any representation or warranty of such Stockholder herein untrue or incorrect in any material respect; provided that the restrictions contained in this Section 4.2 shall not apply with respect to any transfer of the Subject Shares by a Stockholder pursuant to applicable Laws of descent and distribution; provided, further, that any such proposed transferee must agree in writing to take such Subject Shares subject to and to be bound by the terms and conditions of this Agreement applicable to such Subject Shares. Any action taken in violation of the foregoing sentence shall be null and void ab initio and such Stockholder agrees that any such prohibited action may be enjoined. If any involuntary Transfer of any of the Subject Shares shall occur (including, but not limited to, a sale by such Stockholder's trustee in any bankruptcy, or a sale to a purchaser at any creditor's or court sale), the transferee (which term, as used herein, shall include any and all transferees and subsequent transferees of the initial transferee) shall take and hold such Subject Shares subject to all of the restrictions, liabilities and rights under this Agreement, which shall continue in full force and effect until valid termination of this Agreement. Such Stockholder agrees that it shall not, and shall cause each of its Affiliates not to, become a member of a "group" (as defined under Section 13(d) of the Exchange Act) with respect to any equity interests in the Company for the purpose of opposing or competing with or taking any actions inconsistent with the transactions contemplated by the Merger Agreement. Notwithstanding the foregoing, such Stockholder may make Transfers of Subject Shares as Parent may agree in writing in its sole discretion. If so requested by Parent, such Stockholder agrees that the Subject Shares shall bear a legend stating that such Subject Shares are subject to this Agreement (provided, such legend shall be removed upon the valid termination of this Agreement).

(b) At all times until the Expiration Date, in furtherance of this Agreement, such Stockholder shall, and hereby does, authorize and instruct the Company or its counsel to notify the Company's transfer agent that, from the date each Stockholder tenders its Subject Shares in accordance with Section 1.1 until the Expiration Date, there is a stop transfer order with respect to all of the Subject Shares of such Stockholder (and that this Agreement places limits on the voting and transfer of such Shares until the Expiration Date) subject to the tendering

of the Subject Shares in accordance with this Agreement and subject to any permitted Transfers pursuant to Section 4.2(a); provided, however, that (i) if this Agreement shall have been terminated in accordance with its terms or (ii) immediately following the Closing (and in any event within such time as would not delay receipt by the Stockholders of the Merger Consideration), the foregoing authorization and instruction shall be null and void and shall have no further force or effect. Until this Agreement is terminated in accordance with its terms, the obligations of such Stockholder specified in this Agreement shall apply whether or not the Company Board shall have made an Adverse Change Recommendation.

- **(c)** For the avoidance of doubt, neither any forfeiture of Shares, net settlement or purchase by the Company to satisfy tax withholding obligations nor sale of Shares to the extent required to satisfy any tax withholding obligation upon time vesting of a Company RSU occurring prior to the Offer Acceptance Time, shall be deemed to breach any provision hereof.
- **(d)** Notwithstanding Section 4.2(a) above, such Stockholder may make Transfers, for estate planning purposes or as a gift or charitable donation, of up to 100,000 shares of Common Stock held by such Stockholder; <u>provided</u>, <u>further</u>, that any such proposed transferee must agree in writing to take such Subject Shares subject to and to be bound by the terms and conditions of this Agreement applicable to such Subject Shares.
- **4.3** No Solicitation. Such Stockholder shall not, and shall cause its Representatives not to, directly or indirectly, if and to the extent prohibited by Section 5.4 of the Merger Agreement (assuming for this purpose that such Stockholder is the "Company" as such term is defined in the Merger Agreement): (i) solicit, initiate or knowingly encourage (including by way of furnishing nonpublic information) any Acquisition Proposal, (ii) enter into, continue or otherwise participate in any discussions or negotiations regarding, or furnish to any Person any non-public information that encourages or facilitates, or could reasonably be expected to lead to, any Acquisition Proposal or otherwise knowingly facilitate any effort or attempt to make any Acquisition Proposal, or (iii) propose, resolve or agree to do any of the foregoing. Upon execution of this Agreement, such Stockholder shall immediately cease and cause to be terminated any negotiations with any Persons (other than Parent and Purchaser and their Representatives) that may be ongoing with respect to an Acquisition Proposal. Such Stockholder agrees that any breach of this Section 4.3 by any of its Subsidiaries, if any, or its or their respective Representatives shall be deemed to be a breach of this Section 4.3 by such Stockholder. Notwithstanding the foregoing, nothing in this Agreement shall prohibit a Stockholder from taking any action in its capacity as a Representative of the Company that is expressly permitted to be taken by the Company's Representatives (in their capacity as such) under the Merger Agreement (including exercising fiduciary duties to the Company and its stockholders in accordance with applicable Law and the Merger Agreement).
- **4.4** No Exercise of Appraisal Rights. Provided that the Merger is consummated in accordance with the terms of the Merger Agreement, that the consideration offered pursuant to the Merger is not less than as set forth in the Merger Agreement and that this Agreement has not been terminated, such Stockholder waives and agrees not to exercise, assert or perfect, or attempt to exercise, assert or perfect any appraisal rights or dissenters' rights in respect of such Stockholder's Subject Shares that may arise with respect to the Merger and/or the transactions contemplated by the Merger Agreement, including, without limitation, under Section 262 of the DGCL, a copy of which is attached hereto as Exhibit A.

- 4.5 <u>Documentation and Information</u>. Such Stockholder shall not make any public announcement regarding this Agreement, the Offer, the Merger, the Merger Agreement or the transactions contemplated hereby or thereby without the prior written consent of Parent, except as may be required by applicable Law (provided that reasonable advance notice of any such disclosure will be provided to Parent). Such Stockholder consents to and hereby authorizes Parent, Purchaser and/or their respective Affiliates to publish and disclose in all documents and schedules filed with the SEC, and in any press release or other disclosure document reasonably identified by Parent, Purchaser and/or their respective Affiliates in connection with the Offer, the Merger or any other transactions contemplated by the Merger Agreement, such Stockholder's identity and ownership of the Subject Shares, the existence of this Agreement and the nature of such Stockholder's commitments and obligations under this Agreement, and such Stockholder acknowledges that Parent, Purchaser and/or their respective Affiliates may, in Parent's sole discretion, file this Agreement or a form hereof with the SEC or any other Governmental Body to the extent required by Law. Such Stockholder agrees to reasonably promptly give each of Parent, Purchaser and/or their respective Affiliates any information it may reasonably require with respect to such Stockholder and such Stockholder's Subject Shares for the preparation of any such disclosure documents, and such Stockholder agrees to promptly notify such parties of any required corrections with respect to any written information supplied by such Stockholder specifically for use in any such disclosure document (including any change in such Stockholder's record or beneficial ownership of shares of Common Stock), if and to the extent that any such information shall have become false or misleading in any material respect.
- **4.6** <u>Litigation</u>. Such Stockholder agrees not to commence, join in, knowingly facilitate, knowingly assist or knowingly encourage, and agrees to take all actions necessary to opt out of any class in any class action with respect to, any claim against Parent, Purchaser, the Company or any of their respective directors or officers related to the Merger Agreement, the Offer or the Merger, including any claim (a) challenging the validity of, or seeking to enjoin the operation of, any provision of this Agreement or the Merger Agreement or (b) alleging a breach of fiduciary duty of any Person in connection with the evaluation, negotiation or entry into the Merger Agreement or any of the transactions contemplated thereby; provided that the foregoing shall not limit any and all actions in any such litigation taken by such Stockholder in response to claims commenced against such Stockholder.
- **4.7** Adjustments. In the event (x) of any stock split, stock dividend, merger, reorganization, recapitalization, reclassification, combination, exchange of shares or the like of the capital stock of the Company affecting the Subject Shares or (y) that such Stockholder shall become the record or beneficial owner of any additional shares of Company Common Stock, then the terms of this Agreement shall apply to the shares of Company Common Stock held by such Stockholder immediately following the effectiveness of the events described in clause (x) above or such Stockholder's becoming the record or beneficial owner thereof as described in clause (y) above, as though, in either case, they were Subject Shares hereunder. In the event that such Stockholder shall become the beneficial owner of any other equity interests entitling the holder thereof to vote or give consent with respect to the matters set forth in Section 4.1 hereof, then the terms of Section 4.1 hereof shall apply to such other equity interests as though they were Subject Shares hereunder.

ARTICLE 5

MISCELLANEOUS

5.1 Notices. Any notice or other communication required or permitted to be delivered to any Party under this Agreement shall be in writing and shall be deemed properly delivered, given and received (a) upon receipt when delivered by hand, (b) two business days after being sent by registered mail or by courier or express delivery service, (c) if sent by email transmission prior to 6:00 p.m. recipient's local time, upon transmission when receipt is confirmed or (d) if sent by email transmission after 6:00 p.m. recipient's local time and receipt is confirmed, the business day following the date of transmission; provided that, in each case, the notice or other communication is sent to the physical address or email address set forth beneath the name of such Party below (or to such other physical address or email address as such Party shall have specified in a written notice given to the other parties hereto):

If to Parent or Purchaser, addressed to it at:

Altair Engineering Inc. 1820 E. Big Beaver Road Troy, Michigan 48083 Email: jrs@altair.com Attention: Mr. James R. Scapa

with a copy to (which shall not constitute actual or constructive notice):

Steven M. Rivkin, Esq. Raoul Maitra, Esq. Altair Engineering Inc. 1820 E. Big Beaver Road Troy, Michigan 48083

Email: srivkin@altair.com and Raoul@altair.com

and

Lowenstein Sandler LLP 1251 Avenue of the Americas New York, New York 10020 Fmail: pehrenberg@lowenstein

Email: pehrenberg@lowenstein.com Attention: Peter H. Ehrenberg, Esq. If to a Stockholder, addressed to such Stockholder at such Stockholder's facsimile number or email address set forth on a signature page hereto or to the last known address of such Stockholder in the Company's records, or to such other address, facsimile number or email address as such Party may hereafter specify in writing for such purpose by notice to each other Party hereto.

- **5.2 Termination**. This Agreement shall terminate automatically, without any notice or other action by any Person, upon the first to occur of (i) the valid termination of the Merger Agreement pursuant to the terms thereof, (ii) the Effective Time or (iii) the date of any modification, waiver or amendment to any provision of the Merger Agreement or any waiver of Parent or Purchaser's obligations under the Merger Agreement, in each case, that results either (x) in a reduction in the amount or change in the form of consideration payable to the holders of Shares pursuant to the Merger Agreement as in effect on the date of this Agreement or (y) a material change in the timing after the Effective Time in which such consideration is to be paid to the Stockholders pursuant to the Merger Agreement as in effect on the date of this Agreement. Upon termination of this Agreement with respect to any Party, such Party shall not have any further obligations or liabilities under this Agreement; provided, however, that (A) nothing set forth in this Section 5.2 shall relieve any Party from liability for any Willful Breach of this Agreement prior to termination hereof and (B) the provisions of this Article 5 shall survive any termination of this Agreement.
- **5.3** <u>Amendments and Waivers</u>. Any provision of this Agreement may be amended or waived if such amendment or waiver is in writing and is signed, in the case of an amendment, by each of (x) Parent and Purchaser, on the one hand, and (y) the Stockholder with respect to which such amendment is to be effective, on the other, or, in the case of a waiver, by each Party against whom the waiver is to be effective. No failure or delay by any Party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege.
- **5.4** Expenses. All fees and expenses incurred in connection herewith and the transactions contemplated hereby shall be paid by the Party incurring such fees and expenses, whether or not the Offer or the Merger is consummated.
- **5.5** <u>Assignability.</u> This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the Parties hereto and their respective successors and permitted assigns; provided, however, that neither this Agreement nor any of the rights hereunder may be assigned without (x) in the case of an assignment by Parent and/or Purchaser, the prior written consent of each Stockholder and (y) in the case of an assignment by a Stockholder, the prior written consent of Parent and Purchaser, and any attempted assignment of this Agreement or any of such rights without such consent shall be void and of no effect; provided, further, however, that Parent or Purchaser may assign this Agreement to any of their respective Affiliates; provided that no such assignment permitted pursuant to this Section 5.5 shall relieve Parent of its obligations hereunder.

5.6 No Third Party Beneficiaries. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the Parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

5.7 Applicable Law; Jurisdiction; Specific Performance; Remedies.

(a) This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without regard to the conflicts of law rules of such State or of any other jurisdiction that would cause the application of the laws of any jurisdiction other than the State of Delaware. Subject to Section 5.7(c), in any action or proceeding arising out of or relating to this Agreement or any of the transactions contemplated by this Agreement: (i) each of the Parties irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Chancery Court of the State of Delaware and any state appellate court therefrom or, if such court lacks subject matter jurisdiction, a federal court sitting in the State of Delaware (it being agreed that the consents to jurisdiction and venue set forth in this Section 5.7(a) shall not constitute general consents to service of process in the State of Delaware and shall have no effect for any purpose except as provided this Section 5.7(a) and shall not be deemed to confer rights on any Person other than the Parties hereto); and (ii) each of the parties irrevocably consents to service of process by first class certified mail, return receipt requested, postage prepaid, to the address at which such Party is to receive notice in accordance with Section 5.1. The Parties hereto agree that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable Law; provided, however, that nothing in the foregoing shall restrict any Party's rights to seek any post-judgment relief regarding, or any appeal from, such final trial court judgment.

(b) The Parties hereto agree that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur in the event that the Parties hereto do not perform their obligations under the provisions of this Agreement in accordance with its specified terms or otherwise breach such provisions. Subject to the following sentence, the Parties acknowledge and agree that (i) the Parties shall be entitled to an injunction or injunctions, specific performance, or other equitable relief, to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in the courts described in Section 5.7(a) without proof of damages or otherwise prior to the valid termination of this Agreement in accordance with Section 5.2, this being in addition to any other remedy to which they are entitled under this Agreement and (ii) the right of specific performance is an integral part of the transactions contemplated by this Agreement and without that right, none of the Parties would have entered into this Agreement and Parent and Purchaser would not have entered into the Merger Agreement. Each of the Parties hereto agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that the other Parties hereto have an adequate remedy at law or an award of specific performance is not an appropriate remedy for any reason at law or equity. The Parties hereto acknowledge and agree that any Party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in accordance with this Section 5.7(b) shall not be required to secure, post or provide any bond or other security in connection with any such order or injunction.

- (c) EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING BETWEEN THE PARTIES HERETO (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE), INCLUDING ANY COUNTERCLAIM, ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THE ACTIONS OF ANY PARTY IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT THEREOF. EACH PARTY (A) MAKES THIS WAIVER VOLUNTARILY AND (B) ACKNOWLEDGES THAT SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND, IN THE CASE OF PARENT AND PURCHASER, TO ENTER INTO THE MERGER AGREEMENT, BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS CONTAINED IN THIS SECTION 5.7(c).
- **5.8** Entire Agreement; Counterparts. This Agreement, together with Schedule A and Exhibit A, constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among or between any of the Parties, with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by PDF shall be sufficient to bind the Parties to the terms and conditions of this Agreement.
- **5.9** Severability. Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective only to the extent of such prohibition or unenforceability without invalidating the remaining provisions of this Agreement or affecting the validity or enforceability of such provision in any other jurisdiction. The application of such invalid or unenforceable provision to Persons or circumstances other than those as to which it is held invalid or unenforceable shall be valid and be enforced to the fullest extent permitted by applicable Law. To the extent any provision of this Agreement is determined to be prohibited or unenforceable in any jurisdiction, or determined to be impermissible by any Governmental Body, each of the Parties agrees to use reasonable commercial efforts to substitute one or more valid, legal and enforceable provisions that, insofar as practicable, implement the purposes and intent of the prohibited, unenforceable, or impermissible provision.

5.10 Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- **(b)** The Parties agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party shall not be applied in the construction or interpretation of this Agreement or any other agreements or documents executed or prepared in connection herewith.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."

- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Articles," "Exhibits" or "Schedules" are intended to refer to Sections or Articles of this Agreement or Exhibits or Schedules to this Agreement.
- **(e)** The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.
- **(f)** Unless the context requires otherwise (i) any reference herein to any Person shall be construed to include such Person's successors and assigns and (ii) the words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this Agreement in its entirety and not to any particular provision hereof.
- **5.11** <u>Further Assurances</u>. Parent, Purchaser and each Stockholder will execute and deliver, or cause to be executed and delivered, all further documents and instruments and use their respective reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable Law, to perform their respective obligations under this Agreement.
- **5.12** Capacity as Stockholder. Each Stockholder signs this Agreement solely in such Stockholder's capacity as a stockholder of the Company, and not in such Stockholder's capacity as a director, officer or employee of the Company or any Company subsidiary. Notwithstanding anything herein to the contrary, nothing herein shall in any way restrict a director or officer of the Company in the taking of any actions (or failure to act) in his or her capacity as a director or officer of the Company, or in the exercise of his or her fiduciary duties as a director or officer of the Company, or prevent or be construed to create any obligation on the part of any director or officer of the Company from taking any action in his or her capacity as such director or officer. Notwithstanding anything to the contrary contained herein but subject in all cases to all obligations imposed by the Merger Agreement, neither any action taken solely in any such capacity as an officer or director of the Company nor any action taken by the Company shall be deemed to constitute a breach of this Agreement, and the Company shall not be restricted or bound by this Agreement in any respect.
- **5.13** No Agreement Until Executed. This Agreement shall not be effective unless and until (i) the Merger Agreement is executed by all parties thereto and (ii) this Agreement is executed by all parties hereto.
- **5.14** Stockholder Obligation Several and Not Joint; Independent Nature of Obligations. The obligations of each Stockholder hereunder shall be several and not joint, and no Stockholder shall be liable for any breach of the terms of this Agreement by any other Stockholder. Each of Parent and Purchaser, on the one hand, and each Stockholder, on the other, shall be entitled to enforce its rights under this Agreement against the other, and it shall not be necessary for any other Stockholder to be joined as an additional party in any proceeding for such purpose. No Stockholder may enforce this Agreement against any other Stockholder party hereto. A default by any Stockholder of its obligations pursuant to this Agreement shall not relieve any other Stockholder of any of its obligations to Parent and/or Purchaser under this Agreement.

5.15 No Ownership Interest. Except as otherwise provided herein, nothing contained in this Agreement shall be deemed to vest in Parent or Purchaser any direct or indirect ownership or incidence of ownership of or with respect to the Subject Shares. Except as otherwise provided herein or in the Merger Agreement, all rights, ownership and economic benefits of and relating to the Subject Shares shall remain vested in and belong to each applicable Stockholder, and neither Parent nor Purchaser shall have any authority to manage, direct, restrict, regulate, govern, or administer any of the policies or operations of the Company or exercise any power or authority to direct such Stockholder in the voting of any of the Subject Shares.

(Signature page follows)

In	WITNESS WHEREOF	the parties hereto have caused	this Agreement to be	duly executed as of the	data first above written
11.	I WILLNESS WHEREUF.	. THE DAILIES HELETO HAVE CAUSED	THIS Agreement to be	' dury executed as or the	uate first above written.

,	۸	T	T.	۸	TD	EN	CIN	NEER	INC	INC
ı	٦	L		ч	ιк	F.IN	(+	N F. F. K	INIT	HNU.

By:	
_	Name: James R. Scapa
	Title: Chief Executive Officer
DAI	LLAS MERGER SUB, INC.
By:	

Name: James R. Scapa Title: President In WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

[STOCKHOLDER]

By:

Name: Email: Facsimile:

Address (optional):

Schedule A1

Name of Stockholder	Number of Subject Shares as of the Date Hereof		Purchases of Company Common Stock During the Prior Six Months
Michael Morrison	***Issued shares of Common Stock: 223,521	None.	
	**Company Options: 75,000		
	**Company RSUs: 168,334		
	The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares. *The foregoing includes 5,985 shares owned by Mr. Morrison's wife as custodian for two UTMA accounts.		
Joan C. McArdle	Issued shares of Common Stock: 14,499	None.	
	**Company Options: 0		
	**Company RSUs: 12,501		
	**The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares.		
James Eliason	Issued shares of Common Stock: 23,195	None.	
	**Company Options: 0		
	**Company RSUs: 99,666		
	**The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares.		
1.0	11 1	6 777 1	1.6 4 (11)

¹ Separate tender and support agreements that are, in all substantive respects identical to the as filed Form of Tender and Support Agreement will be signed by Carnegie Hill Associates, LLC covering 143,370 shares of Common Stock and WC Capital, LLC covering 665,966 shares of Common Stock.

Ken Tacelli

Issued shares of Common Stock: 35,017

None.

**Company Options: 0

**Company RSUs: 111,667

**The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares.

Christopher T. Cox

***Issued shares of Common Stock: 67,727

None.

**Company Options: 0

**Company RSUs: 12,501

**The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares.

*** The foregoing includes 35,000 shares owned by Mr. Cox's wife.

Richard de J. Osborne ***Issued shares of Common Stock: 239,631 **Company Options: 0 **Company RSUs: 12,501 **The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares. *** The foregoing includes 3,000 shares owned by Mr. Osborne's wife. Issued shares of Common Stock: 331,902 David C. Mahoney None. **Company Options: 0 **Company RSUs: 12,501 **The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares. Thomas H. Kelly Issued shares of Common Stock: 41,833 None. **Company Options: 0 **Company RSUs: 12,501 **The foregoing may be deemed to be

beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares.

Donald Friedman Issued shares of Common Stock: 15,665 None. **Company Options: 0 **Company RSUs: 13,335 **The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares. Issued shares of Common Stock: 12,665 Randy Seidl None. **Company Options: 0 **Company RSUs: 13,335 **The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares. Colin Mahony Issued shares of Common Stock: 2,333 None. **Company Options: 0 **Company RSUs: 11,667 **The foregoing may be deemed to be beneficially owned by the Stockholder, but the

**The foregoing may be deemed to be beneficially owned by the Stockholder, but the Stockholder does not have the right to vote any such Subject Shares.

Exhibit A

§ 262 Appraisal rights

- (a) Any stockholder of a corporation of this State who holds shares of stock on the date of the making of a demand pursuant to subsection (d) of this section with respect to such shares, who continuously holds such shares through the effective date of the merger or consolidation, who has otherwise complied with subsection (d) of this section and who has neither voted in favor of the merger or consolidation nor consented thereto in writing pursuant to § 228 of this title shall be entitled to an appraisal by the Court of Chancery of the fair value of the stockholder's shares of stock under the circumstances described in subsections (b) and (c) of this section. As used in this section, the word "stockholder" means a holder of record of stock in a corporation; the words "stock" and "share" mean and include what is ordinarily meant by those words; and the words "depository receipt" mean a receipt or other instrument issued by a depository representing an interest in 1 or more shares, or fractions thereof, solely of stock of a corporation, which stock is deposited with the depository.
- (b) Appraisal rights shall be available for the shares of any class or series of stock of a constituent corporation in a merger or consolidation to be effected pursuant to § 251 (other than a merger effected pursuant to § 251(g) of this title), § 252, § 254, § 255, § 256, § 257, § 258, § 263 or § 264 of this title:
- (1) Provided, however, that, except as expressly provided in § 363(b) of this title, no appraisal rights under this section shall be available for the shares of any class or series of stock, which stock, or depository receipts in respect thereof, at the record date fixed to determine the stockholders entitled to receive notice of the meeting of stockholders to act upon the agreement of merger or consolidation (or, in the case of a merger pursuant to § 251(h), as of immediately prior to the execution of the agreement of merger), were either: (i) listed on a national securities exchange or (ii) held of record by more than 2,000 holders; and further provided that no appraisal rights shall be available for any shares of stock of the constituent corporation surviving a merger if the merger did not require for its approval the vote of the stockholders of the surviving corporation as provided in § 251(f) of this title.
- (2) Notwithstanding paragraph (b)(1) of this section, appraisal rights under this section shall be available for the shares of any class or series of stock of a constituent corporation if the holders thereof are required by the terms of an agreement of merger or consolidation pursuant to §§ 251, 252, 254, 255, 256, 257, 258, 263 and 264 of this title to accept for such stock anything except:
 - a. Shares of stock of the corporation surviving or resulting from such merger or consolidation, or depository receipts in respect thereof;
- b. Shares of stock of any other corporation, or depository receipts in respect thereof, which shares of stock (or depository receipts in respect thereof) or depository receipts at the effective date of the merger or consolidation will be either listed on a national securities exchange or held of record by more than 2,000 holders;
 - c. Cash in lieu of fractional shares or fractional depository receipts described in the foregoing paragraphs (b)(2)a. and b. of this section; or

- d. Any combination of the shares of stock, depository receipts and cash in lieu of fractional shares or fractional depository receipts described in the foregoing paragraphs (b)(2)a., b. and c. of this section.
- (3) In the event all of the stock of a subsidiary Delaware corporation party to a merger effected under § 253 or § 267 of this title is not owned by the parent immediately prior to the merger, appraisal rights shall be available for the shares of the subsidiary Delaware corporation.
- (4) In the event of an amendment to a corporation's certificate of incorporation contemplated by § 363(a) of this title, appraisal rights shall be available as contemplated by § 363(b) of this title, and the procedures of this section, including those set forth in subsections (d) and (e) of this section, shall apply as nearly as practicable, with the word "amendment" substituted for the words "merger or consolidation," and the word "corporation" substituted for the words "constituent corporation" and/or "surviving or resulting corporation."
- (c) Any corporation may provide in its certificate of incorporation that appraisal rights under this section shall be available for the shares of any class or series of its stock as a result of an amendment to its certificate of incorporation, any merger or consolidation in which the corporation is a constituent corporation or the sale of all or substantially all of the assets of the corporation. If the certificate of incorporation contains such a provision, the provisions of this section, including those set forth in subsections (d), (e), and (g) of this section, shall apply as nearly as is practicable.
- (d) Appraisal rights shall be perfected as follows:
- (1) If a proposed merger or consolidation for which appraisal rights are provided under this section is to be submitted for approval at a meeting of stockholders, the corporation, not less than 20 days prior to the meeting, shall notify each of its stockholders who was such on the record date for notice of such meeting (or such members who received notice in accordance with § 255(c) of this title) with respect to shares for which appraisal rights are available pursuant to subsection (b) or (c) of this section that appraisal rights are available for any or all of the shares of the constituent corporations, and shall include in such notice a copy of this section and, if 1 of the constituent corporations is a nonstock corporation, a copy of § 114 of this title. Each stockholder electing to demand the appraisal of such stockholder's shares shall deliver to the corporation, before the taking of the vote on the merger or consolidation, a written demand for appraisal of such stockholder's shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such stockholder's shares. A proxy or vote against the merger or consolidation shall not constitute such a demand. A stockholder electing to take such action must do so by a separate written demand as herein provided. Within 10 days after the effective date of such merger or consolidation, the surviving or resulting corporation shall notify each stockholder of each constituent corporation who has complied with this subsection and has not voted in favor of or consented to the merger or consolidation of the date that the merger or consolidation has become effective; or

(2) If the merger or consolidation was approved pursuant to § 228, § 251(h), § 253, or § 267 of this title, then either a constituent corporation before the effective date of the merger or consolidation or the surviving or resulting corporation within 10 days thereafter shall notify each of the holders of any class or series of stock of such constituent corporation who are entitled to appraisal rights of the approval of the merger or consolidation and that appraisal rights are available for any or all shares of such class or series of stock of such constituent corporation, and shall include in such notice a copy of this section and, if 1 of the constituent corporations is a nonstock corporation, a copy of § 114 of this title. Such notice may, and, if given on or after the effective date of the merger or consolidation, shall, also notify such stockholders of the effective date of the merger or consolidation. Any stockholder entitled to appraisal rights may, within 20 days after the date of mailing of such notice or, in the case of a merger approved pursuant to § 251(h) of this title, within the later of the consummation of the offer contemplated by § 251(h) of this title and 20 days after the date of mailing of such notice, demand in writing from the surviving or resulting corporation the appraisal of such holder's shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such holder's shares. If such notice did not notify stockholders of the effective date of the merger or consolidation, either (i) each such constituent corporation shall send a second notice before the effective date of the merger or consolidation notifying each of the holders of any class or series of stock of such constituent corporation that are entitled to appraisal rights of the effective date of the merger or consolidation or (ii) the surviving or resulting corporation shall send such a second notice to all such holders on or within 10 days after such effective date; provided, however, that if such second notice is sent more than 20 days following the sending of the first notice or, in the case of a merger approved pursuant to § 251(h) of this title, later than the later of the consummation of the offer contemplated by § 251(h) of this title and 20 days following the sending of the first notice, such second notice need only be sent to each stockholder who is entitled to appraisal rights and who has demanded appraisal of such holder's shares in accordance with this subsection. An affidavit of the secretary or assistant secretary or of the transfer agent of the corporation that is required to give either notice that such notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein. For purposes of determining the stockholders entitled to receive either notice, each constituent corporation may fix, in advance, a record date that shall be not more than 10 days prior to the date the notice is given, provided, that if the notice is given on or after the effective date of the merger or consolidation, the record date shall be such effective date. If no record date is fixed and the notice is given prior to the effective date, the record date shall be the close of business on the day next preceding the day on which the notice is given.

(e) Within 120 days after the effective date of the merger or consolidation, the surviving or resulting corporation or any stockholder who has complied with subsections (a) and (d) of this section hereof and who is otherwise entitled to appraisal rights, may commence an appraisal proceeding by filing a petition in the Court of Chancery demanding a determination of the value of the stock of all such stockholders. Notwithstanding the foregoing, at any time within 60 days after the effective date of the merger or consolidation, any stockholder who has not commenced an appraisal proceeding or joined that proceeding as a named party shall have the right to withdraw such stockholder's demand for appraisal and to accept the terms offered upon the merger or consolidation. Within 120 days after the effective date of the merger or consolidation, any stockholder who has complied with the requirements of subsections (a) and (d) of this

section hereof, upon written request, shall be entitled to receive from the corporation surviving the merger or resulting from the consolidation a statement setting forth the aggregate number of shares not voted in favor of the merger or consolidation (or, in the case of a merger approved pursuant to § 251(h) of this title, the aggregate number of shares (other than any excluded stock (as defined in § 251(h)(6)d. of this title)) that were the subject of, and were not tendered into, and accepted for purchase or exchange in, the offer referred to in § 251(h)(2)), and, in either case, with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. Such written statement shall be mailed to the stockholder within 10 days after such stockholder's written request for such a statement is received by the surviving or resulting corporation or within 10 days after expiration of the period for delivery of demands for appraisal under subsection (d) of this section hereof, whichever is later. Notwithstanding subsection (a) of this section, a person who is the beneficial owner of shares of such stock held either in a voting trust or by a nominee on behalf of such person may, in such person's own name, file a petition or request from the corporation the statement described in this subsection.

- (f) Upon the filing of any such petition by a stockholder, service of a copy thereof shall be made upon the surviving or resulting corporation, which shall within 20 days after such service file in the office of the Register in Chancery in which the petition was filed a duly verified list containing the names and addresses of all stockholders who have demanded payment for their shares and with whom agreements as to the value of their shares have not been reached by the surviving or resulting corporation. If the petition shall be filed by the surviving or resulting corporation, the petition shall be accompanied by such a duly verified list. The Register in Chancery, if so ordered by the Court, shall give notice of the time and place fixed for the hearing of such petition by registered or certified mail to the surviving or resulting corporation and to the stockholders shown on the list at the addresses therein stated. Such notice shall also be given by 1 or more publications at least 1 week before the day of the hearing, in a newspaper of general circulation published in the City of Wilmington, Delaware or such publication as the Court deems advisable. The forms of the notices by mail and by publication shall be approved by the Court, and the costs thereof shall be borne by the surviving or resulting corporation.
- (g) At the hearing on such petition, the Court shall determine the stockholders who have complied with this section and who have become entitled to appraisal rights. The Court may require the stockholders who have demanded an appraisal for their shares and who hold stock represented by certificates to submit their certificates of stock to the Register in Chancery for notation thereon of the pendency of the appraisal proceedings; and if any stockholder fails to comply with such direction, the Court may dismiss the proceedings as to such stockholder. If immediately before the merger or consolidation the shares of the class or series of stock of the constituent corporation as to which appraisal rights are available were listed on a national securities exchange, the Court shall dismiss the proceedings as to all holders of such shares who are otherwise entitled to appraisal rights unless (1) the total number of shares entitled to appraisal exceeds 1% of the outstanding shares of the class or series eligible for appraisal, (2) the value of the consideration provided in the merger or consolidation for such total number of shares exceeds \$1 million, or (3) the merger was approved pursuant to § 253 or § 267 of this title.

- (h) After the Court determines the stockholders entitled to an appraisal, the appraisal proceeding shall be conducted in accordance with the rules of the Court of Chancery, including any rules specifically governing appraisal proceedings. Through such proceeding the Court shall determine the fair value of the shares exclusive of any element of value arising from the accomplishment or expectation of the merger or consolidation, together with interest, if any, to be paid upon the amount determined to be the fair value. In determining such fair value, the Court shall take into account all relevant factors. Unless the Court in its discretion determines otherwise for good cause shown, and except as provided in this subsection, interest from the effective date of the merger through the date of payment of the judgment shall be compounded quarterly and shall accrue at 5% over the Federal Reserve discount rate (including any surcharge) as established from time to time during the period between the effective date of the merger and the date of payment of the judgment. At any time before the entry of judgment in the proceedings, the surviving corporation may pay to each stockholder entitled to appraisal an amount in cash, in which case interest shall accrue thereafter as provided herein only upon the sum of (1) the difference, if any, between the amount so paid and the fair value of the shares as determined by the Court, and (2) interest theretofore accrued, unless paid at that time. Upon application by the surviving or resulting corporation or by any stockholder entitled to participate in the appraisal proceeding, the Court may, in its discretion, proceed to trial upon the appraisal prior to the final determination of the stockholders entitled to an appraisal. Any stockholder whose name appears on the list filed by the surviving or resulting corporation pursuant to subsection (f) of this section and who has submitted such stockholder's certificates of stock to the Register in Chancery, if such is required, may participate
- (i) The Court shall direct the payment of the fair value of the shares, together with interest, if any, by the surviving or resulting corporation to the stockholders entitled thereto. Payment shall be so made to each such stockholder, in the case of holders of uncertificated stock forthwith, and the case of holders of shares represented by certificates upon the surrender to the corporation of the certificates representing such stock. The Court's decree may be enforced as other decrees in the Court of Chancery may be enforced, whether such surviving or resulting corporation be a corporation of this State or of any state.
- (j) The costs of the proceeding may be determined by the Court and taxed upon the parties as the Court deems equitable in the circumstances. Upon application of a stockholder, the Court may order all or a portion of the expenses incurred by any stockholder in connection with the appraisal proceeding, including, without limitation, reasonable attorney's fees and the fees and expenses of experts, to be charged pro rata against the value of all the shares entitled to an appraisal.
- (k) From and after the effective date of the merger or consolidation, no stockholder who has demanded appraisal rights as provided in subsection (d) of this section shall be entitled to vote such stock for any purpose or to receive payment of dividends or other distributions on the stock (except dividends or other distributions payable to stockholders of record at a date which is prior to the effective date of the merger or consolidation); provided, however, that if no petition for an appraisal shall be filed within the time provided in subsection (e) of this section, or if such stockholder shall deliver to the surviving or resulting corporation a written withdrawal of such stockholder's demand for an appraisal and an acceptance of the merger or consolidation, either within 60 days after the effective date of the merger or consolidation as provided in subsection

(e) of this section or thereafter with the written approval of the corporation, then the right of such stockholder to an appraisal shall cease. Notwithstanding the foregoing, no appraisal proceeding in the Court of Chancery shall be dismissed as to any stockholder without the approval of the Court, and such approval may be conditioned upon such terms as the Court deems just; provided, however that this provision shall not affect the right of any stockholder who has not commenced an appraisal proceeding or joined that proceeding as a named party to withdraw such stockholder's demand for appraisal and to accept the terms offered upon the merger or consolidation within 60 days after the effective date of the merger or consolidation, as set forth in subsection (e) of this section.

(l) The shares of the surviving or resulting corporation to which the shares of such objecting stockholders would have been converted had they assented to the merger or consolidation shall have the status of authorized and unissued shares of the surviving or resulting corporation.

FIRST AMENDMENT TO THIRD AMENDED AND RESTATED CREDIT AGREEMENT

This First Amendment to Third Amended and Restated Credit Agreement, dated as of October 31, 2018 (this "Amendment"), is among Altair Engineering Inc. (the "Borrower"), the Lenders party hereto, and JPMorgan Chase Bank, N.A., as Administrative Agent (in such capacity, the "Administrative Agent").

RECITAL

The Borrower, the Lenders party thereto and the Administrative Agent are parties to a Third Amended and Restated Credit Agreement dated as of October 18, 2017 (as it may be amended or modified from time to time, the "Credit Agreement"). The Borrower desires to amend the Credit Agreement as set forth herein and the Lenders are willing to do so in accordance with the terms hereof. Capitalized terms used herein, but not otherwise defined shall have the meanings ascribed to them in the Credit Agreement.

TERMS

In consideration of the premises and of the mutual agreements herein contained, the parties agree as follows:

ARTICLE 1. AMENDMENTS

Upon the satisfaction of the conditions set forth in Article 4 hereof, the Credit Agreement shall be amended as follows:

1.1 The following definitions are added to Section 1.01 of the Credit Agreement in their respective alphabetical location.

"First Amendment" means the First Amendment to the Third Amended and Restated Credit Agreement dated as of the First Amendment Effective Date by and among the Borrower, the Lenders party thereto and the Administrative Agent.

"First Amendment Effective Date" means October 31, 2018.

1.2 The following definitions in Section 1.01 of the Credit Agreement are restated as follows:

"Prime Rate" means the rate of interest last quoted by The Wall Street Journal as the "Prime Rate" in the U.S. or, if The Wall Street Journal ceases to quote such rate, the highest per annum interest rate published by the Federal Reserve Board in Federal Reserve Statistical Release H.15 (519) (Selected Interest Rates) as the "bank prime loan" rate or, if such rate is no longer quoted therein, any similar rate quoted therein (as determined by the Administrative Agent) or any similar release by the Federal Reserve Board (as determined by the Administrative Agent). Each change in the Prime Rate shall be effective from and including the date such change is publicly announced or quoted as being effective.

"Revolving Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Revolving Loans and to acquire participations in Letters of Credit and Swingline Loans hereunder, expressed as an amount representing the maximum aggregate permitted amount of such Lender's Revolving Exposure hereunder, as such commitment may be reduced or increased from time to time pursuant to (a) Section 2.09 or 2.23 of this Agreement and (b) assignments by or to such Lender pursuant to Section 9.04. The initial amount of each Lender's Revolving Commitment is set forth on the Commitment Schedule, or in the Assignment and Assumption pursuant to which such Lender shall have assumed its Revolving Commitment, as applicable. The aggregate amount of the Lenders' Revolving Commitments as of the Effective Date is \$100,000,000, and the aggregate amount of the Lenders' Revolving Commitments as of the First Amendment Effective Date is \$150,000,000.

- 1.3 The Commitment Schedule attached to the Credit Agreement is replaced with the Commitment Schedule attached hereto.
- 1.4 Notwithstanding anything in the Credit Agreement to the contrary, the parties hereto acknowledge and agree that this Amendment shall be not deemed to have used the allowable increase of the Aggregate Revolving Commitments under Section 2.23.

ARTICLE 2. REPRESENTATIONS.

In order to induce the Lenders and the Administrative Agent to enter into this Amendment, each Loan Party represents and warrants to each Lender and the Administrative Agent that the following statements are true, correct and complete as of the First Amendment Effective Date:

- 2.1 The execution, delivery and performance of this Amendment are within its powers and have been duly authorized by it.
- 2.2 This Amendment is the legal, valid and binding obligation of it, enforceable against it in accordance with the terms hereof, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally and subject to general principles of equity, regardless of whether considered in a proceeding in equity or at law.
- 2.3 After giving effect to the amendments herein contained, the representations and warranties of each Loan Party set forth in the Credit Agreement or in any other Loan Document shall be true and correct in all material respects on and as of the First Amendment Effective Date (it being understood and agreed that any representation or warranty which by its terms is made as of a specified date shall be required to be true and correct in all material respects only as of such specified date, and that any representation or warranty which is subject to any materiality qualifier shall be required to be true and correct in all respects).

2.4 After giving effect to the amendments herein contained, no Default shall have occurred and be continuing.

ARTICLE 3. CONDITIONS PRECEDENT.

This Amendment shall be effective as of the date hereof when each of the following conditions is satisfied or waived by the Administrative Agent:

- 3.1 This Amendment shall be executed by the Borrower and the Lenders.
- 3.2 The Guarantors shall have signed the Consent and Agreement hereto.
- 3.3 The Loan Parties shall have delivered such officers' certificates, resolutions, opinions of counsel and other documents as required by the Administrative Agent.

ARTICLE 4. MISCELLANEOUS.

- 4.1 References in the Loan Documents to the Credit Agreement shall be deemed to be references to the Credit Agreement as amended hereby and as further amended from time to time. This Amendment is a Loan Document.
- 4.2 Except as expressly amended hereby, the Borrower agrees that the Loan Documents are ratified and confirmed and shall remain in full force and effect and that it has no set off, counterclaim, defense or other claim or dispute with respect to any of the foregoing. Nothing herein shall be deemed to entitle the Borrower to any future consent to, or waiver, amendment, modification or other change of, any of the terms, conditions, obligations, covenants or agreements contained in the Credit Agreement or any other Loan Document in similar or different circumstances.
- 4.3 This Amendment may be signed upon any number of counterparts with the same effect as if the signatures thereto and hereto were upon the same instrument and signatures sent by facsimile or other electronic imaging shall be enforceable as originals.
 - 4.4 This Amendment is governed by, and construed in accordance with, the law of the State of Michigan.

[Signature Pages Follow]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed by their respective authorized officers as of the day and year first above written.

ALTAIR ENGINEERING INC.

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Chief Financial Officer

SIGNATURE PAGE - FIRST AMENDMENT TO THIRD AMENDED AND RESTATED CREDIT AGREEMENT - ALTAIR ENGINEERING INC.

JPMORGAN CHASE BANK, N.A.,

as a Lender and as Administrative Agent

By: /s/ Daniel J. Maniaci

Name: Daniel J. Maniaci Title: Vice President

SIGNATURE PAGE - FIRST AMENDMENT TO THIRD AMENDED AND RESTATED CREDIT AGREEMENT - ALTAIR ENGINEERING INC.

ROYAL BANK OF CANADA,

as Syndication Agent and as a Lender

By: /s/ Sheldon Pinto

Name: Sheldon Pinto Title: Authorized Signatory

SIGNATURE PAGE – FIRST AMENDMENT TO THIRD AMENDED AND RESTATED CREDIT AGREEMENT – ALTAIR ENGINEERING INC.

CONSENT AND AGREEMENT

Each of the undersigned hereby fully consents to the terms and provisions of the First Amendment to Third Amended and Restated Credit Agreement dated as of October , 2018 (the "First Amendment") among Altair Engineering Inc. (the "Borrower"), the Lenders party hereto, and JPMorgan Chase Bank, N.A., as Administrative Agent, and the consummation of the transactions contemplated by the First Amendment, and acknowledges and agrees to all terms and provisions of the First Amendment applicable to it, including without limitation all covenants, representations and warranties, releases, indemnifications, and all other terms and provisions, and further confirms and agrees that the Loan Documents, including without limitation the Credit Agreement as amended by the First Amendment, are ratified and confirmed and shall remain in full force and effect and that it has no set off, counterclaim, defense or other claim or dispute with respect to any of the foregoing.

[Signature Pages Follow]

ALTAIR ENGINEERING INC.

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Chief Financial Officer

ALTAIR PRODUCT DESIGN, INC.

By: /s/ Howard N. Morof

Name: Howard N. Morof

Title: Treasurer

ILUMISYS, INC.

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Treasurer

ALTAIR BELLINGHAM II, LLC

By: Altair Engineering Inc., its sole member

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Treasurer

ALTAIR CLEAN TECHNOLOGY CENTER, LLC

By: Altair Engineering Inc., its sole member

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Treasurer

SIGNATURE PAGE – CONSENT AND AGREEMENT – GUARANTORS – FIRST AMENDMENT TO THIRD AMENDED AND RESTATED CREDIT AGREEMENT – ALTAIR ENGINEERING INC.

ALTAIR BELLINGHAM LLC

By: Altair Engineering Inc., its sole member

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Treasurer

SOLIDTHINKING, INC.

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Treasurer

WEYV, INC.

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Treasurer

RUNTIME DESIGN AUTOMATION

By: /s/ Howard N. Morof

Name: Howard N. Morof Title: Treasurer

SIGNATURE PAGE – CONSENT AND AGREEMENT – GUARANTORS – FIRST AMENDMENT TO THIRD AMENDED AND RESTATED CREDIT AGREEMENT – ALTAIR ENGINEERING INC.

Commitment Schedule

Lender	Title	Revolving Commitment
JPMorgan Chase Bank, N.A.	Sole Lead Arranger and Administrative Agent	\$100,000,000.00
Royal Bank of Canada	Syndication Agent	\$50,000,000.00
	Total	\$150,000,000.00



Altair Announces Agreement to Acquire Datawatch

TROY, Michigan and BEDFORD, Massachusetts – November 5, 2018 – Altair (Nasdaq: ALTR) and Datawatch Corporation (Nasdaq-CM: DWCH) today announced the signing of a definitive merger agreement under which Altair has agreed to acquire Datawatch. Under the terms of the agreement, Altair will pay \$13.10 per share in cash, representing a fully diluted equity value of approximately \$176 million. The transaction was unanimously approved by the Boards of Directors of both companies.

James Scapa, Altair's Founder, Chairman, and Chief Executive Officer, commented, "Bringing Datawatch into Altair should result in a powerful offering consistent with our vision to transform product design and decision making by applying simulation, data science and optimization throughout product lifecycles. We see a convergence of simulation with the application of machine learning technology to live and historical sensor data as essential to creating better products, marketing them efficiently, and optimizing their in-service performance. Datawatch is a great team of people with best-in-class products, and we look forward to their joining us."

Altair believes the acquisition of Datawatch is compelling for a number of reasons, including:

- The data analytics and data science markets are evolving rapidly to leverage many of the same technologies, such as high-performance computing and visualization, as Altair has been leveraging in simulation for many years.
- Datawatch's solutions, which include data prep, data prediction, and real-time high-volume data visualization technologies, are highly
 relevant and applicable to almost any company and vertical market.
- There is strong opportunity to cross-sell Datawatch products into Altair's primarily manufacturing customer base, which will be facilitated by applying Altair's proven licensing models to Datawatch solutions.
- Datawatch's historical strength has been in the financial services and capital markets sectors. There is meaningful opportunity to disrupt these markets and expand usage by making it easier to access products through Altair's licensing model, and there is a cross-sell opportunity with some relevant Altair products in these markets.

Michael Morrison, Chief Executive Officer of Datawatch, added, "The Datawatch team is excited to join Altair and benefit from its long track record of success with developing and bringing to market highly differentiated software technology across diverse industry verticals. We feel great about the cultural alignment and look forward to driving continued innovation in our market-leading solutions as an integral part of Altair's vision."

Terms of the Transaction

Under the terms of the definitive merger agreement, Altair will commence a tender offer within ten business days to acquire all of the outstanding shares of common stock of Datawatch for \$13.10 per share in cash. This represents a 35% percent premium to the closing price of Datawatch's common stock on November 2, 2018. The tender offer is subject to customary closing conditions, including the tender of at least a majority of the outstanding shares of Datawatch common stock and the expiration or early termination of the applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976. Following the closing of the tender offer, a wholly-owned subsidiary of Altair will merge with and into Datawatch, with each share of Datawatch common stock that has not been tendered being converted into the right to receive the same \$13.10 per share in cash offered in the tender offer. The transaction is anticipated to close in the fourth quarter of 2018.

Funding for the transaction will come primarily from cash, coupled with utilization of borrowings under Altair's existing credit facility.

RBC Capital Markets, LLC is acting as exclusive financial advisor to Altair. GCA Advisors is acting as exclusive financial advisor to Datawatch. Legal counsel for Altair is Lowenstein Sandler LLP and legal counsel for Datawatch is Choate Hall & Stewart LLP.

Conference Call Details

Altair will host a conference call and webcast today, November 5, 2018, at 8:30 a.m. (Eastern Time), to discuss the details of the acquisition. Please note this call will only review today's announcement. The Company will release and review its third quarter 2018 results in its regularly scheduled earnings conference call after the close of market on Thursday, November 8th, 2018 at 4:30 p.m.

When: Monday, November 5, 2018

Time: 8:30 a.m. ET

Live Call: (866) 754-5204, Domestic

(636) 812-6621, International

Webcast: http://investor.altair.com (live and replay)

About Altair (Nasdag: ALTR)

Altair transforms design and decision making by applying simulation, machine learning and optimization throughout product lifecycles. Altair's broad portfolio of simulation technology and patented units-based software licensing model enable Simulation-Driven InnovationTM for Altair's customers. With more than 2,000 employees, Altair is headquartered in Troy, Michigan, USA and operates 71 offices throughout 24 countries. Altair serves more than 5,000 customers across broad industry segments.

About Datawatch (Nasdag-CM: DWCH)

Datawatch is the data intelligence provider with market leading enterprise data preparation, predictive analytics and visualization solutions that fuel business analytics. Datawatch positions individuals and organizations to master all data – no matter the origin, format or narrative – resulting in faster time to insight. Datawatch solutions are architected to drive the use of more data, foster more trust and incorporate more minds into business analytics. Thousands of organizations of all sizes in more than 100 countries worldwide use Datawatch products, including 93 of the Fortune 100. Datawatch is headquartered in Bedford, Massachusetts, with offices in New York, London, Toronto, Stockholm, Singapore and Manila.

Additional Information about the Transaction

The tender offer described in this communication (the "Offer") has not yet commenced, and this communication is neither an offer to purchase nor a solicitation of an offer to sell any shares of the common stock of Datawatch or any other securities. On the commencement date of the Offer, a tender offer statement on Schedule TO, including an offer to purchase, a letter of transmittal and related documents, will be filed with the United States Securities and Exchange Commission (the "SEC") and Datawatch will file a Solicitation/Recommendation Statement on Schedule 14D-9 relating to the Offer with the SEC. The offer to purchase shares of Datawatch common stock will only be made pursuant to the offer to purchase, the letter of transmittal and related documents filed with such Schedule TO. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ BOTH THE TENDER OFFER STATEMENT AND THE SOLICITATION/RECOMMENDATION STATEMENT REGARDING THE OFFER, AS THEY MAY BE AMENDED FROM TIME TO TIME, WHEN THEY BECOME AVAILABLE, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. The tender offer statement will be filed with the SEC by Datawatch. Investors and security holders may obtain a free copy of these statements (when available) and other documents filed with the SEC at the website maintained by the SEC at www.sec.gov or by directing such requests to D.F. King & Co., Inc. toll-free at (877) 864-5060.

Cautionary Language Concerning Forward-Looking Statements

This communication may contain, in addition to historical information, certain "forward-looking statements" within the meaning of the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995, including but not limited to, statements relating to Altair's and Datawatch's expectations regarding the completion and timing of the closing of the pending acquisition and their views regarding the opportunities resulting from the combination of Altair and Datawatch. These forward-looking statements are made as of the date of this release and are based on current expectations, estimates, forecasts and projections as well as the beliefs and assumptions of management. Words such as "expect," "anticipate," "should," "believe," "hope," "target," "project," "goals," "estimate," "protential," "predict," "may," "will," "might," "could," "intend," variations of these terms or the negative of these terms and similar expressions are intended to identify these forward-looking statements. Such forward-looking statements are not guarantees or predictions of future

performance, and are subject to known and unknown risks, uncertainties and other factors, many of which are beyond our control, that could cause actual results, performance or achievements of Altair or Datawatch following completion of the Offer and the merger described herein (collectively, the "Transactions") to differ materially from any future results, performance or achievements expressed or implied by such forward-looking statements. Such risks and uncertainties include: (i) the risk that not all conditions of the Offer or the merger will be satisfied or waived, (ii) uncertainties associated with any aspect of the Transactions, including uncertainties relating to the anticipated timing of filings and approvals relating to the Transactions, the outcome of any legal proceedings that may be instituted with respect to the Transactions, the expected timing of completion of the Transactions, the satisfaction of the conditions to the consummation of the Transactions and the ability to complete the Transactions, (iii) the risk that the combined company may not be able to sustain Altair's recent growth rate or grow at all, (iii) the risk that the combined company may not be able to maintain Altair's culture of innovation, teamwork, and communication, (iv) the risk that the combined company's existing customers or users do not increase their usage of the combined company's software, or that new customers are not added, (v) risks associated with the difficulty of predicting new customer generation due to the length of sales cycles impacting the combined company, (vi) risks associated with integrating acquired companies, (vii) the risk that the combined company's customers may reduce their spending on product design and development activities, (viii) risks associated with dependence on annual renewals of software licenses, (ix) risks associated with the need for the combined company to develop additional industry verticals, (x) risks associated with the significant competition that the combined company is likely to face, (xi) the susceptibility of Altair and Datawatch to factors affecting the primary industries that they serve, (xii) the risks associated with the fact that the quarterly results of the combined company may fluctuate significantly and not fully reflect the underlying performance of the combined company's business and (xiii) other risks detailed in the section entitled "Risk Factors" in Altair's Annual Report on Form 10-K for the year ended December 31, 2017, as updated by Altair's Quarterly Report on Form 10-Q for the quarters ended March 31, 2018 and June 30, 2018, and Datawatch's Annual Report on Form 10-K for the year ended September 30, 2017. Past performance is not necessarily indicative of future results. The forward-looking statements included in this press release represent Altair's and Datawatch's views as of the date of this press release. Altair and Datawatch anticipate that subsequent events and developments may cause their views to change. Altair and Datawatch undertake no intention or obligation to update or revise any forward-looking statements, whether as a result of new information, future events, changes in their expectations or otherwise. These forward-looking statements should not be relied upon as representing Altair's views or Datawatch's views as of any date subsequent to the date of this press release.

Investor Relations – Altair

Brian Denyeau ICR 248-614-2400 ext. 346 <u>ir@altair.com</u>

Media Relations -Altair

Dave Simon Altair 248-614-2400 ext. 332 pr@altair.com

Source: Altair Engineering Inc.

Investor Relations – Datawatch

Mary Conway Conway Communications (781) 772-1679 marytconway@comcast.net

Media Relations - Datawatch

Frank Moreno
Datawatch
978-275-8225
frank moreno@datawatch.com